EFFECT OF POTASSIUM, NITROGEN AND SILICON FERTILISATION ON SUGARCANE GROWTH AND QUALITY, NUTRIENT UPTAKE DYNAMICS AND SOIL CHEMISTRY IN TWO CONTRASTING SOILS OF KWAZULU-NATAL, SOUTH AFRICA

Ruth Rhodes

BSc Agric., MSc Cons. Biol.

Submitted in fulfilment of the academic requirements for the degree of **Doctor of Philosophy**

Discipline of Soil Science School of Agricultural, Earth and Environmental Sciences College of Agriculture, Engineering and Science University of KwaZulu-Natal Pietermaritzburg

ABSTRACT

The South African sugarcane industry covers an area of approximately 380 000 ha, producing an average of 2.2 million tonnes of sugar per year. South African sugarcane growers are supported by, amongst others, the South African Sugarcane Research Institute (SASRI), which conducts research and provides technical support and advice. The Fertiliser Advisory Service (FAS) at SASRI analyses growers' soil and plant samples and provides nutrient recommendations.

The three nutrients required in the largest amounts by sugarcane are potassium (K), nitrogen (N) and silicon (Si). Accurate estimates of the K, N and Si requirements of sugarcane are essential. This study was thus designed to examine the impact of K, N and Si treatments on sugarcane growth and quality. Also investigated were nutrient interactions, nutrient balances in the soil and crop, and rooting characteristics. The ultimate goal was to assist in the refinement of the industry nutrient recommendations given by the FAS.

Two contrasting soil types on the KwaZulu-Natal north coast were selected. One trial was situated on an Oxisol, and the other on an Inceptisol. At each trial site, three crops of sugarcane (*Saccharum officinarum* x *S. spontaneum* hybrid) were harvested: the 'plant crop' (the first crop harvested after planting) and the first two ratoon crops (ratoons one and two, which resprout following each harvest). Treatments were applied as potassium chloride at 0, 100, 200 and 300 kg K ha⁻¹ (both sites); urea at 0, 80 and 160 kg N ha⁻¹ (Oxisol) and 0, 105 and 210 kg N ha⁻¹ (Inceptisol); and Calmasil®, a calcium silicate slag, at 0 and 300 kg Si ha⁻¹ (both sites). The trial was factorial, with four replications of each nutrient combination in a randomised complete block design. Following harvest of the plant and first ratoon crops, K and N were re-applied. Silicon was reapplied, in the form of cement (CIMPOR NPC PRO blastfurnace Cement CEM III/A 32.5 N), following harvest of the first ratoon crop at both trial sites. The crops were harvested and stalk yield and percent estimated recoverable crystal (% ERC, a measure of extractable sucrose) determined.

Leaf samples were collected at both sites and analysed for nutrient content. Soil samples were collected in 20 cm increments to a depth of 80 cm following harvest of

each crop, and analysed at the FAS. Topsoils (0-20 cm) were analysed following fertiliser application at the start of each crop cycle. In addition, incremental depth 'topsoil' samples were collected at each site at 0-2.5, 2.5-7.5, 7.5-15 and 15-28 cm. Root samples were collected in 20 cm increments to a depth of 80 cm, and total root length per sample determined.

To investigate the behaviour of K further, a K-exhaustion pot trial was established using topsoil from each trial site and sown to ryegrass (*Lolium multiflorum* L.). The ryegrass was harvested periodically and analysed for K content. Total K removed by the plants over 340 days was determined and a K balance calculated.

In general, the K thresholds and recommendations currently used at the FAS appear appropriate for the soils under study. At the fertiliser and sucrose prices used, however, application of 200 kg K ha⁻¹ was not economically justifiable over the three crops at each site, but it is possible that the economics of K application might have become more favourable in later ratoons, where response to K is usually more marked.

At present, the FAS uses only topsoil readily-available K results to formulate K recommendations. This study showed that combining subsoil K data (20-60 cm) with those from the topsoil (0-20 cm) improved K recommendations.

Both field trials and the pot experiment clearly indicated that previously 'unavailable' soil K was released and made available for crop use, more so in the Inceptisol than in the Oxisol. Such pools may contribute to crop nutrition to a greater extent than currently recognised. Further investigation is warranted.

Based on the results of this study, little change would be required to the N rates currently recommended by the FAS. This study confirmed that excessive N may depress sucrose percent, and that plant crops respond less to N than do ratoon crops. Crop yield, sucrose percent and total sucrose yield in the plant crops did not respond positively to applied N. On the humic Oxisol, N application served to decrease sucrose yield significantly, and this requires further investigation. Reduced N rates, as are currently recommended by FAS on humic soils and in plant crops, were justified, although the present results suggest that a zero N recommendation could be applicable.

Results showed that the current threshold for soil Si (extracted with 0.01 M CaCl₂) of 15 mg L⁻¹ is insufficient to promote optimal yield. It is suggested that the FAS increase this to approximately 20 mg L⁻¹ and that further trials are required to finalise this threshold.

High soil K limited calcium (Ca) and magnesium (Mg) uptake by the plant. This finding is of considerable significance to the South African sugar industry and has implications for lime recommendations. Although this cationic antagonism is not directly related to Si application, Si is, at present, applied as a calcium silicate slag (such as Calmasil®). This is used in place of dolomitic lime as it fulfils a similar role by providing Ca and Mg (as well as Si), and raising soil pH. Furthermore, the FAS currently gives higher K recommendations for soils with high levels of Ca and Mg. It is recommended, based on the outcomes of this study, that the reverse should also apply so that where K levels are high, soil Ca and Mg thresholds should be increased. The details of such a recommendation will need to be confirmed by further studies.

Leaf sample results did not prove a reliable means of establishing nutrient sufficiency levels. Moisture stress during the study likely played a role in this disparity. This study showed that the soil sampling depth recommended by the laboratory should be strictly adhered to. Collection of samples at a shallower or deeper depth than that used in field calibration studies, will lead to incorrect recommendations, possibly resulting in reduced yield or quality, or wasteful expenditure on inputs.

Root distribution appeared to be normal with about 70% of the roots found in the top 40 cm of both soils. Potassium application had no effect on total root length. It is possible that roots obtained some K from deeper in the soil profile, and also from less readily-available K pools, to maintain root growth in the zero-K treatment. Application of Si decreased total root length to a depth of 60 cm, significantly so in the Inceptisol. In addition, total root length in the Oxisol was less than in the Inceptisol. Both of these responses may have been due to water stress.

COLLEGE OF AGRICULTURE, ENGINEERING AND SCIENCE

DECLARATION 1 - PLAGIARISM

I, Ruth Rhodes, declare that

- The research reported in this thesis, except where otherwise indicated, is my original research.
- This thesis has not been submitted for any degree or examination at any other university.
- This thesis does not contain other persons' data, pictures, graphs or other information, unless specifically acknowledged as being sourced from other persons.
- 4. This thesis does not contain other persons' writing, unless specifically acknowledged as being sourced from other researchers. Where other written sources have been quoted, then:
 - a. Their words have been re-written but the general information attributed to them has been referenced.
 - b. Where their exact words have been used, then their writing has been placed in italics and inside quotation marks, and referenced.
- This thesis does not contain text, graphics or tables copied and pasted from the Internet, unless specifically acknowledged, and the source being detailed in the thesis and in the References.

Signed:	Candidate name: R. Rhodes	Date: 25 Oct 2020
As the candidate's supervisor I hav	e/have not approved this dissertation	for submission
Signed: .	Name: Prof. J.C. Hughes	Date: 22 Oct 2020
As the candidate's co-supervisor I have/have not approved this dissertation for submission		

-

Signed: .

Name: Dr. N. Miles

Date: 18/10/20

DECLARATION 2 – PUBLICATION

Rhodes R, Miles N, Hughes JC. 2018. Interactions between potassium, calcium and magnesium in sugarcane grown on two contrasting soils in South Africa. *Field Crops Research* 223: 1-11.

Name: R Rhodes Date: 25 October 2020 Signed: ..

ACKNOWLEDGEMENTS

To Almighty God, for His never-ending grace, and every opportunity I have ever had.

To Will, Olly and Katie – all three born during my thesis! - for doing without Mommy during those long nights (and for putting up with tired, grumpy Mommy next day), for being my inspiration to get it done, and for filling my heart with everything good.

To my supervisors! – Prof and Neil... for your supreme patience, gentle reminders (Prof), general abuse (Neil), clever ideas, amazingly thorough editing, help with funding, laughs and shared frustrations – I remain forever in your debt.

To Louis Titshall for initial comments on my study proposal, advice on possible trial designs and initial field site visits.

To SASRI – for the time, funding, opportunities and a superb place to work.

To growers Quinn Klusener (and later Pete Harris... and also Brett Williams), John (and later Myles) Osborn, as well as extension specialists Rowan Stranack, and later Oom Aderjan Naude, for your enthusiasm, willingness to help, fieldside chats, laughs and practical wisdom.

To my colleagues, who were always more than 'just' that – Nerd Shaun Berry, Tannie Peta Campbell, Peter Lyne – for your visits, high-fives (Peter), badgering (Shaun), 'How's it going?' (Tannie) – because you all knew my pain - and most of all, for your friendship.

To the SASRI Soils team – no better colleagues! – for your constant help (Wiseman!, Dimpho), encouragement, analyses (Andre, Gayshree), cheerfulness and friendship.

To Nikki – gorgeous AND clever... for helping to make sense of my numbers!

To Nanda (and later Sid), and the Technical Team (especially Kathree, Havish, Sipho-sethu, Suran, Prince, Linda, Musa...), and Moses (and Willard) for your field guys (especially Chocolate, Tony, Boyzo...) – for your sustained hard work - too much to mention.

To Ingrid – for those deep chats, arranged marriages ;-), laughs and fashion fabulosity.

To the Mexican Meisies! – for laughter, friendship, wisdom, help, goss and skande babas.

To my girls... for remembering to ask how it was going, even after so many years.

To Dollface Alana – for the chats, laughs and tears, and for picking me up, shaking me off and (once) shouting at me to get myself into gear!

To COVID-19! – for giving me a bit of space during lockdown, April 2020, to give a big push. Also to Tam for being here then ☺.

To Dear Sharon, the only one who really, really understood how it was going, because it was going the same way for you... for your endless wisdom, mentorship, encouragement, acceptance, understanding, listening ear and laughs.

To RRRROP! – for the shared papers, for your encouragement (without being too pushy), your thoughtful questions, for Chloe, and for the motivation of seeing you graduate in your red cape!

To Mom & Dad – for making me ☺, and for the time, space, support and endless "So when is your next deadline?" questions Dad!

And to Grant... for being my rock, my protector, my safety belt, my "Do the kids so I can study", for ALWAYS being in my corner, for letting me sleep in, for putting up with my endless fossicking, for allowing me to spend so many years (and so much money) pursuing this... my words could never do justice to what you mean to me.

THANK YOU

TABLE OF CONTENTS

	i
Declaration 1: Plagiarism	iv
Declaration 2: Publication	v
Acknowledgements	vi
Table of Contents	viii
List of Figures	xiv
List of Tables	xix
List of Appendices	xxii
Glossary	xxiv
CHAPTER 1: Introduction	
1.1. Sugarcane industry overview	
1.1.1. Location	
1.1.2. Economic contribution	
1.1.3. Industry support	
1.1.4. Soils and soil testing	
1.2. Sugarcane crop nutrition	
1.3. Purpose and outline of the study	
1.3.1. Aims and objectives	
1.3.2. Key questions	
1.3.3. Dissertation structure and chapter outline	5
CHAPTER 2: Potassium, nitrogen and silicon in soils and plants with special	
reference to sugarcane.	6
2.1. Introduction	6
	······································
2.2. Potassium	
2.2. Potassium	
2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops	
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition	
 2.2. Potassium	
 2.2. Potassium	
 2.2. Potassium	7
 2.2. Potassium	
 2.2. Potassium	7 7 9 10 10 11 11 11 12
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition 2.2.3.1. Consequences of excess potassium in sugarcane 2.2.3.2. Potassium and moisture stress 2.2.4.1. Potassium interactions with calcium and magnesium 2.2.4.2. Potassium interactions with nitrogen 2.2.4.3. Potassium interactions with aluminium and manganese. 	7 7 9 10 10 11 11 11 11 12 12
 2.2. Potassium	7 7 9 10 10 11 11 11 11 12 12 12 13
 2.2. Potassium	7 7 9 10 10 11 11 11 11 12 12 12 13 14
 2.2. Potassium	7 7 9 10 10 10 10 11 11 11 12 12 12 12 13 14 14
 2.2. Potassium in soils	7 7 9 10 10 11 11 11 12 12 12 12 12 12 12 12 14 14 14
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition 2.2.3.1. Consequences of excess potassium in sugarcane 2.2.3.2. Potassium and moisture stress 2.2.4. Nutrient interactions 2.2.4.1. Potassium interactions with calcium and magnesium 2.2.4.2. Potassium interactions with nitrogen 2.2.4.3. Potassium interactions with aluminium and magnese 2.2.5. Soil analysis and plant uptake 2.3.1. Nitrogen in sugarcane nutrition. 2.3.2.1. Nitrogen interactions with phosphorus and potassium 	7 7 9 10 10 10 11 11 11 12 12 12 12 12 13 14 14 14 14 16 16
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition 2.2.3.1. Consequences of excess potassium in sugarcane 2.2.3.2. Potassium and moisture stress 2.2.4.1. Potassium interactions with calcium and magnesium 2.2.4.2. Potassium interactions with nitrogen 2.2.4.3. Potassium interactions with aluminium and manganese 2.2.5. Soil analysis and plant uptake 2.3.1. Nitrogen in sugarcane nutrition 2.3.2.1. Nitrogen interactions with phosphorus and potassium 2.3.2.2. Nitrogen interactions with silicon/lime 	7 7 9 10 10 11 11 11 12 12 12 12 12 12 12 12 13 14 14 14 16 16 16
 2.2. Potassium in soils	7 7 9 10 10 11 11 11 11 12 12 12 12 13 14 14 14 14 16 16 16 16 17
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition 2.2.3.1. Consequences of excess potassium in sugarcane 2.2.3.2. Potassium and moisture stress 2.2.4. Nutrient interactions. 2.2.4.1. Potassium interactions with calcium and magnesium 2.2.4.2. Potassium interactions with nitrogen 2.2.4.3. Potassium interactions with aluminium and manganese. 2.2.5. Soil analysis and plant uptake 2.3.1. Nitrogen in sugarcane nutrition. 2.3.2.1. Nitrogen interactions with phosphorus and potassium. 2.3.2.2. Nitrogen interactions with silicon/lime. 2.4.1. Silicon in soils 	7 7 9 10 10 10 11 11 11 12 12 12 12 12 12 12 13 14 14 14 14 16 16 16 16 17
 2.2. Potassium 2.2.1. Potassium in soils 2.2.2. Potassium supply to crops 2.2.3. Potassium in sugarcane nutrition 2.2.3.1. Consequences of excess potassium in sugarcane 2.2.3.2. Potassium and moisture stress 2.2.4. Nutrient interactions. 2.2.4.1. Potassium interactions with calcium and magnesium 2.2.4.2. Potassium interactions with nitrogen 2.2.4.3. Potassium interactions with aluminium and manganese. 2.2.5. Soil analysis and plant uptake 2.3.1. Nitrogen in sugarcane nutrition. 2.3.2.1. Nitrogen interactions with phosphorus and potassium. 2.3.2.2. Nitrogen interactions with silicon/lime. 2.4.1. Silicon 2.4.2. Plant availability and uptake of silicon 	7 7 9 10 10 11 11 11 11 12 12 12 12 12 12 12 12 12

2.4.4. Nutrient interactions	22
2.4.4.1. Silicon interactions with potassium	22
2.4.4.2. Silicon interactions with nitrogen	22
2.4.4.3. Silicon interactions with phosphorus	23
2.4.4.4. Silicon interactions with calcium and magnesium	23
2.5. Nutrient recommendations	24
2.5.1. Potassium	24
2.5.2. Nitrogen	24
2.5.3. Silicon	25
2.6. The thesis in context: Gap analysis and motivation for the study	25
2.6.1. Potassium	26
2.6.2. Nitrogen	26
2.6.3 Silicon	27
OUADTED O. The offerst of surglinestic method of methods in the surge of the surge	
CHAPTER 3: The effect of application rates of potassium, nitrogen and silicon on	20
2 1 Introduction	29 20
3.2 Materials and methods	29 20
3.2. Materials and methods	30 20
3.2.1. That sites	30 21
3.2.1.1. Railliai	22
3.2.2. Join Sampling	33 22
3.2.4 Trial harvost	35
3.2.5. Statistical analysis	35 36
3.2.6. Economic analysis	30 36
3.3 Results and discussion	
3.3.1 Initial soil conditions	30 36
3.3.2 Effect of potassium on sugarcane barvest characteristics	37
3.3.2.1 Oxisol	07
3.3.2.1.1. Stalk vield	07
33212 Sucrose percent	07
3.3.2.1.3. Sucrose vield	39
3.3.2.2. Inceptisol	
3.3.2.2.1. Stalk vield	
3.3.2.2.2. Sucrose percent and vield	40
3.3.2.3. Economic aspects	41
3.3.3. Effect of nitrogen on sugarcane harvest characteristics	43
3.3.3.1. Oxisol	43
3.3.3.1.1. Stalk yield	43
3.3.3.1.2. Sucrose percent	44
3.3.3.1.3. Sucrose vield	46
3.3.3.2. Inceptisol	46
3.3.3.2.1. Stalk yield	46
3.3.3.2.2. Sucrose percent	47
3.3.3.2.3. Sucrose yield	48
3.3.3.3. Economic aspects	48
3.3.4. Effect of silicon on sugarcane harvest characteristics	52
3.3.4.1. Oxisol	52

3.3.4.1.1. Stalk yield, sucrose percent and sucrose yield	. 52
3.3.4.2. Inceptisol	. 53
3.3.4.2.1. Stalk yield, sucrose percent and sucrose yield	. 53
3.3.4.3. Economic aspects	. 55
3.3.5. Effect of nutrient interactions on sugarcane harvest characteristics	. 57
3.3.5.1. Oxisol	. 57
3.3.5.2. Inceptisol	. 58
3.4. Conclusions	. 59
CHAPTER 4: The effect of applied potassium, nitrogen and silicon on leaf nutrient	
concentrations and relationships with sugarcane yield	. 61
4.1. Introduction	. 61
4.2. Materials and methods	. 62
4.2.1. Trial details	. 62
4.2.2. Leaf sampling	. 62
4.2.3. Statistical analysis	. 63
4.3. Results and discussion	. 64
4.3.1. Effect of applied potassium on leaf nutrient content	. 64
4.3.1.1. Potassium	. 64
4.3.1.2. Sulphur, zinc and copper	. 65
4.3.1.3. Calcium and magnesium	. 65
4.3.1.4. Sum of cations	. 69
4.3.2. Effect of applied nitrogen on leaf nutrient content	. 71
4.3.2.1. Nitrogen	. 71
4.3.2.2. Phosphorus, potassium, sulphur, zinc, copper and silicon	. 71
4.3.2.3. Calcium and magnesium	. 73
4.3.3. Effect of applied silicates on leaf nutrient content	. 74
4.3.3.1. Silicon	. 74
4.3.3.2. Nitrogen and phosphorus	. 76
4.3.3.3. Calcium, magnesium and potassium	. 76
4.3.4. Leaf potassium, nitrogen and silicon status and crop yield	. 78
4.3.4.1. Potassium	. 78
4.3.4.2. Nitrogen	. 80
4.3.4.3. Silicon	. 81
4.4. Conclusions	. 82
CHAPTER 5. The effect of notassium, nitrogen and silicon application on	
relationships between soil nutrient dynamics and yield of sugarcane.	. 84
5.1. Introduction	. 84
5.2. Materials and methods	. 86
5.2.1. Trial details and pre-trial soil sampling	. 86
5.2.2. Topsoil and depth sampling	. 86
5.2.3. Chemical analysis	. 86
5.2.4. Mineralogy and total potassium	. 87
5.2.5. Slowly-available potassium	. 88
5.2.6. Statistical analysis	. 88
5.2.7. Potassium balance	. 88
5.3. Results and discussion	. 89

5.3.1. Initial soil properties	89
5.3.2. Potassium	89
5.3.2.1. Mineralogy and total potassium content	89
5.3.2.2. Effect of potassium application on topsoil (0-20 cm) readily-avail	able
potassium	91
5.3.2.3. Effect of potassium application on mean soil profile characteristic	CS
following harvests	94
5.3.2.4. Potassium movement down the profile	95
5.3.2.5. Slowly-available (nitric) potassium	97
5.3.2.6. Potassium balance	98
5.3.2.7. Effect of topsoil potassium on sugarcane yield and quality	100
5.3.2.8. Inclusion of subsoil potassium data into topsoil calibration	104
5.3.3. Nitrogen	106
5.3.4. Silicon	106
5.3.4.1. Effect of silicate application on topsoil (0-20 cm) properties	106
5.3.4.2. Effect of silicate application on mean soil profile characteristics	
following harvest	107
5.3.4.3. Silicon, calcium and magnesium movement down the profile	108
5.3.4.4. Effect of topsoil silicon on sugarcane yield and quality	110
5.3.4.5. Inclusion of subsoil silicon data into topsoil calibration	111
5.4. Conclusions	112
5.4.1. Potassium	112
542 Nitrogen	113
5.4.3. Silicon	114
5.4.3. Silicon	114
5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs	114 ne 115
5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and th effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115
5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and th effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction 6.2. Materials and methods	114 ne 115 115 116 116
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction 6.2. Materials and methods 6.2.1. Trial details 6.2.2. Incremental soil depth and topsoil (0-20 cm) sampling 6.2.3. Statistical analysis	114 ne 115 115 116 116 116 117
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 117 118 118
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118 118 118
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118 118 118 118 121
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 118 118 118 118 121 122
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118 118 121 122 122
 5.4.3. Silicon	114 ne 115 115 116 116 116 116 118 118 118 118 121 122 122 125
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118 118 118 118 121 121 122 125 126
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118 118 118 118 121 121 122 125 126
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118 118 121 122 125 126
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118 118 121 122 122 125 126
 5.4.3. Silicon CHAPTER 6: Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs 6.1. Introduction	114 ne 115 115 116 116 116 117 118 118 118 121 122 122 125 126 128 128
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118 118 121 122 122 125 126 128 128 129
 5.4.3. Silicon	114 ne 115 115 116 116 116 117 118 118 118 118 121 122 122 125 126 128 129 129

7.2.3. Statistical analysis	130
7.3. Results and discussion	130
7.3.1. Profile root distribution	130
7.3.2. Effect of soil type and applied nutrients on total root length	132
7.4. Conclusions	136
CHAPTER 8: Potassium exhaustion study of an Oxisol and an Inceptisol, using	Italian
ryegrass (<i>Lolium multiflorum</i> L.)	137
8.1. Introduction	137
8.2. Materials and methods	138
8.2.1. Pot trial procedure	138
8.2.2. Data analysis	140
8.3. Results and discussion	140
8.3.1. Ryegrass uptake of potassium and potassium balance	141
8.3.2. Soil potassium release: comparison between pot and field potassi	um
balances	145
8.3.3. Long-term potassium supply	146
8.4. Conclusions	149
CHAPTER 9: A summary conceptual model, general discussion, conclusions a	nd
future work	150
9.1. Introduction	150
9.2. Conceptual model: Effects of potassium, nitrogen and slag-based silic	ate
fertiliser on sugarcane yield and quality	150
9.3. Nutrient recommendations	153
9.3.1. Potassium	153
9.3.2. Nitrogen	154
9.3.3. Silicon	154
9.3.4. Nutrient interactions	155
9.3.4.1. Nitrogen and silicon	155
9.3.4.2. Potassium and silicon	156
9.3.4.3. Nitrogen, silicon and potassium	156
9.3.5. Sampling and crop nutrition	156
9.3.6. Rooting	157
9.4. Future work	157
9.4.1. Potassium	157
9.4.1.1. Subsoil	157
9.4.1.2. Non-exchangeable potassium release	158
9.4.2. Nitrogen	159
9.4.3. Silicon	159
9.4.3.1. Silicon uptake	159
9.4.3.2. Silicon thresholds	160
9.4.3.3. Cationic antagonism	160
9.4.4. Nutrient interactions	161
9.4.5. General	161
9.4.5.1. Inclusion of subsoil data into nutrient recommendations	161
9.4.5.2. Leaf sampling	161
9.4.5.3. Rooting	162

REFERENCES	163
APPENDICES	184

LIST OF FIGURES

Figure 1.1: Location of the sugarcane-growing regions and sugar mills of South Africa. (Image: S. Doorsamy, South African Sugarcane Research Institute)......1

Figure 2.1: Kinetic reactions of potassium (K) in soil, where: a = adsorption, b = desorption, c = non-exchangeable fixation, d = non-exchangeable release, e = immobilisation, and f = mineralisation (adapted from Selim *et al.*, 1976)......8

Figure 3.7: Sucrose yield (t ERC ha⁻¹) response to the 200 kg K ha⁻¹ application rate as a percentage of the 0 kg K ha⁻¹ control for the three crops on the Oxisol and the Inceptisol. n=24 in all crops except Inceptisol second ratoon, where n=16.......43

Figure 5.5: Oxisol: Mean (n=24) sugarcane (A) stalk and (B) sucrose yield, as a percentage of the maximum yield per harvest in relation to readily-available topsoil potassium (K). For the second ratoon crop in B, points with different letters are significantly different to each other (P < 0.05). Each point is shaded to represent one of the four K application rates (0 kg K ha⁻¹: clear symbol; 100 kg K ha⁻¹: grey symbol; 200 kg K ha⁻¹: hatched symbol; 300 kg K ha⁻¹: black symbol)......101

Figure 6.1: Readily-available soil potassium (K) across both rates of applied silicon (0 and 300 kg ha⁻¹), in the (A) Oxisol and (B) Inceptisol, at four K application rates, following fertiliser application at the start of the second ratoon crop. Within each soil, points with different letters are significantly different to each other (P < 0.001)....120

Figure 6.3: Inceptisol: Extractable soil silicon (Si) at two Si application rates following fertiliser application at the start of the second ratoon crop (n=4). Points with different letters are significantly different to each other (P = 0.001)......122

Figure 8.1: Ryegrass leaf potassium (K) and cumulative K removed in harvested leaves over time on the (A) Oxisol and (B) Inceptisol......141

LIST OF TABLES

Table 4.2: Oxisol: Effect of applied potassium (K) on mean \pm s.e. (n=24) third-leaf calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age......67

Table 4.3: Inceptisol: Effect of applied potassium (K) on mean ± s.e. (n=24) third-leaf calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age......68

Table 4.4: Effect of applied potassium (K) on the mean \pm s.e. (n=16) third-leaf sum of potassium, calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age, on the Oxisol and Inceptisol. Cation sum calculated for the 0 kg Si ha⁻¹ and intermediate N rate (80 kg ha⁻¹ [Oxisol]; 105 kg ha⁻¹ [Inceptisol]) treatments......70

Table 4.5: Effect of applied nitrogen (N) on mean \pm s.e. (n = 32) third-leaf nutrient status on the (A) Oxisol and (B) Inceptisol, each at one sampling event......72

Table 4.7: Effect of applied Calmasil® / cement on mean ± s.e. (n=48) third-leaf silicon (Si) content, at 4- and 6-months' crop age, on the Oxisol and Inceptisol.....75

Table 5.6: Mean readily-available topsoil (0-20 cm) potassium (K) (n=27), mean yield (n=27) in harvested stalks and mean K percent in stalks (n=27) per season, and total K removed (n=36), as a sum over the three crops at each trial site......103

Table 8.2: Ryegrass leaf dry matter at each harvest on the Oxisol andInceptisol142

 Table 8.4:
 Potassium balance in the ryegrass pot trial on the Oxisol and Inceptisol

 143

LIST OF APPENDICES

Appendix 5.1: Effect of nitrogen application on topsoil (0-20 cm) properties189

Appendix 5.3: Nitrogen movement down the profile......191

Appendix 5.4: Nitrogen: Conclusions......191

GLOSSARY

Explanation of sugarcane-related terms used in the thesis

FAS (Fertiliser Advisory Service): User-pays analytical laboratory, situated at SASRI, which conducts analyses on soil, plants, water and nutrient sources. The FAS provides nutrient recommendations based on soil test results. Situated at Mount Edgecombe, near Durban.

Furrows: shallow parallel trenches drawn into the soil, into which seedcane setts are placed. Some fertiliser is placed into the furrow, and the furrow is then closed by drawing soil over the sugarcane setts.

Interrow: the space between the rows of sugarcane.

Plant crop: the first crop of sugarcane, grown after planting sugarcane setts.

Ratoon crop: after harvesting a sugarcane crop, a new crop will resprout without replanting new sett material. This resprouted crop is called a ratoon crop. Ratoon crops are named successively, so that a field will typically grow a plant crop, first ratoon crop, second ratoon crop and so on.

Row, Row spacing: sugarcane is planted in rows; 'row' refers to the actual row of sugarcane, while row spacing refers to the distance between rows of sugarcane. Common row spacing in the South African sugar industry ranges from 0.9 m (rainfed) to 1.5 m (irrigated).

SASA (South African Sugar Association): an organisation which provides support to the growers and millers of the South African sugar industry and thereby works to sustain the industry's global competitiveness, profitability and sustainability. Under SASA's auspices, a number of divisions operate to provide specialist support to these clients. Head office is situated at Mount Edgecombe, near Durban.

SASRI (South African Sugarcane Research Institute): a division of SASA which conducts scientific research on sugarcane agriculture and provides specialist advice and support to sugarcane growers. Head office is situated at Mount Edgecombe, near Durban, while satellite stations are located throughout the industry.

Stool: sugarcane plant or clump. Sugarcane does not grow in uniform, one-by-one tillers spaced evenly within a row; rather, there are clumps of tillers or 'stools' originating from a bud on the planted sugarcane sett, from which a number of tillers emerge.

Sugarcane setts */* **seedcane:** short sticks of sugarcane, planted into furrows in the soil to vegetatively propagate a new crop of sugarcane.

Tin and string method: Method of fertiliser application. A standard-sized tin is filled with fertiliser and weighed. Depending on the sugarcane row spacing and fertiliser rate required, a distance (length of row) can be calculated over which the full tin should be distributed, and a piece of string measured and cut. This length of string is used by the field supervisor to indicate the distance to the fertiliser application team.

CHAPTER 1

Introduction

1.1. Sugarcane industry overview

1.1.1. Location

The South African sugarcane industry is located along the eastern seaboard of the country. Most of the cane is grown in eastern KwaZulu-Natal and Mpumalanga, with a small portion in the Eastern Cape (Figure 1.1). The northern part is irrigated, with the remainder largely rainfed. Fourteen mills service the industry (Figure 1.1).



Figure 1.1: Location of the sugarcane-growing regions and sugar mills of South Africa. (Image: S. Doorsamy, South African Sugarcane Research Institute).

1.1.2. Economic contribution

An area of approximately 380 000 ha is currently under sugarcane (*Saccharum officinarum* x *S. spontaneum* hybrid), of which about 270 000 ha is harvested

annually to produce an average of 2.2 million tonnes of sugar per year. South African sugar sales generate an annual average direct income of R14 billion (US\$966 million¹) (SASA, 2020), representing approximately 20% of agriculture, forestry and fishing's contribution towards South Africa's gross domestic product (Stats SA, 2019).

About 85 000 people are directly employed in the production, transport, milling, refining and by-product sectors of the sugar industry. Indirect employment is estimated at 350 000 people, with a total of approximately one million, including dependents, relying on the industry for their livelihood (SASA, 2015).

1.1.3. Industry support

South Africa's sugarcane growers and millers are supported by the South African Sugar Association (SASA) that works to sustain the industry's global competitiveness, profitability and sustainability. Under SASA's auspices, a number of divisions operate to provide specialist support to clients. One such division is the South African Sugarcane Research Institute (SASRI), which conducts scientific research on sugarcane agriculture and provides specialist advice to sugarcane growers on, amongst others, soil and crop nutrition management strategies.

1.1.4. Soils and soil testing

South African sugarbelt soils are characterised by considerable heterogeneity, with large differences in, *inter alia*, depth, base status and texture (SASEX, 1999). Such variation provides challenges in developing accurate, field-specific nutrient recommendations. Soil testing is thus strongly encouraged throughout the industry.

In order for soil analyses to be of practical use in terms of the derivation of nutrient recommendations, laboratory tests need to be calibrated with crop responses in the field. A commercial laboratory, the Fertiliser Advisory Service (FAS), was established at SASRI in 1954 to analyse soil, plant, water and fertiliser samples from the sugar industry. The FAS makes nutrient recommendations based on the results of

¹ Exchange rate of ZAR1 = 0.069 US\$, 28 January 2020

hundreds of field trials conducted during the past 90 years. Trial work, of which the present study forms a part, is ongoing in order to improve nutrient recommendations.

1.2. Sugarcane crop nutrition

The three nutrients required in the largest amounts by sugarcane are potassium (K), nitrogen (N) and silicon (Si), and they have received significant research attention. Potassium fulfils a number of metabolic functions in sugarcane, including anionic charge balance and turgor control, and thus plays an important role in the growth, metabolism and ripening of the crop (Kingston, 2000; Wood and Schroeder, 2004). Potassium reserves in many topsoils in the rainfed sugarcane areas of South Africa are low (Miles and Farina, 2014) and more accurate estimates of the K requirements of sugarcane are required. Potassium's interaction with other nutrients such as N (Miles, 2010), calcium (Ca) and magnesium (Mg) (Gosnell and Long, 1971; Grunes *et al.*, 1992; Marschner, 1995) makes understanding its patterns of uptake essential.

Nitrogen plays a key role in protein synthesis and is associated with vigorous vegetative growth (Anderson and Bowen, 1990). It continues to be investigated due to its pivotal role in sugarcane growth and yield and its characteristic as a 'driver' of the uptake of other nutrients (Miles, 2010).

The importance of Si has only been recognised by sugarcane researchers recently (Laing *et al.*, 2006). Although not considered 'essential' for sugarcane growth, Si plays an important role in protecting the crop from both biotic and abiotic stress. Research has highlighted the extensive Si deficiency of sugarbelt soils in the rainfed parts of the country (Van der Laan and Miles, 2010) and further research is required to understand its uptake by, and effects on, sugarcane.

While the individual effects of nutrients on sugarcane uptake and performance are important and, in some cases, well documented, the interactions between nutrients are less well understood. South African sugarcane growers spend millions of rands each year on fertilisers for their crop, and it is essential that the nutrient recommendations provided by SASRI, via the FAS, are as appropriate as possible.

1.3. Purpose and outline of the study 1.3.1. Aims and objectives

This study was designed to examine, on two contrasting soil types that are representative of common soils of the sugarbelt, the effect of K, N and Si treatments on sugarcane growth and quality. Also investigated were nutrient interactions, nutrient balances in the soil and crop, and rooting characteristics. The ultimate goal was to assist in refinement of the industry nutrient recommendations given by the FAS.

1.3.2. Key questions

In particular, the following questions were addressed in this study, approved by SASRI as Project Number: 08RE05.

- a) How do applied K, N and Si, and their interactions, influence sugarcane crop yield and quality?
- b) How do different K, N and Si application rates affect plant uptake of these and other nutrients, and how is plant uptake related to crop yield and quality?
- c) Does leaf analysis provide an accurate estimation of nutrient sufficiency or deficiency indicating the necessity of further nutrient application for optimal yield?
- d) How do applied K, N and Si influence soil concentrations of these and other nutrients?
- e) What is the soil profile distribution of K, N and Si following fertiliser application?
- f) How does the movement of applied K, N and Si through the soil profile affect soil sampling results and hence fertiliser recommendations and costs?
- g) Do different rates of K and Si application affect root distribution patterns?
- h) What are the underlying reasons for any differences in crop behaviour on different soils?
- i) What are the implications of the above for K, N and Si fertiliser recommendations in the South African sugarcane industry?

1.3.3. Dissertation structure and chapter outline

This dissertation contains nine chapters, the latter seven of which address the key questions identified above, as follows:

- Chapter 2 is a literature review of K, N and Si research in sugarcane soils.
- Chapter 3 gives details of the field experiments and examines the effects of different rates of K, N and Si, and their interactions, on sugarcane quality and yield.
- Chapter 4 reports the effect of K, N and Si on leaf nutrient concentrations and yield of sugarcane.
- Chapter 5 discusses the effect of K, N and Si application on topsoil and subsoil nutrient dynamics.
- Chapter 6 examines the movement and distribution of applied K, N and Si through the topsoil, and the possible effects on subsequent soil sample results and fertiliser costs.
- Chapter 7 investigates the effects of applied K, N and Si on root distribution throughout the soil profile.
- Chapter 8 reports on a K-exhaustion pot trial, using ryegrass (*Lolium multiflorum* L.) as an indicator crop, to assess the soils' K-supplying power in the absence of K fertilisation. This was carried out to investigate the limited yield response to applied K found on both soils in the field trials.
- Chapter 9 comprises a summary conceptual model, further discussion, conclusions, and proposed future work.

CHAPTER 2

Potassium, nitrogen and silicon in soils and plants with special reference to sugarcane

2.1. Introduction

Modern sugarcane production began in Tahiti in 1768, and the crop is now grown in over 100 countries, occupying more than 20 million hectares of land (Meyer and St John Clowes, 2013). It is the eighth most planted crop in the world, after wheat (*Triticum aestivum*), maize (*Zea mays*), rice (*Oryza sativa*), soybeans (*Glycine max*), beans (*Phaseolus* spp), barley (*Hordeum vulgare*) and rapeseed (*Brassica napus*), for area harvested (Desjardins, 2014). As a provider of income and employment, sugarcane agriculture plays an important role, particularly in developing countries, which produce approximately 75% of the sugar derived from sugarcane (Voora *et al.*, 2019).

Sugarcane was first grown commercially in South Africa in 1848 and crop nutrition began to receive scientific attention following the establishment of the South African Sugar Association Experiment Station in 1925 (renamed the South African Sugarcane Research Institute (SASRI) in 2004). Considerable research emphasis has been placed on the three major nutrients - nitrogen (N), phosphorus (P) and potassium (K) - and their role in sugarcane production. Of these, N and K are required in the greatest amounts (De Oliveira *et al.*, 2017), and understanding their behaviour in soils is important in order to make informed fertiliser recommendations. In addition, silicon (Si) has received much research attention recently (De Camargo *et al.*, 2014; Haynes, 2014, 2017; Miles *et al.*, 2014).

The dynamics of K, N and Si in both soil and plants will be discussed in this review, with a particular focus on sugarcane. In addition, the interaction between these nutrients will be discussed, following the observation by Sumner and Farina (1986) that the interplay between nutrients in field cropping systems is seldom given the attention it deserves.

2.2. Potassium

Potassium is the cation required in the largest amounts by most plants (Askegaard *et al.*, 2004). It performs a number of biochemical functions in plants such as activating enzymes used, for example, in protein synthesis, although for these roles only small amounts are required. Larger amounts of K are needed for its biophysical roles in osmoregulation, cation-anion balance and water balance (Askegaard *et al.*, 2004). Potassium plays a major role in solute movement within the plant and, of particular interest to crop farmers, the movement of photosynthates to storage organs such as grains and tubers (Askegaard *et al.*, 2004). It is also associated with protection of the plant against abiotic (frost, drought, heat, salinity) and biotic (fungal and pest) stresses (Huber and Arny, 1985; Askegaard *et al.*, 2004; Ashraf *et al.*, 2009).

In unamended systems, the main source of K for plants is from the weathering of soil minerals (Mengel and Kirkby, 2001). Despite the presence of large amounts of K in most soils, only a small portion is readily available to plants (Sparks and Huang, 1985); the rest is present in slowly available forms, released over longer time spans. As a plant nutrient, the importance of K is easily overlooked. The ability of soils to buffer K supply can mask their true K status (Krauss, 2001). Potassium can be washed out of senescent leaves, recycling it from the whole of the root zone into the topsoil and this can mask a decrease in overall soil K status if only the topsoil is sampled (Ritchey, 1979). The effects of K on crop growth are often subtle and, in contrast to the rapid, easily-visible crop response to applied N, for example, are often only apparent at harvest, as a function of yield (Krauss, 2001). Other less obvious benefits associated with K nutrition, such as protection against pests, diseases or climatic stress, are not often attributed to K (Krauss, 2001).

2.2.1. Potassium in soils

Potassium exists in four different forms or 'pools' in soils namely, K in solution, exchangeable K, 'fixed' (non-exchangeable) K, and structural or mineral K (Figure 2.1) and this order also reflects their ease of availability to plants and microbes (Sparks and Huang, 1985). The reactions between the four fractions of soil K (Figure 2.1) affect the concentration of K in soil solution at any one time (Sparks, 2001).



Figure 2.1: Kinetic reactions of potassium (K) in soil, where: a = adsorption, b = desorption, c = non-exchangeable fixation, d = non-exchangeable release, e = immobilisation, and f = mineralisation (adapted from Selim *et al.*, 1976).

Reactions **a** and **b** (Figure 2.1) – adsorption and desorption – result in the transfer of exchangeable K from the soil solution to the surfaces of clay minerals and humic substances, and *vice versa*. The reaction rate between soil solution and exchangeable pools of K depends strongly on the type of clay minerals present (Sparks, 2001).

Reactions **c** and **d** – non-exchangeable fixation and release – significantly affect K availability (Sparks, 2001). Potassium may be 'fixed' or held within mineral layers; this non-exchangeable (or 'slowly exchangeable') K can normally be released again over a period of a season or year/s (Askegaard *et al.*, 2004). The concentration of solution K markedly affects the release of non-exchangeable K from clays (Sparks and Huang, 1985). Release of non-exchangeable K to the exchangeable form occurs when amounts of exchangeable and soil solution K are decreased by crop removal, leaching and perhaps by high microbial activity (Sparks, 2000).

The degree of K fixation in clays and soils depends on the type of clay mineral and its charge density. The major clay minerals responsible for K fixation are the expanding 2:1 types, including montmorillonite and vermiculite, and weathered micas (Rich, 1968). Potassium fixation is also affected by soil moisture content, competing cations within the soil (Sparks and Huang, 1985) and soil pH (Martin *et al.*, 1946; Rich and co-workers, summarised by Sparks and Huang, 1985). The degree of interlayering can also affect K fixation, with hydroxy-aluminium and -iron interlayer groups serving to decrease K fixation (Rich and Obenshaim, 1955; Rich and Black, 1964).

Reactions **e** (immobilisation) and **f** (mineralisation) occur between the nonexchangeable and mineral pools of K. Micas and feldspars are the most important mineral sources of K in soils. The release or mineralisation (reaction **f**) of lattice K is considered by some to be irreversible (Askegaard *et al.*, 2004), indicating that, in reality, reaction **e** does not occur to a significant extent.

2.2.2. Potassium supply to crops

Potassium moves between the four pools (Figure 2.1) in response to soil conditions. With weathering, mineral K is converted into non-exchangeable and/or more readily-available forms (Kingston, 2000). Non-exchangeable, exchangeable and soil solution K are in dynamic equilibrium with each other. Therefore, any losses (e.g. leaching or crop uptake) of readily-available K may be replenished from the less available sources (Kingston, 2000). This so-called 'replenishing power' can be substantial, depending on the soil. Nonetheless, continued depletion of soil K reserves is unsustainable in the long term (Kingston, 2000) and it is not possible to continue 'mining' soil K without eventually decreasing crop production (Askegaard *et al.*, 2004). Replenishment of K is thus necessary, particularly for sugarcane, where luxury uptake is common (Kingston, 2000).

Plant roots take up K from the soil solution in the form of the K^+ ion (Marschner, 1995). Depletion of K at the root surface sets up a gradient between the low concentrations (at the root surface) and the higher concentrations in the bulk soil, leading to diffusion of K towards the root surface (Jungk and Claassen, 1989, cited in Askegaard *et al.*, 2004). A high rate of diffusion can be facilitated by applying K to the soil, or by the release of K from other fractions (Askegaard *et al.*, 2004). The buffering power of a soil is related to clay content (Sharpley, 1990) and clay mineralogy (Mengel and Busch, 1982). In the short and medium term, the extent to which the solution and exchangeable K pools are maintained will depend on the amount of K added as fertiliser which has been retained in the exchangeable and non- (or slowly) exchangeable fractions (Askegaard *et al.*, 2004). In the longer term, in soils with limited K inputs, the release of mineral K becomes an important source for plants (Askegaard *et al.*, 2004).

2.2.3. Potassium in sugarcane nutrition

Sugarcane has a high demand for K and over 300 kg K ha⁻¹ can be taken up in a single crop (Meyer, 2013). In sugarcane, K is required for starch and protein synthesis, photosynthesis, osmoregulation, electron balance in cells, and stimulation of phloem transport of sugars, amongst other functions (Kingston, 2000; Wood and Schroeder, 2004).

Low soil K can inhibit sugarcane germination (Anderson and Bowen, 1990), although ratoon crops respond more to added K than do plant crops (Du Toit, 1957; Wood and Meyer, 1986). Potassium deficiencies may also lead to a decrease in cane yield and quality (Wood and Schroeder, 2004). Although K deficiency symptoms may be apparent in sugarcane, yield depression, even where symptoms appear, may not be marked (Wood and Schroeder, 2004), possibly due to replenishment of soil solution and exchangeable K from the non-exchangeable and mineral pools.

Worldwide, it is estimated that approximately 31.5 million tons of fertiliser K was applied to all crops in 2019 (FAO, 2019). In South Africa, approximately 100 000 tons of K were applied to all crops in 2015 (the latest year for which data are available), of which 18% was applied to sugarcane (DAFF, 2016). Depending on the soil test result, most cane-growing countries recommend a maximum of between 100 and 250 kg K ha⁻¹ (Kingston, 2000; Miles, 2014). At a cost of about R12 kg⁻¹ and an assumed average of 150 kg K ha⁻¹ applied to the 270 000 ha harvested annually, South African sugarcane growers apply 40.5 million kg K to their fields every year at a cost of over R485 million.

2.2.3.1. Consequences of excess potassium in sugarcane

The luxury uptake of K by sugarcane (Kingston, 2000) generally has little negative effect on the crop or production system and there have been no reports of harmful effects of K on the environment, or to humans (Anonymous, 1984, 1998, cited in Askegaard *et al.*, 2004). Economically, however, this has a direct adverse effect on profits (Ritchey, 1979) if K fertiliser is applied in excess of that required for optimum yield. However, the major impact of excess K is in the milling operation, where increased ash in cane juice and raw sugar results in lower recovery of sugar crystal from cane (Kingston, 1982; Munsamy, 2013).
2.2.3.2. Potassium and moisture stress

Moisture stress may depress the amount of K taken up by sugarcane (Schroeder *et al.*, 1993). Sugarcane leaves sampled after an extended dry period contained considerably less K than the same field sampled two months later, after ample rain (Schroeder *et al.*, 1993). Possible reasons for this included better root growth and functioning after rainfall, increased mass flow of soil K and release of non-exchangeable soil K (Wood and Schroeder, 2004). Kee Kwong *et al.* (1990), however, reported that leaf K concentration could not be predicted by the moisture regime at or near the time of sampling. Where K was deficient, moisture stress tended to decrease sugarcane yields, while K-rich fields subjected to the same stress did not show this yield penalty (Wood and Schroeder, 2004).

Potassium supply plays an important role in drought tolerance of plants, including sugarcane. Potassium sufficiency affects cell osmosis, water potential and stomatal function (Huang *et al.*, 2009a), assisting K-replete plants to withstand drought better than their K-deficient counterparts. Under conditions of moisture deficit, K increases the efficiency of water use and leaf biomass (Valeri *et al.*, 2016). Foliar application of K has also been reported to improve growth, dry matter accumulation, relative water content, gas exchange capacity, stomatal conductance and net photosynthesis rate of drought stressed plants (Ihsan *et al.*, 2013).

2.2.4. Nutrient interactions

Potassium is known to interact in various ways with other nutrients. Some interactions are fairly general. For instance, if another element such as P or N limits plant growth, K uptake will be low, even in the presence of large amounts in the soil (Ritchey, 1979). Other interactions, however, are more specific, as discussed below.

2.2.4.1. Potassium interactions with calcium and magnesium

Researchers have reported conflicting results with regard to interactions involving K, calcium (Ca) and magnesium (Mg), and their subsequent uptake by plants. Some have recorded a decrease in K absorption by plants as a consequence of Ca and/or Mg fertilisation (Stevens, 1970; Ritchey, 1979; Wood and Meyer, 1986). High levels

of Ca and Mg favour uptake of these elements via mass flow, interfering with K uptake (Kingston, 2000). Other researchers have shown a depressive effect of K on Ca and Mg content of plants (Hossner and Doll, 1970; Ritchey, 1979; Jakobsen, 1993; Daliparthy *et al.*, 1994; Jaskulska *et al.*, 2015; Bauw *et al.*, 2016), while little or no interactive effects have also been reported (Rodriguez, 1975; Gonzalez, 1976). Liming further complicates the interpretation of K trial results, mainly through yield effects. Where the lime rate is varied, under conditions of high soil acidity, insufficient lime may lead to decreased yields, reducing K uptake and leading to more unused K remaining in the system, thus confounding the interpretation of K distribution patterns (Ritchey, 1979). Conversely, increased crop yields following liming may result in decreased K throughout the profile due to greater crop uptake of K (Ritchey, 1979). Results are less confounded when lime rates are within the range unlikely to affect yield or K uptake (Ritchey, 1979).

2.2.4.2. Potassium interactions with nitrogen

Although an increase in the amount of N supplied to the crop generally encourages growth and the uptake of K from the soil (Johnston, 1986), the opposite has also been reported. High NH_4^+ in the soil can interfere with K uptake (Spalding *et al.*, 1999). Nitrogen deficiency can suppress K uptake by the plant (Miles, 2010). Similarly, inadequate K supply can lead to reduced crop yields, causing underutilisation of N (and P) fertiliser, thereby rendering N vulnerable to loss by leaching, erosion or volatilisation (Krauss, 2001). In this regard, K has an important environmental function. An adequately nourished crop provides better soil anchorage, so reducing nutrient losses via erosion, and higher yields with sufficient nutrition promote greater N use efficiency, reducing the potentially harmful effects of N losses in runoff (Krauss, 2001).

2.2.4.3. Potassium interactions with aluminium and manganese

Although K and aluminium (AI) or manganese (Mn) do not compete for uptake by plants, high exchangeable AI and Mn, often present in acid soils, can seriously affect the healthy development of crop roots. A poorly functioning root system is therefore less efficient at K uptake (Ritchey, 1979). In such cases, liming would decrease AI

and Mn availability, improving root growth and therefore K uptake. Liming also renders Al insoluble, thereby increasing the portion of the cation exchange capacity (CEC) available for K adsorption, along with the general increase in effective CEC due to the deprotonation of pH-dependent sites effected by liming (Ritchey, 1979).

2.2.5. Soil analysis and plant uptake

Unfortunately, the relationship between soil K and crop response to K fertiliser is often poorly defined. While glasshouse trials may show strong correlations, under field conditions relationships can be confounded by many factors (Askegaard *et al.*, 2004). Whereas soil samples are typically taken from the plough layer, dried, ground, homogenised and extracted with chemicals targeting specific pools, plant roots *in situ* experience a very different environment. A wide range of factors affect the amount of K available to, and taken up by, plant roots in the field. These include (after Askegaard *et al.*, 2004):

- Soil texture and mineralogy: the type of clay mineral, rather than the amount of clay present, tends to control the rate of available K replenishment (Rao and Khera, 1994).
- Temperature and soil moisture content: Low temperatures limit root extension and K uptake by plant roots (Schimansky, 1981), and affect the availability of K in the soil (Sparks and Liebhardt, 1982). Low soil moisture limits K diffusion (Kuchenbuch *et al.*, 1986).
- Soil compaction and waterlogging: Soil compaction may impede root elongation and diffusion of K (Marschner, 1995), potentially causing K deficiency in plants. Waterlogging also inhibits K uptake, which is highly dependent on oxygen-controlled metabolic processes.
- Potassium status: Potassium taken up from the soil solution by plant roots can be replaced from the other pools (Sparks, 2001) with the result that, over a few years, slight positive or negative K balances in terms of crop uptake may have little effect on exchangeable K values.
- Status of other nutrients in the plant: In order for K to be efficiently taken up by the plant, other nutrients (N, in particular) need to be present in the plant at

adequate levels. Interactions are particularly important when both nutrients are near the deficiency range (Marschner, 1995).

- Crop species and variety: Monocotyledons tend to exploit soil K reserves more efficiently than dicotyledons (Johnston *et al.*, 1998; Krauss, 2001), with the result that cereals and grasses frequently do not respond to K fertilisation (Mengel, 1982).
- Crop uptake from the subsoil: Plants can take up considerable amounts of K from the subsoil. Cereals can derive as much as 70% of their K from the subsoil.

2.3. Nitrogen

Nitrogen is a major constituent of nucleic acids, proteins, enzymes and chlorophyll (Kingston, 2000). It promotes vegetative growth, and the leaves of crops replete with N are a deep green colour. A shortage of N can have dramatic effects, such as yellowing of the leaves, stunted growth and reduced yield (Kingston, 2000).

Nitrogen may be lost from cropping systems by runoff, leaching, volatilisation and denitrification (Schumann, 2000), all of which may render N unavailable for crop use and represent wasteful expenditure. Excessive N may result in eutrophication of surface waters and, in drinking water, causes blue baby syndrome (Knobeloch *et al.*, 2000), while nitrogen oxides are potential atmospheric pollutants (Paton-Walsh *et al.*, 2011; Davis *et al.*, 2016). Nitrous oxide, a potent greenhouse gas, is emitted in exponentially greater amounts when N fertiliser is applied in excess of the crop's needs as opposed to applications which satisfy (but do not exceed) the plants' requirements (Shcherbak *et al.*, 2014).

2.3.1. Nitrogen in sugarcane nutrition

Nitrogen deficiency in sugarcane results in thin, stunted stalks and reduced tiller numbers (Wood, 1968; Kingston, 2000). Yellowing of the leaves usually occurs, affecting the older leaves first, although yield reduction is usually experienced before chronic visual deficiency symptoms become evident (Kingston, 2000).

Excessive rates of applied N result in reduced sucrose concentration in the fresh mass of harvested sugarcane (Stevenson *et al.*, 1992; Chapman, 1994; Muchow and Robertson, 1994; Muchow *et al.*, 1996). According to Muchow *et al.* (1996), this reduction is related to an increase in moisture percent with increasing N application rates. An excess of available N too close to harvest can thus lead to lower sugarcane quality and affect profitability (Nassar *et al.*, 2005). Excessive applied N may also be associated with lodging (Kingston, 2000). Stalk yield is typically increased (up to a point) with N application (Muchow *et al.*, 1996). To enable optimal sugarcane growth, appropriate N recommendations must, therefore, achieve a balance between supplying enough N to maximise stalk yield, while avoiding luxury uptake of N.

The plant crop (the first crop grown after planting sugarcane stalks or setts) often shows little response to N application, while ratoon crops usually respond better and more frequently than their plant crop counterparts (Du Toit, 1957; Wood, 1964; Chapman, 1994). This phenomenon is usually attributed to the amount of inorganic N released by increased soil micro-organism activity when soil organic matter is exposed to the atmosphere during land preparation (ploughing and drawing of furrows) (Wood 1964, 1965). Ratoon fields are not ploughed following the previous harvest, and so less N is released by the soil, justifying greater N application rates than for plant crops (Chapman, 1994). Du Toit (1959) also reported an increased response to applied N from the fourth ratoon crop, and greater responses to N, P and K with each successive crop following the plant crop. Meyer and Wood (1994) recommended a 20 kg ha⁻¹ increase in N application from the fourth ratoon onwards.

The season during which a sugarcane field is harvested/ratooned affects the pattern in which N is taken up by the crop. In work by Thompson (1988), 82% of the final amount of N accumulated was taken up within the first four months of growth of a summer-cut crop. In contrast, only 12% of the final N tally was taken up in the first four months of an April (autumn)-harvested crop. This difference in N uptake has implications for the timing of N fertiliser applications. Nitrogen fertiliser may be applied shortly after the harvest of a summer-harvested crop, as the following crop will be growing actively and will be able to take up and utilise the N relatively rapidly. For an autumn/winter-harvested crop, however, where the next crop will only grow very slowly until spring, applying N too soon after harvest will leave the N unutilised and vulnerable to loss (Meyer *et al.*, 2007).

Nitrogen is well documented to affect plant disease development, usually increasing the plants' susceptibility to disease (in the case of obligative parasites) with increasing N application rates (Dordas, 2008). However, less evidence exists to link N application rates with diseases of sugarcane. Pannuti *et al.* (2015) recorded increased infestation of the borer *Diatraea saccharalis* - red rot complex - with increasing N application in sugarcane. In South Africa, high plant N has been reported to favour increased damage by the stalk borer *Eldana saccharina* (Goebel *et al.*, 2005; Meyer and Keeping, 2005). Lowering N application rates in an effort to reduce eldana damage, however, inevitably leads to yield reductions, and may be unwarranted on fine-textured soils where moisture stress, which is strongly correlated with eldana damage, is generally less severe (Rhodes *et al.*, 2013).

2.3.2. Nutrient interactions

2.3.2.1. Nitrogen interactions with phosphorus and potassium

Nitrogen promotes the uptake of a number of nutrients, including P and K (Du Toit, 1957; Miles, 2010). Without adequate N nutrition, a crop is often unable to function effectively, reducing the uptake of other nutrients. However, maximum crop response to N is dependent on the availability of sufficient quantities of other nutrients, such as P and K (Du Toit, 1957).

2.3.2.2. Nitrogen interactions with silicon/ lime

Increases in plant N may decrease plant Si concentration, due to dilution by growth (Hsieh *et al.*, 1983; Marschner, 1995). Silicon is required to minimise pest and disease levels which may increase with the amount of N required for maximum growth (Hsieh *et al.*, 1983; Meyer and Keeping, 2005; Kumara *et al.*, 2016). Plant-available N is affected in another way by Si sources. Calmasil®, for example, a calcium-magnesium silicate slag widely used in the South African sugar industry, acts similarly to dolomitic lime, and when moderately to highly acidic soils are limed, N mineralisation is greatly increased due to stimulation of nitrifying bacteria (Wood, 1979). Plant uptake of N is increased, in some cases to such an extent that sucrose

percent is reduced, necessitating a lowering of N fertiliser recommendations following liming (Wood, 1979). This effect is not directly due to Si itself, but rather due to the changes in soil chemistry associated with application of the Si source. The reported responses may thus have been due to Ca, Mg or Si or any combination of these nutrients.

2.4. Silicon

Silicon and its functionality in plants have been studied for many years (Whittenberger, 1945), but it is only recently that the extent of its importance has been recognised (Epstein, 1999; Guntzer *et al.*, 2012). It is not considered an 'essential' plant element, because most plants can grow in nutrient solutions to which no Si has been added. Due to its numerous roles in the plant, however, for example increasing mechanical strength and resistance to pests and diseases, Si is considered a 'functional' (Meyer and Keeping, 2000) or 'quasi-essential' (Epstein, 1999) plant nutrient.

2.4.1. Silicon in soils

Silica (SiO₂) is slowly dissolved to form neutral orthosilicic acid [Si(OH)₄] or monosilicic acid (H₄SiO₄) in soil solution at a pH below 9.0 (Ma *et al.*, 2001). Monosilicic acid is the only form of soluble Si which plants can use (Epstein, 1999). Most gramineae, including sugarcane, actively take up Si and once absorbed by plants, the silicic acid is deposited in the stems and leaves as hydrated silica (SiO₂·*n*H₂O) to form deposits known as phytogenic Si (Smithson, 1956; Haynes, 2014). As plant matter decomposes, this silica is released and becomes incorporated into the soil. When sugarcane is burned pre-harvest, most of the plant Si is retained in the ash in an amorphous form, with a small proportion as quartz (Le Blond *et al.*, 2010). Soil analysis may reveal the presence of silt-sized particles of phytogenic Si (termed phytolith Si) (Haynes, 2014). Their dissolution transforms the silica back into monomeric silicic acid, thus completing the biogeochemical cycle of Si (Sangster *et al.*, 2001). In many forest and grassland systems, a large percentage of the Si taken up by plants is from phytogenic silica. This recycling of Si may be particularly important in highly weathered tropical and subtropical soils that have low available Si and where loss of mineral Si results in most of the Si in soil solution being supplied phytogenically (Cornelis *et al.*, 2011; Haynes, 2014).

2.4.2. Plant availability and uptake of silicon

Under conditions of high rainfall and temperature, basic cations are stripped from the soil exchange complex leading to acidification of the soil, and dissolution of aluminosilicate clay minerals. The subsequent leaching of Si (desilication) is a natural process and results in relatively little plant-available Si. An Australian study of sugarcane soils on the wet, tropical coast of north Queensland showed that more than 85% of the soils tested had sub-optimal or marginal Si levels (<20 mg Si kg⁻¹) (Berthelsen *et al.*, 2001a). Plant Si generally corresponded to the extractable soil Si, being low where soil Si was sub-optimal. Sumner *et al.* (1991) commented that natural leaching (and acidification) associated with highly weathered soils may result in these soils being an order of magnitude lower in soluble Si than less weathered (and naturally more fertile) soils. These findings concur with the general pattern found in the South African sugar industry, where higher rainfall areas have reduced soil and leaf Si, while the drier, irrigated regions with more base-rich soils have higher Si (Van der Laan and Miles, 2010).

Silicon availability (and hence plant uptake) may be reduced by factors other than desilication. Monosilicic acid may be adsorbed by iron (Fe) and Al oxides, so decreasing its concentration in soil solution (Jones and Handreck, 1963; Haynes, 2014). Within a particular soil, soluble Si tends to be higher at lower pH values, while increasing pH (up to ~pH 9) (McKeague and Cline, 1963; Haynes *et al.*, 2013) leads to lower Si in solution (Haynes, 2014). Curtin and Smillie (1983) recorded a dramatic reduction in soil solution Si after liming. Ayres (1966) and Savant *et al.* (1999) also reported that addition of lime to a soil can reduce Si solubility, due to an increase in the soil pH. Jones and Handreck (1967) similarly reported a number of studies indicating that an increase in pH caused a reduction in plant uptake of Si. This was confirmed in sugarcane by Du Preez (1970), where leaf Si decreased following liming. De Camargo *et al.* (2007), however, reported that an increase in pH(caCl₂) from ~pH 4 to pH 6.3, without an added Si source, did not affect Si uptake by rice plants. If Si slag materials are applied, however, soil pH may increase (due to the

liming effect of the product), along with soluble Si (due to the Si contained in the product) (Haynes, 2014). While increasing soil pH may render mineral Si less available (due to adsorption by Fe and Al oxides), the dissolution (and hence plant-availability) of phytolith Si is increased significantly (Haynes, 2014). The overall effect of pH on Si availability is, at present, unclear, and differs according to the Si source applied (Haynes, 2014). Keeping *et al.* (2017) applied lime to an acid soil simultaneously with, or 1 or 3 months prior to, the application of Calmasil®. They found that Si uptake by sugarcane plants and, in some cases, available soil Si itself, was not improved by liming either with or before Calmasil® application, in comparison with Calmasil® application alone.

Restricted plant uptake of Si may also be due, in part, to the low solubility of orthosilicic acid, which renders a practical limit to increasing the availability of Si to plants in the field (Côté-Beaulieu *et al.*, 2009). Although applications of siliceous material may increase soil and leaf Si, these increases are often not enough to bring the plant above the critical Si threshold (Berthelsen *et al.*, 2001b). Keeping and Meyer (2002) applied 5 and 10 t ha⁻¹ of calcium silicate to sugarcane plants, and found that leaf and stalk Si content was, in most cases, increased by 5 t ha⁻¹, but decreased at the higher rate. In Mauritius, Ross *et al.* (1974) reported low uptake of applied Si, such that, after six sugarcane crops, the amount of Si recovered from the original application at planting was only 11-14% of the total.

A further factor contributing to the depletion of plant-available Si in sugarcane systems may be continuous monocropping, with large annual removals of Si to the mill at each harvest (Meyer and Keeping, 2000). Kraska and Breitenbeck (2010) found similar patterns in rice where Si deficiency was most pronounced in strongly acidic soils, with a long history of rice production. In this case, however, the correlation between soil and plant Si was weaker than the (inverse) correlation between soil pH and plant Si, indicating that soil pH was a more important determinant of plant Si uptake than the duration of cropping (represented by available soil Si).

Fertilisation with Si has become a widespread practice for various crops. A number of Si sources exist, including potassium silicate, sodium silicate and meta-silicate, calcium silicate, cement, rock dust, silicic acid, magnesium silicate, silicon gel and industrial by-products such as fly-ash and bagasse furnace ash. The type of Si fertiliser applied affects plant uptake (Ranganathan *et al.*, 2006), possibly due to the ease of formation of orthosilicic acid from the fertiliser products (Mecfel *et al.*, 2007). Uptake of Si by plants is also affected by the water regime or rainfall pattern and is greater under adequate rainfall or irrigation than under conditions of water deficit (Eneji *et al.*, 2008). Improved uptake of Si by sugarcane plants was recorded by Bell *et al.* (2002) after bare or crop fallows, or soil fumigation. Similar results were found for K. The authors did not record increased levels of these nutrients in the soil; rather, they attributed the increased plant uptake to a reduction in pathogen pressure, leading to an improved root system.

2.4.3. The effects of silicon on sugarcane production

Sugarcane is a Si accumulator plant, and takes up more Si than any other mineral nutrient, accumulating up to 380 kg Si ha⁻¹ in a 100 t ha⁻¹ crop (Savant *et al.*, 1999). The effects of applying Si to sugarcane were described by Mauritian researchers in the 1940s (De Villiers, 1947, cited in Meyer and Keeping, 2000). Since then numerous researchers have recorded sugarcane yield increases (Clements, 1965; Ayres, 1966; Fox et al., 1967; Ross et al., 1974; Korndörfer and Lepsch, 2001; Huang et al., 2009b) and a reduction in the leaf freckling that is associated with Si deficiency (Clements, 1965; Fox et al., 1967; Wong You Cheong et al., 1973) following application of silicate sources. The yield increases are generally associated with longer stalks, larger stalk diameters and an increased number of millable stalks, rather than an increase in the sucrose percent (Korndörfer and Lepsch, 2001). Such yield responses may last for six crops or more after calcium silicate (Ca_2SiO_4) application at planting (Ross et al., 1974). Silicon deficiency has been shown to decrease the rate of photosynthesis and sugar production when compared to Si-rich cane, even when deficiency symptoms are not evident (Wong You Cheong et al., 1971).

When calcium silicate slags are applied, there is difficulty in differentiating between the sugarcane response to the applied Ca (and other nutrients), the increased pH, and that obtained as a result of the Si (Sumner *et al.*, 1991). Du Preez (1970) grew sugarcane on various KwaZulu-Natal Midlands soils in pots, and added a number of Si sources, including slags and cement, as well as calcium carbonate (CaCO₃). In general, yields were increased by the addition of both siliceous materials and calcium carbonate and were attributed to both decreased AI and Mn, and increased Si in the soils. Silicate slag appeared to be superior to calcium carbonate because it caused a similar increase in pH, gave higher yields and the consequences of over-application were considered less harmful. Gascho and Andreis (1974) reported increased sugarcane and sugar yields following application of calcium silicate slags in Florida, USA, and concluded that increased Si, rather than any alteration in uptake of other nutrients (such as P, Mn or Fe), or a liming effect, was the reason for the yield increases.

Berthelson *et al.* (2001b) applied various siliceous products (calcium silicate slags, mill ash, mud ash, cement, cement board by-product and rock dust) to sugarcane soils in 'product trials' in the Mossman, Innisfail and Bundaberg districts of Australia. Several of these products, especially the calcium silicate slags, cement and mill ash, resulted in significantly higher sugarcane yields than the control on low-Si soils. In a report on the same study, Berthelsen *et al.* (2003) stated that a strong relationship existed between relative cane yield and the Si status of the top visible dewlap leaf (the plant part usually sampled for nutrient studies in sugarcane).

Some of the most widely reported benefits of Si include its protective role against pests and diseases. The physical effects of Si on insect pests can include both reduced growth and mandibular wear (Goussain *et al.*, 2002; Massey and Hartley, 2009). In sugarcane, Si application has consistently been shown by South African researchers to reduce incidence, growth and damage of the eldana stalk borer (Keeping and Meyer, 2002; Kvedaras and Keeping, 2007; Kvedaras *et al.*, 2007) through increased Si deposition in the epidermal tissue (Keeping *et al.*, 2009). The application of Si has also resulted in a degree of control over other sugarcane pests, including Spittlebug (*Mahanarva fimbriolata*) in Brazil (Korndörfer *et al.*, 2011), sugarcane top borer (*Scirpophaga nivella intacta*) in Indonesia (Saeroji *et. al.*, 2010), and nematodes in South Africa (Berry *et al.*, 2011) and South America (Silva *et al.*, 2010). Silicon may also reduce the incidence and severity of fungal plant diseases as in the protection of sugarcane against common rust (*Puccinia melanocephala*) (Cadet *et al.*, 2003).

Meyer and Keeping (2000) list a number of other benefits of Si, including enzyme regulation in sugar synthesis, storage and retention in the cane plant, and increasing the resistance of cane to "freeze damage". Savant *et al.* (1999) also mention the latter, as well as other benefits such as improved water economy and reduced lodging. As a result of the numerous recorded benefits, Si is widely added as a sugarcane fertiliser in Brazil, Australia, South Africa, India and the USA (Epstein, 1999; Savant *et al.*, 1999).

2.4.4. Nutrient interactions

2.4.4.1. Silicon interactions with potassium

After application of calcium silicate to sugarcane, Huang *et al.* (2009b) found that increased rates of Si were associated with a decrease in plant K at 12 months. Miao *et al.* (2010) applied Si to K-deficient soybean seedlings, increasing root and shoot mass and increasing the K concentration in the leaves, stems and roots by between 59 and 105%. The authors concluded that Si not only increases tolerance to nutrient toxicity, but also that deficiency symptoms associated with essential nutrients can be ameliorated by Si (Miao *et al.*, 2010). Silicon application to wheat seedlings also increased K concentration in the shoots, enhancing salinity tolerance by increasing K uptake and hence increasing the K:Na ratio, thereby lowering Na translocation to the shoot (Tahir *et al.*, 2010). Potassium fertilisation has been reported to have an effect on Si uptake, though this effect is variable and often small. Hartt (1934) found that increased K caused a slight decrease in the Si content of sugarcane.

2.4.4.2. Silicon interactions with nitrogen

When plants are N-deficient, application of N generally leads to decreased Si uptake because N fertilisation leads to more efficient water use by plants. The effect on Si is therefore an indirect one, because the more efficient plant produces more dry matter for each unit of water and Si absorbed (Jones and Handreck, 1967). This interaction would be particularly notable in plants in which Si uptake is passive (and hence driven by water uptake). Once plants are receiving adequate N, dry matter production, and hence Si content, level off. Dilution by growth effects (Marschner, 1995) were also reported by Hsieh *et al.* (1983), who studied the effects of dolomitic lime, N rates and rice hull (a Si source) applications on the Si content and growth of rice in Taiwan. As N rates increased, growth generally improved, but the Si content of the rice tended to decrease. Dolomitic lime and rice hull applications at the highest N rates tended to increase rice growth and yield, and improved resistance to lodging and blast. Lower N rates did not facilitate maximum growth. Thus the highest rate of N was required to improve growth and yield, while lime and Si were needed for resistance to lodging and blast (Hsieh *et al.*, 1983). Huang *et al.* (2009b) found that increasing application rates of calcium silicate to sugarcane were associated with a decrease in plant N at 12 months. Eneji *et al.* (2008), however, found that Si uptake by four grass species was positively correlated with N uptake under deficit irrigation.

2.4.4.3. Silicon interactions with phosphorus

Many researchers have reported beneficial effects of Si on plant growth when available P is low, but the reasons for this are poorly understood (Tavakkoli *et al.*, 2011a). It is suggested that increases in extractable P (and plant uptake) following Si application could be due to Si displacing P from the soil adsorption sites, or to an increase in P solubility resulting from an increase in soil pH (Berthelsen *et al.*, 2003; Eneji *et al.*, 2008). Adding P to P-deficient plants tends to decrease plant Si, via dilution by growth, similar to N (Jones and Handreck, 1967).

2.4.4.4. Silicon interactions with calcium and magnesium

In rice, Ma and Takahashi (1993) found that Si addition resulted in a decrease in Ca content of the shoots, as well as Ca uptake, at each Ca level tested. The authors believed that this mainly resulted from a decreased transpiration rate caused by Si. Silicon uptake was not affected when Ca levels were increased.

Anderson (1991) reported nutrient antagonism between Si and Mg in sugarcane. The author suggested that application of Mg fertiliser may be necessary to maintain high Mg availability when Si is applied. Kidder and Gascho (1977) also recommended concurrent Mg fertilisation (along with Si) as a precaution if the soil Mg test value was below a certain threshold.

2.5. Nutrient recommendations

2.5.1. Potassium

In most countries, fertiliser K recommendations for sugarcane are based on the topsoil test K value, and are modified according to various factors, as determined by field trials. Most of the countries listed by Kingston (2000) and Meyer (2013) base their K recommendations on concentrations of exchangeable soil K. In Australia and Mauritius, K recommendations are also modified according to non-exchangeable soil K reserves (Schroeder and Wood, 2002; Meyer, 2013). Potassium recommendations may be further adjusted by taking one or more of the following factors into account:

- Texture and base status of the soil, attainable (potential) yield, harvesting method (green or burnt cane harvesting) (South Africa) (Miles, 2014).
- Crop age (plant versus ratoon) (Brazil, Cuba, USA and South Africa) (Meyer, 2013).
- Leaf K concentration (Cuba) (Meyer, 2013).
- Potassium concentration in irrigation water (Hawaii, USA) (Meyer, 2013).
- Percentage of total CEC occupied by K (Dominican Republic) (Redman, 1991).
- Soil acidity (Indonesia) (Soepardi, 1991).
- Soil parent material (Indonesia) (Meyer, 2013).

Meyer (2013) presents a summary of selected soil K threshold values and K fertiliser recommendations for a number of sugarcane-producing areas.

2.5.2. Nitrogen

Soil N tends to be ephemeral, with N liable to change form due to biological and chemical reactions and to be lost by various mechanisms. For these reasons, N fertiliser recommendations for sugarcane are not typically based on N soil tests. Rather, N recommendations tend to be based on N response trials, and are modified according to a number of factors. Nitrogen mineralisation potential of the soil (Schroeder *et al.*, 2005, 2007; Meyer, 2013; Miles, 2014), district yield potentials (Schroeder *et al.*, 2006) and attainable yield (yield target per field) (Miles, 2014) are

some of the main drivers of N fertiliser recommendations. Other factors used to modify N recommendations include:

- Crop age (plant versus ratoon) (Australia, USA, Brazil, India and South Africa) (Meyer, 2013).
- Residual N from leguminous cover crops (South Africa (Miles, 2014); Australia (Schroeder *et al.*, 2005); India (Meyer, 2013)).

Meyer (2013) presents a summary of selected N fertiliser recommendations for a number of sugarcane-producing areas.

2.5.3. Silicon

Plant-available Si in soils may be estimated using both acid and neutral extractants (Sauer *et al.*, 2006). Threshold values for the various extractants have been established through field trials and soils with Si concentrations below these thresholds should receive Si applications in order to attain optimal sugarcane yields.

In South Africa, recommendations for Si application are not routinely made by the FAS. Rather, when soil Si falls below the threshold of 15 mg Si L⁻¹ in 0.01 M CaCl₂ (Miles, 2014), it is suggested that growers consider some form of Si application. At present, Calmasil® is commonly used to replace some, or all, of the dolomitic lime recommendation.

2.6. The thesis in context: Gap analysis and motivation for the study

As shown by this review, K, N and Si have been the subject of extensive research on a number of different crops, and much is known about their functions and requirements in sugarcane. Questions remain, however, which require addressing in order to refine the recommendations currently made by soil testing laboratories, and the FAS in particular.

2.6.1. Potassium

Dynamics between the different pools of K (Section 2.2.1) result in a range of K availability to plants. Potassium release from less readily-available pools can buffer K availability, satisfying a crop's needs even when exchangeable (readily-available) K is low. The extent of this availability has not, until recently, been explored in the South African sugar industry, and the K supplying power of local soils is not well established. The extent to which even low reserve K can contribute to crop K supply has not been investigated in South African sugarcane soils and, because excessive K may reduce mill sucrose recovery, appropriate K recommendations are essential.

Fertiliser recommendations are typically based on measurements of readily-available K in the top 20 cm of soil. However, international research suggests that plant roots can source K from greater depths. This has not been studied in the South African sugar industry and it is not known if subsoil K should be taken into account when making K fertiliser recommendations. Stratification of K, especially in higher clay soils, may also affect soil sampling and general management, and requires investigation. The effect of K application on sugarcane root growth (depth and general root proliferation) is also largely unknown.

Potassium's interactions with other nutrients, particularly Ca and Mg, have been well documented in a number of crops. In sugarcane, high Ca and Mg are known to inhibit K uptake, necessitating higher K recommendations under certain conditions. The reverse, however, has not been widely investigated internationally, and not at all in the South African sugarcane industry. Understanding the interactive effects of high soil K on Ca and Mg uptake may lead to improved nutrient recommendations within the industry. In addition, interactions between K and other nutrients, including N and Si, require further elucidation. Potassium fertilisation of sugarcane may affect Si uptake, though this effect is variable and often small. The extent and regularity with which K application affects plant Si uptake needs further study.

2.6.2. Nitrogen

Although N nutrition in sugarcane has been the subject of substantial research, some knowledge gaps remain. One relates to N fertiliser recommendations in plant crops in comparison with the ratoon crops. In the South African set of

recommendations, plant crops routinely receive less N than ration crops, assuming N release following land preparation. In humic soils, it may be possible to reduce the N application further, but no substantial trial work to investigate this suggestion has been performed.

Other gaps involve the interactions between N and other nutrients. For example, increased plant N may lead to a decrease in plant Si concentration, following dilution by growth. This phenomenon, little studied in sugarcane, may have relevance in the South African context where adequate plant Si concentrations are required to offset increased N in restricting eldana damage. A further consideration is that the application of Si products (which usually have a liming effect) may lead to the proliferation of soil bacteria, stimulating N release and subsequent plant uptake. Further study is required to investigate the possible effects of Si application on N uptake and, consequently, sucrose concentration.

Nitrogen is a driver of plant growth and may, therefore, alter the uptake and concentration of other applied nutrients. A study of nutrient application rates, and K in particular, is more complete when the effects of different N levels are taken into consideration.

2.6.3. Silicon

While the benefits to sugarcane of added Si are well documented, the appropriate soil threshold under South African conditions is still a matter of debate. The current industry soil Si threshold of 15 mg L⁻¹ has not been researched sufficiently to be confident that this level provides the known benefits.

Whatever the threshold used, it has proven challenging to increase plant uptake of Si from products such as Calmasil®. Further work is required to establish the extent to which plant Si can be increased by the application of Calmasil® or other products, and whether this is sufficient to reach (or exceed) threshold values. It is necessary to know, in cases where Si application increases plant Si without actually reaching the threshold, whether sugarcane yield is nonetheless increased. The general effects of Si application on cane yield also require confirmation on South African soils. Understanding will be improved if Si-related sucrose yield increases (if any) can be apportioned to either increased stalk yield or sucrose concentration, or both.

The behaviour of applied Si in soil also requires elucidation. The extent to which Si moves in the soil, or results in stratification, has not been investigated locally. In addition, little is known of the effects of Si on sugarcane rooting and whether such effects can explain why drought tolerance is improved with Si application.

These aspects are reflected in the key questions given in Chapter 1 (Section 1.3.2). They centre around the effects of K, N and Si on sugarcane crop yield and quality, which, in turn, will inform the nutrient recommendations which are made by the FAS in order to achieve optimal yield.

CHAPTER 3

The effect of application rates of potassium, nitrogen and silicon on sugarcane harvest characteristics

3.1. Introduction

Potassium (K) fertiliser recommendations for sugarcane have not kept pace with K removals and unsustainable K 'mining' has occurred, both in South Africa and overseas (Kingston, 2000). In South Africa, this has resulted in low K reserves in many topsoils of the rainfed sugarcane areas (Van der Laan and Miles, 2010). The importance of K can be underestimated for a number of reasons (Section 2.2). The role of K in soils [and plants] is, however, "prodigious" (Sparks and Huang, 1985), and accurate estimates of the K requirements of sugarcane are needed.

Nitrogen (N) has been extensively studied, and continues to be investigated due to its pivotal role in sugarcane growth and yield (Stranack and Miles, 2011), its characteristic as a 'driver' of the uptake of other nutrients (Miles, 2010) and the widely differing extent to which N is supplied by different soils (Meyer *et al.*, 1986). Too much N adversely affects crop quality (Singh, 1973), and the effect of excessive N on crop susceptibility to pests (Carnegie, 1981) and diseases (Mengel and Kirkby, 2001) is well known.

Silicon (Si), although not considered essential for sugarcane, is a 'beneficial' nutrient (Meyer, 2013) due to its importance in protecting against biotic and abiotic stress. Research has highlighted the extensive Si deficiency of sugarbelt soils in the rainfed parts of the country (Van der Laan and Miles, 2010) and further research is required to understand its effects on sugarcane growth and to establish sound approaches for correcting Si deficiencies.

While the individual effects of nutrients on sugarcane are important and, in some cases, well documented, the interactions between nutrients are less well understood. Interactions between N and Si, for example, have been recorded by Meyer and Keeping (2005). Other interactions between K, N and Si, on different soils, require elucidation. This study was therefore designed to determine the effect of various application rates of K, N and Si, along with their interactions, on sugarcane yield and

other harvest characteristics, to assist in refinement of the industry nutrient recommendations.

3.2. Materials and methods 3.2.1. Trial sites

The contrasting soil types at the two trial sites on the KwaZulu-Natal north coast were selected as they represent approximately 70% of the rainfed area of the South African sugar industry and 50-60% of the total area cropped to sugarcane. One trial was situated in the Inanda area, on the farm 'Inanda' (29.629°S, 30.928°E; 560 m a.s.l.), inland of Verulam (Figure 3.1). Mean annual rainfall is approximately 1 090 mm, with most (74%) occurring during the summer months (October to March). The area has mean daily maximum and minimum temperatures of 25.7°C and 16.2°C, respectively. The soil is an Oxisol (Humic Eutrustox; Soil Survey Staff, 2014); Humic Ferralsol (IUSS Working Group WRB, 2014); Inanda form, Glenariff family (Soil Classification Working Group, 1991), with an effective rooting depth of over 100 cm.

The second trial was located on the farm 'Thornwood' at Doringkop (29.220°S, 31.240°E; 427 m a.s.l.), inland of KwaDukuza (Stanger) (Figure 3.1). Mean annual rainfall is 1 130 mm, with most (77%) occurring between September and March. The mean daily maximum temperature is 26.7°C and the mean daily minimum 16.5°C. The soil is an Inceptisol (Typic Haplustept; Soil Survey Staff, 2014); Leptic Cambisol (IUSS Working Group WRB, 2014); Glenrosa form, Dumisa family (Soil Classification Working Group, 1991), with an effective rooting depth of approximately 80 cm. The parent material at both trial sites is Natal Group Sandstone (Mariannhill Formation, Tulini Member) (Marshall and Von Brunn, 1999). The trial sites are henceforth referred to by their soil type, i.e. Oxisol and Inceptisol.



Figure 3.1: Location of trial sites (Oxisol and Inceptisol) within the Province of KwaZulu-Natal, South Africa. (Image: modified from http://d-maps.com/m/africa/southafrica/kwazulunatal/kwazulu-natal22.pdf accessed 19 Sep 2017).

3.2.1.1. Rainfall

The Oxisol (Figure 3.2A) received 101%, 103% and 134% and the Inceptisol (Figure 3.2B) 98%, 115% and 104% of the long-term mean (LTM) for each of the three growing seasons, respectively.



Figure 3.2: Monthly long-term mean (LTM) and actual rainfall for the duration of the trial on the (A) Oxisol and (B) Inceptisol. Arrows indicate the months in which leaf sampling (Section 4.2.2) was done. Black, hatched and white arrows indicate possible limiting, marginal and acceptable moisture conditions, respectively, at the time of leaf sampling.

There was, however, a severe drought during the winter of 2010. From February to September 2010, the Oxisol received only 47% of its LTM; the Inceptisol, 46%. Both sites received some rainfall into April 2010 (Figure 3.2), lessening the effects of the drought.

Following the 2010 drought, each site experienced intermittent periods of above- or below-average rainfall, but the higher than average rainfall of spring 2012 was the only other period where rainfall was consistently different to the LTM over a period of three to four months.

3.2.2. Soil sampling

Soil samples were collected with an auger, from 0-20, 20-40, 40-60 and 60-80 cm at both sites in October 2009, before trial establishment. Composite samples were taken to represent each trial site: four on the Oxisol, and three on the Inceptisol. Soils were analysed by the Fertiliser Advisory Service (FAS) at the South African Sugarcane Research Institute (SASRI) according to the methods given in Appendix 3.1. Topsoils (0-20 cm) were also sampled and analysed after each harvest to monitor nutrient levels.

3.2.3. Trial establishment and treatments

Both trials were established during summer 2009/2010 (Table 3.1). South Africanbred sugarcane varieties N37 (Oxisol) and N39 (Inceptisol) were chosen due to their general suitability to the soils and areas (SASRI, 2006a and b). At each trial site, three crops of sugarcane were harvested: the 'plant crop' (the first crop harvested after planting) and the first two ratoon crops (ratoons one and two) (Figure 3.2).

The trials were factorial, and four replications of each combination of nutrients were broadcast in a randomised complete block design. Potassium (as potassium chloride) was applied at rates of 0, 100, 200 and 300 kg K ha⁻¹ at both sites, to all crops. The FAS recommendation was 250 kg K ha⁻¹ (plant crop) and approximately 200 kg K ha⁻¹ (ratoon crops) for both soils. The N rates (as urea) were 0, 80 and 160 kg N ha⁻¹ on the Oxisol, and 0, 105 and 210 kg ha⁻¹ on the Inceptisol. The middle N rates corresponded to the FAS recommendation for each soil. At both sites 300 kg Si ha⁻¹ (as Calmasil®, a calcium silicate slag) was applied to the plant crop.

This was the standard FAS recommendation for any soil where Si was below the threshold of 15 mg L⁻¹ (Miles, 2014). The Calmasil® treatment also provided approximately 870 kg ha⁻¹ and 195 kg ha⁻¹ of calcium (Ca) and magnesium (Mg), respectively.

Due to the shape of the fields, treatment application and incorporation differed between the two sites (Table 3.1). On the Oxisol, the K, N and Si were applied between four and ten days prior to planting, and disced into the soil to a depth of 15 cm. On the Inceptisol, use of a tractor for incorporation was not possible, and treatments were applied three weeks after planting and raked into the soil to a depth of 3-5 cm.

At each site, basal rates of phosphorus (P), zinc (Zn) and dolomitic lime were broadcast across the entire trial at planting, in accordance with site-specific recommendations. Each site received 20 kg P ha⁻¹ (as single superphosphate 10.5% P), 5 kg Zn ha⁻¹ (as zinc sulphate, 22% Zn) and 2.5 t dolomitic lime ha⁻¹. In addition, gypsum was applied to the Oxisol (5 t ha⁻¹) and the Inceptisol (3 t ha⁻¹).

Parameter	Oxisol	Inceptisol	
Sugarcane variety	N37	N39	
Date of planting	20 November 2009	12 December 2009	
Plot size	63 m ² (7 rows x 9 m, 1 m row spacing)	56 m ² (7 rows x 8 m, 1 m row spacing)	
Horizontal (end to end) spacing between plots	1 m	1 m	
Number of guard rows flanking each plot	1	1	
Number of replicates	4	4	
Total number of plots	96	96	
Method of fertiliser incorporation	¹ Tractor-drawn disc harrow (to ~15 cm)	² Raked in by hand (to ~5 cm)	

Table 3.1: Gen	eral trial detai	Is for the Oxisc	ol and Ince	ptisol sites
----------------	------------------	------------------	-------------	--------------

¹Rectangular field, allowing the trial to be pegged, treatments applied and incorporated (by tractor), and planting furrows drawn in straight lines whilst leaving the existing pegs in place. ²Curved field, necessitating planting of sugarcane first, followed by pegging and treatment application. Treatments then incorporated by hand between the rows, so as not to damage the cane. Following harvest of the plant and first ration crops, K and N fertilisers were reapplied (see Appendix 3.2 for application dates). In accordance with FAS recommendations, 30 kg P ha⁻¹ (as single superphosphate) was also broadcast across the entire trial at the Oxisol site after harvest of the first ration crop. No P was required at the Inceptisol site as soil P levels were above the threshold.

Silicon uptake in sugarcane can be limited under certain conditions (Keeping, 2017), and even increasing soil Si by Si application may not raise plant levels above local thresholds (Keeping *et al.*, 2017). Due to limited plant uptake of Si in the first two crops, the decision was made to use a different Si source after harvest of the first ratoon crop at both trial sites. Cement (CIMPOR NPC PRO blastfurnace Cement CEM III/A 32.5 N) was applied at a rate of 6 t ha⁻¹ (equivalent to 300 kg Si ha⁻¹) to all plots that had received Si previously. In addition to the Si, this quantity of cement supplied 1 600 kg Ca ha⁻¹ and 223 kg Mg ha⁻¹. The dry cement was broadcast and the plots hand-raked to incorporate the cement to approximately 3 cm. The zero Si plots, which received no cement, were also hand-raked to treat the plots equitably.

At each trial site, weeds were controlled by the grower, according to the standard procedure and herbicide formulation used on the rest of the farm.

3.2.4. Trial harvest

The crops were harvested (see Appendix 3.3 for harvest dates) according to the area, milling season and grower preference. At each harvest, the sugarcane was burnt a maximum of 12 h before sampling. After burning, 12 stalks of sugarcane were randomly selected and cut from each plot. All leaf material (including the green tops) was removed from these stalks, and the bundles sent to the SASRI millroom where percent estimated recoverable crystal (%ERC) was determined. Estimated recoverable crystal indicates the amount of sucrose that will be extracted per unit mass of sugarcane during the milling process, and is used as a measure of cane quality.

Plots were harvested manually using cane knives. The outermost edge row (guard row) at the top and bottom of each plot was left standing, and the remaining 5 net rows cut. The green tops and immature meristem area were cut off as per

commercial practice, and the harvested stalks weighed using a grab and balance mounted on a vehicle to determine yield.

At the Inceptisol site, 32 plots were accidentally sprayed with a hormonal ripening agent by the grower during the growth of the second ratoon crop, necessitating that they be destroyed. The second ratoon harvest, therefore, was conducted on only 64 plots.

3.2.5. Statistical analysis

Mean sugarcane yield (t ha⁻¹), %ERC and sucrose yield (t ERC ha⁻¹) (yield x %ERC; on which sugarcane growers' payments are based), per treatment were analysed using analysis of variance (three-way ANOVA) and, in the case of the unbalanced dataset from the Inceptisol second ratoon harvest, residual maximum likelihood (REML) (Genstat, 14th Edition, VSN International, 2011). Where treatments showed significant effects, values were analysed using the Holm Sidak multiple comparison separation or Fisher's protected least significant difference (LSD) test to identify where the differences between treatment means lay. Differences were considered to be significant at P< 0.05.

3.2.6. Economic analysis

Partial net returns (partial budgets) were calculated to determine the economic viability of various fertiliser (type and rate) applications. A partial budget allows evaluation of the financial effect of management decisions, and only includes costs that will be changed; it does not consider resources that are left unchanged (Tigner, 2018). The partial budgets included costs for fertiliser/Calmasil® purchase and application, harvesting and transport (Tweddle, 2020), and increases in sucrose yield (t ERC ha⁻¹).

3.3. Results and discussion

3.3.1. Initial soil conditions

Pre-trial soil conditions are summarised in Table 3.2. The Oxisol had higher clay and organic matter and thus a lower sample density than the Inceptisol. Both soils had

moderately low pH values, while acid saturation was high, particularly at depth in the Oxisol. Topsoil (0-20 cm) K was relatively low in both soils, and topsoil Si marginally lower than the FAS threshold of 15 mg L⁻¹. Topsoil Ca and Mg were above their FAS threshold values of 300 and 50 mg L⁻¹, respectively (Miles, 2014).

	Depth	Clav	Organic	nН	nH	^a Total	^b Acid sat	Sample
Soil	(cm)	(g	C kg ⁻¹)	С рп .1) (CaCl ₂)		(%)	density (g mL⁻¹)	
Oxisol	0-20	410 ± 12	39.0 ± 0.7	4.43 ± 0.02	8.4 ± 0.2	24.7 ± 1.1	1.09 ± 0.01	
	20-40	400 ± 4	34.3 ± 1.6	4.29 ± 0.03	7.0 ± 0.2	44.0 ± 2.7	1.11 ± 0.01	
	40-60	410 ± 10	26.7 ± 0.5	4.29 ± 0.03	7.5 ± 0.4	55.3 ± 5.1	1.14 ± 0.01	
	60-80	420 ± 25	22.1 ± 1.1	4.30 ± 0.04	5.2 ± 0.2	60.0 ± 5.0	1.15 ± 0.01	
Incepti -sol	0-20	240 ± 2	14.0 ± 1.3	4.17 ± 0.02	5.7 ± 0.4	23.7 ± 2.0	1.26 ± 0.01	
	20-40	250 ± 10	12.8 ± 0.5	4.06 ± 0.06	5.3 ± 0.8	39.3 ± 4.8	1.26 ± 0.01	
	40-60	280 ± 7	12.2 ± 0.6	4.37 ± 0.07	5.2 ± 1.0	22.6 ± 4.1	1.24 ± 0.01	
	60-80	390 ± 29	9.9 ± 0.2	4.46 ± 0.07	7.8 ± 0.8	19.6 ± 3.4	1.16 ± 0.01	
	Depth	Р	К	Ca	a	Mg	Si	
O I		(mg L ⁻¹)						
Soli	(cm)			(r	ng L ⁻¹)			
5011	(cm) 0-20	 52 ± 5	 55 ± 2	(r 415 ±	ng L ⁻¹)	64 ± 39	 13.8 ± 0.9	
	(cm) 0-20 20-40	 52 ± 5 24 ± 4	55 ± 2 29 ± 1	(r 415 ± 313 ±	mg L ⁻¹) 121 10 99 1	64 ± 39 18 ± 32	13.8 ± 0.9 15.2 ± 1.4	
Oxisol	(cm) 0-20 20-40 40-60	 52 ± 5 24 ± 4 8 ± 2	55 ± 2 29 ± 1 17 ± 2	(r 415 ± 313 ± 264 ±	ng L ⁻¹) 121 1 - 99 1 - 54 1	64 ± 39 18 ± 32 09 ± 20	13.8 ± 0.9 15.2 ± 1.4 11.8 ± 1.7	
Oxisol	(cm) 0-20 20-40 40-60 60-80	 52 ± 5 24 ± 4 8 ± 2 9 ± 1	55 ± 2 29 ± 1 17 ± 2 13 ± 0	(r 415 ± 313 ± 264 ± 243 ±	ng L ⁻¹) 121 1 - 99 1 - 54 1 - 54 1	64 ± 39 18 ± 32 09 ± 20 02 ± 21	13.8 ± 0.9 15.2 ± 1.4 11.8 ± 1.7 7.2 ± 1.9	
Oxisol	(cm) 0-20 20-40 40-60 60-80 0-20	52 ± 5 24 ± 4 8 ± 2 9 ± 1 16 ± 5	$55 \pm 2 \\ 29 \pm 1 \\ 17 \pm 2 \\ 13 \pm 0 \\ 44 \pm 3$	(r 415 ± 313 ± 264 ± 243 ± 324 ±	ng L ⁻¹) 121 1 2 99 1 2 54 1 54 1 55 1	64 ± 39 18 ± 32 09 ± 20 02 ± 21 10 ± 9	13.8 ± 0.9 15.2 ± 1.4 11.8 ± 1.7 7.2 ± 1.9 12.3 ± 1.1	
Oxisol	(cm) 0-20 20-40 40-60 60-80 0-20 20-40	$ 52 \pm 5 24 \pm 4 8 \pm 2 9 \pm 1 16 \pm 5 6 \pm 1 $	55 ± 2 29 ± 1 17 ± 2 13 ± 0 44 ± 3 41 ± 2	(1 415 ± 313 ± 264 ± 243 ± 324 ± 227 ±	ng L ⁻¹) 121 1 99 1 54 1 54 1 55 1 47 7	64 ± 39 18 ± 32 09 ± 20 02 ± 21 10 ± 9 71 ± 12	13.8 ± 0.9 15.2 ± 1.4 11.8 ± 1.7 7.2 ± 1.9 12.3 ± 1.1 13.7 ± 1.2	
Oxisol Incepti -sol	(cm) 0-20 20-40 40-60 60-80 0-20 20-40 40-60	52 ± 5 24 ± 4 8 ± 2 9 ± 1 16 ± 5 6 ± 1 8 ± 3	55 ± 2 29 ± 1 17 ± 2 13 ± 0 44 ± 3 41 ± 2 40 ± 2	(r 415 ± 313 ± 264 ± 243 ± 324 ± 227 ± 238 ±	$\begin{array}{c} \text{ng } \text{L}^{-1} \text{)}\\ 121 & 10\\ \hline 99 & 1\\ \hline 54 & 10\\ \hline 55 & 1\\ \hline 55 & 1\\ \hline 47 & 7\\ \hline 64 & 7\end{array}$	64 ± 39 18 ± 32 09 ± 20 02 ± 21 10 ± 9 71 ± 12 77 ± 15	13.8 ± 0.9 15.2 ± 1.4 11.8 ± 1.7 7.2 ± 1.9 12.3 ± 1.1 13.7 ± 1.2 8.3 ± 1.5	

Table 3.2: Selected pre-trial soil properties (mean \pm s.e.) of the Oxisol (n=4) and the Inceptisol (n=3)

^aTotal cations = sum of (Al^Ψ + H^Ψ + K + Ca + Mg + Na^Ψ); ^ΨData not shown ^bAcid saturation = 100 * [(Al + H) / (Al + H + K + Ca + Mg + Na)]

3.3.2. Effect of potassium on sugarcane harvest characteristics

3.3.2.1. Oxisol

3.3.2.1.1. Stalk yield

Yields of the plant, first and second ratoon crops, and when all three crop yields were combined, showed no significant differences at any of the K rates applied (data not shown). However, when only the two ratoon crops were combined, the yield from the 300 kg K ha⁻¹ treatment (103.1 t ha⁻¹) was significantly (P < 0.05) greater than the zero K treatment (97.7 t ha⁻¹). Neither of the intermediate K rates yielded

significantly higher than the control (data not shown). Removal of K by successive crops in the three lowest K treatments led to increasing K deficiency, with the consequent response to K becoming significant in the later crops. Du Toit (1957) and Wood and Meyer (1986) also noted that ration crops responded more to added K than plant crops.

3.3.2.1.2. Sucrose percent

As was the case for yield, the first two crops on the Oxisol showed no effect of K application rate on sucrose percent (%ERC), but in the second ratoon, all rates of K increased %ERC significantly compared to the control (Figure 3.3). The sum of all crops, as well as the ratoon crops only, showed a similar pattern. The effect of K on %ERC is not surprising, given the role that K plays in the stimulation of phloem transport of sugars (Kingston, 2000; Wood and Schroeder, 2004). This effect has not been widely reported in the industry, however, perhaps because the %ERC of cane will not usually increase with increased K application on K-sufficient soils (Du Toit, 1960).



Figure 3.3: Oxisol: The effect of potassium (K) application rate on percent estimated recoverable crystal (%ERC) for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.001).

3.3.2.1.3. Sucrose yield

Although sugarcane stalk yields (t cane ha⁻¹) on the Oxisol were generally unaffected by K application, the significant effect of K on %ERC resulted in similar significant effects of K on the tonnes of sucrose obtained (Figure 3.4). The t ERC ha⁻¹ were significantly increased by K application rates of 200 and 300 kg ha⁻¹, compared to the zero K treatment, in the second ratoon, the average over all crops, and the ratoon crops only.



Figure 3.4: Oxisol: The effect of potassium (K) application rate on sucrose yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (*P < 0.05; **P < 0.01; ***P < 0.001).

3.3.2.2. Inceptisol

3.3.2.2.1. Stalk yield

In contrast to the Oxisol, K application to the Inceptisol significantly increased sugarcane yield in the first ration, as well as when all crop data were combined and when the rations only were considered (Figure 3.5). In each case, yields from the 200 kg K ha⁻¹ treatment were significantly greater than those from the control plots; neither the 100 nor the 300 kg K ha⁻¹ treatments were significantly different to the control. As found on the Oxisol, ration crop yields responded more to K than the plant crop.



Figure 3.5: Inceptisol: The effect of potassium (K) application rate on sugarcane stalk yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.05).

3.3.2.2.2. Sucrose percent and yield

In contrast to the Oxisol, there was no significant effect of K on %ERC in any of the crops or crop combinations on the Inceptisol (data not shown). Application of K to the Inceptisol led to increases in sucrose yield in all crops and crop combinations except the second ratoon (Figure 3.6). In contrast to the Oxisol, however, these increases were attributable to increases in cane yield rather than %ERC. Compared to the control, sucrose yields on the Inceptisol were significantly increased by 200 kg K ha⁻¹ in the first ratoon, all crops together and ratoon crops only, and by 300 kg K ha⁻¹ in the plant crop.



Figure 3.6: Inceptisol: The effect of potassium (K) application rate on sucrose yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.05).

3.3.2.3. Economic aspects

On the Oxisol, the 200 kg K ha⁻¹ treatment yielded an additional 1.24 t ERC ha⁻¹ in the second ratoon; the plant crop and first ratoon were unaffected by K rate. This K application would therefore have resulted in a loss of R3 377 ha⁻¹ in comparison with the zero treatment (Table 3.3A).

On the Inceptisol, the addition of 200 kg K ha⁻¹ yielded an additional 1.15 t ERC ha⁻¹ in the first ration. The second ration was unaffected by K and the plant crop only showed a sucrose yield increase with 300 kg K ha⁻¹. Potassium application at 200 kg ha⁻¹ would therefore have resulted in a loss of R3 665 ha⁻¹ over the three crops (Table 3.3B).

A: Oxisol						
¹ Added income due to change:	(R ha⁻¹)	¹ Added costs due to change:	(R ha⁻¹)			
Additional 1.24 t ERC ha ⁻¹	5 518	Cost of K fertiliser (R12 kg ⁻¹ K)	7 200			
(R4 450 t ⁻¹ ERC)		(Potassium chloride)				
		Cost of K application (R0.26 kg⁻¹ KCl [€])	156			
		Harvest cost (manual harvest) (² R88.90 t ⁻¹ cane)	919			
		Transport cost (² R60.03 t ⁻¹ cane)	620			
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)			
None		None				
Subtotal	5 518	Subtotal	8 895			
Net change: R5 518 – R8 895 = –R3 377 ha ⁻¹						
B: Inceptisol						
¹ Added income due to change:	(R ha⁻¹)	¹ Added costs due to change:	(R ha⁻¹)			
Additional 1.15 t ERC ha ⁻¹	5 118	Cost of K fertiliser (R12 kg ⁻¹ K)	7 200			

Table 3.3: Partial budget for the application of 200 kg K ha⁻¹ (as opposed to 0 kg K ha⁻¹) in the plant and first two ratoon crops on the (A) Oxisol and (B) Inceptisol

(R4 450 t ⁻¹ ERC)		(Potassium chloride)			
		Cost of K application (R0.26 kg⁻¹ KCl [€])	156		
		Harvest cost (manual harvest) (² R88.90 t ⁻¹ cane)	852		
		Transport cost (² R60.03 t ⁻¹ cane)	575		
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha ⁻¹)		
None		None			
Subtotal	5 118	Subtotal	8 783		
Net change: R5 118 – R8 783 = –R3 665 ha ⁻¹					

¹Sum over three crops

²At 12% estimated recoverable crystal, such that 8.33 t cane ha⁻¹ produced 1 t ERC ha⁻¹.

[€]Tin and string method (see Glossary)

Figure 3.7 shows that in the plant crop, the 200 kg K ha⁻¹ application rate resulted in very little sucrose yield (t ERC ha⁻¹) response on either soil but by the second ratoon, the response was noticeably greater on the Oxisol. The yield response to K may have continued into further ratoon crops, and possibly even increased as K deficiency became more marked in the zero treatment. The economics of K application at both sites might therefore become more favourable in the latter part of

the crop cycle. Successful farming hinges upon a balance between short- and longterm sustainability. Although a reduction in K application may save money in the short term, it is important that soils are not mined, leading to very low soil K that compromises longer-term sustainability.



Figure 3.7: Sucrose yield (t ERC ha⁻¹) response to the 200 kg K ha⁻¹ application rate as a percentage of the 0 kg K ha⁻¹ control for the three crops on the Oxisol and the Inceptisol. n=24 in all crops except Inceptisol second ration, where n=16.

3.3.3. Effect of nitrogen on sugarcane harvest characteristics

3.3.3.1. Oxisol 3.3.3.1.1. Stalk yield

Application of 80 kg N ha⁻¹ to the Oxisol had a significant depressive effect on the plant crop yield (Figure 3.8). The topsoil had high organic matter (3.9% organic carbon) (Table 3.2), and during field preparation a considerable amount of N could have been released. Approximately 82-88 kg N ha⁻¹ was released from an Inanda (Oxisol) topsoil (Wood, 1964, 1965), and Wood (1965) noted that, on certain soils, the rate of N mineralisation may be so high that no yield response is obtained from N application. There is thus little surprise that plant crop yields were not improved by the application of N as the crop's N needs were already supplied by the soil. The reason for the depression (rather than the absence of an increase) is less clear. Wood (1964) refers to "[a] negative or small response to applied nitrogen shown by

plant cane", and concluded that high N release from the soil was responsible, although a full explanation was not given.



Figure 3.8: Oxisol: The effect of nitrogen (N) application rate on sugarcane stalk yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (*P < 0.05; ***P < 0.001).

Nitrogen rate did not influence yield significantly in the first ratoon crop on the Oxisol (Figure 3.8). Wood (1964) stated that mineralisation effects were likely to persist into ratoon cane on heavier soils with high organic matter content. In the second ratoon, however, and for all crops in combination, as well as the ratoon crops only, N application significantly increased sugarcane yield (Figure 3.8) but only at 80 kg ha⁻¹. Doubling the N rate had no further effect on yield.

3.3.3.1.2. Sucrose percent

In every crop grown on the Oxisol, N application reduced %ERC (Figure 3.9). This reduction was significant in all crops except the first ratoon. In the plant crop, 80 kg N ha⁻¹ significantly reduced %ERC, while in the second ratoon crop, only the higher rate of 160 kg ha⁻¹ led to a significant reduction. The amount of N mineralised during land preparation was evidently sufficient to decrease %ERC with only a small (80 kg ha⁻¹) addition of N to the plant crop. By the second ratoon crop, however, N

mineralisation would have been lower so that only the larger application of N (160 kg ha⁻¹) was sufficient to cause sucrose reduction.



Figure 3.9: Oxisol: The effect of nitrogen (N) application rate on percent estimated recoverable crystal (%ERC) for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (*P < 0.05; ***P < 0.001).

The negative effect of N on sucrose concentration is characteristic of most sugarcane varieties, and has been widely reported (Singh, 1973; Muchow *et al.*, 1996; Meyer and Wood, 2001; Nassar *et al.*, 2005; Inoue *et al.*, 2009). Increasing N application tends to increase crop yields, lowering stalk dry matter content with a concomitant reduction in %ERC through dilution (Muchow *et al.*, 1996). Meyer and Wood (2001) reported an average 0.38 percentage unit decrease in sucrose per 50 kg of applied N, at application rates ranging from 50 to 150 kg N ha⁻¹. It was not specified whether this average was for plant or ratoon crops, or both. In the present study, the equivalent value for the Oxisol's ratoon crops was a 0.21 percentage unit drop in sucrose, per 50 kg applied N, from 80 to 160 kg ha⁻¹ applied N. The difference is likely accounted for by variety, amongst other factors, as reported by Meyer and Wood (2001).

3.3.3.1.3. Sucrose yield

Sucrose yield (t ERC ha⁻¹) on the Oxisol was significantly reduced by N application in the plant crop (Figure 3.10), but in the later crops there was an overall increase in sucrose yield, significant in the second ratoon and combined ratoon crops. The plant crop reduction in total sucrose yield is attributable to the reduction in %ERC with N application, while the significant cane yield increases in the ratoon crops resulted in an overall increase in sucrose yield. Muchow *et al.* (1996) and Nassar *et al.* (2005) reported similar results where, although increased N application led to a decrease in sucrose yield.



Figure 3.10: Oxisol: The effect of nitrogen (N) application rate on sucrose yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.05).

3.3.3.2. Inceptisol

3.3.3.2.1. Stalk yield

The application of N to the Inceptisol did not affect sugarcane yield in the plant crop (Figure 3.11). With a topsoil (0-20 cm) organic carbon content of 1.4% (Table 3.2), approximately $45 - 60 \text{ kg N ha}^{-1}$ could have been released during cultivation and
land preparation, according to figures reported by Wood (1965). This would have partly satisfied the N requirement of the plant crop.

Both ratoon crops and the crop combinations showed a significant increase in sugarcane yield with N application (Figure 3.11). In the first ratoon, a significant response was found up to the highest N level (210 kg N ha⁻¹). In the second ratoon and crop combinations, however, no further significant yield response was recorded above 105 kg N ha⁻¹. In this regard, the two trial sites were similar in that the intermediate N level (which would have been recommended at each site by the FAS) was sufficient for the crop.



Figure 3.11: Inceptisol: The effect of nitrogen (N) application rate on sugarcane stalk yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.001).

3.3.3.2.2. Sucrose percent

In sharp contrast to the Oxisol, N application to the Inceptisol did not affect %ERC in any of the crops or crop combinations (data not shown), except in the second ration crop, where %ERC increased at the highest N level. The mean of 13.2 ± 0.1 %ERC at 0 kg N ha⁻¹ was significantly lower than 13.8 ± 0.2 %ERC at 210 kg N ha⁻¹, while the intermediate N level of 105 kg N ha^{-1} , at 13.4 ± 0.1 %ERC, did not differ

significantly from either. This increase with N in the second ratoon, with no significant effects in the other crops, may have indicated N deficiency in the zero control.

3.3.3.2.3. Sucrose yield

Sucrose yield (t ERC ha⁻¹) on the Inceptisol was significantly increased by N application in all crops and crop combinations, except for the plant crop (Figure 3.12). With the exception of the first ratoon, where sucrose yield increased up to the highest N rate, there was no further response to N application above 105 kg N ha⁻¹. Sucrose yield increases mirrored the stalk yield increases, indicating the underlying cause of this response. Again, this pattern was similar to those reported by Muchow *et al.* (1996) and Nassar *et al.* (2005).



Figure 3.12: Inceptisol: The effect of nitrogen (N) application rate on sucrose yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within crops or crop combinations, columns with different letters are significantly different to each other (P < 0.001).

3.3.3.3. Economic aspects

A partial budget, calculated to determine the economic effect of applying 80 kg N ha⁻¹ to each crop grown on the Oxisol (Table 3.4A), indicated a cost : benefit ratio of 1 : 1.43 when compared to zero N application. As with K, however, the

response to N in the ratoons was greater than that in the plant crop (Figure 3.13). When only the ratoons were considered, the cost : benefit ratio increased to 1 : 2.05 (Table 3.4B).

Table 3.4: Oxisol: Partial budget for the application of 80 kg N ha⁻¹ (as opposed to 0 kg N ha⁻¹) in the (A) plant and first two ratoon crops and (B) first and second ratoon crops only

A: Plant, first ratoon and second ratoon crops:				
¹ Added income due to change:	(R ha⁻¹)	¹ Added costs due to change:	(R ha⁻¹)	
Additional 1.35 t ERC ha ⁻¹	6 007	Cost of N fertiliser (R10 kg ⁻¹ N)	2 400	
(R4 450 t ⁻¹ ERC)		(Urea)		
		Cost of N application (R0.26 kg⁻¹ urea [€])	136	
		Harvest cost (manual harvest) (³ R88.90 t ⁻¹ cane)	1000	
		Transport cost (³ R60.03 t ⁻¹ cane)	675	
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)	
None		None		
Subtotal	6 007	Subtotal	4 211	
Net change: R6 007 – R4 211 = R1 796 ha ⁻¹				

B: First ratoon and second ratoon crops only:					
² Added income due to change:	(R ha⁻¹)	² Added costs due to change:	(R ha⁻¹)		
Additional 1.82 t ERC ha ⁻¹	8 099	Cost of N fertiliser (R10 kg ⁻¹ N)	1 600		
(R4 450 t ⁻¹ ERC)		(Urea)			
		Cost of N application (R0.26 kg⁻¹ urea [€])	90		
		Harvest cost (manual harvest) (³ R88.90 t ⁻¹ cane)	1 348		
		Transport cost (³ R60.03 t ⁻¹ cane)	910		
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)		
None		None			
Subtotal	8 099	Subtotal	3 948		
Net change: R8 099 – R3 948 = R4 151 ha ⁻¹					

¹Sum over three crops ²Sum over two crops

³At 12% estimated recoverable crystal, such that 8.33 t cane ha⁻¹ produced 1 t ERC ha⁻¹. [€]Tin and string method



Figure 3.13: Sucrose yield (t ERC ha⁻¹) response to the intermediate N application rates (80 kg ha⁻¹ (Oxisol), 105 kg ha⁻¹ (Inceptisol)), as a percentage of the 0 kg ha⁻¹ N control for the three crops on each soil. n=32 in all crops except Inceptisol second ratoon, where n=19.

On the Oxisol, N application of either 80 or 160 kg N ha⁻¹ to the plant crop led to a significant decrease in sucrose yield (Figure 3.10). This response was likely caused by the high organic matter in the topsoil that released a substantial amount of N during land preparation. It is therefore necessary to consider the financial implications of applying no N to the plant crop on the Oxisol. In comparison to the application of 80 kg N ha⁻¹, zero N application would have saved fertiliser costs and negated the yield 'penalty' associated with N application. Applying no N to the plant crop on the Oxisol would thus have resulted in a financial benefit of R2 354 ha⁻¹ over the application of 80 kg N ha⁻¹ (Table 3.5).

Added income due to change:	(R ha⁻¹)	Added costs due to change:	(R ha⁻¹)		
Additional 0.47 t ERC ha ⁻¹ (R4 450 t ⁻¹ ERC)	2 092	Harvest cost (manual harvest) (¹ R88.90 t ⁻¹ cane)	348		
(Transport cost (¹ R60.03 t ⁻¹ cane)	235		
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)		
Cost of N fertiliser (R10 kg ⁻¹ N) (Urea)	800	None			
Cost of N application (R0.26 kg ⁻¹ urea [€])	45				
Subtotal	2 937	Subtotal	583		
Net change: R2 937 – R583 = R2 354 ha ⁻¹					

Table 3.5: Oxisol: Partial budget for the application of 0 kg N ha⁻¹ (as opposed to 80 kg N ha⁻¹) in the plant crop

¹At 12% estimated recoverable crystal, such that 8.33 t cane ha⁻¹ produced 1 t ERC ha⁻¹. [€]Tin and string method

On the Inceptisol, application of 105 kg N ha⁻¹ to each crop yielded an additional 4.2 t ERC ha⁻¹ over that of the zero N control (Table 3.6), resulting in a cost : benefit ratio of 1 : 2.19. Again the response to N in the ratoons was greater than that in the plant crop (Figure 3.13). At both sites if the patterns of N response (Figure 3.13) were to continue into later ratoons, the economics of N application would likely remain favourable.

Table 3.6: Inceptisol: Partial budget for the application of 105 kg N ha^{-1} (as opposed to 0 kg N ha⁻¹) in the plant and first two ratoon crops

A: Plant, first ratoon and second ratoon crops:					
¹ Added income due to change:	(R ha⁻¹)	¹ Added costs due to change:	(R ha⁻¹)		
Additional 4.20 t ERC ha ⁻¹ (R4 450 t ⁻¹ ERC)	18 690	Cost of N fertiliser (R10 kg ⁻¹) (Urea)	3 150		
		Cost of N application (R0.26 kg⁻¹ urea [€])	178		
		Harvest cost (manual harvest) (² R88.90 t ⁻¹ cane)	3 111		
		Transport cost (² R60.03 t ⁻¹ cane)	2 101		
Reduced costs due to change:	(R ha ^{₋1})	Reduced income due to change:	(R ha ⁻¹)		
None		None			
Subtotal	18 690	Subtotal	8 540		
Net change: R18 690 – R8 540 = R10 150 ha ⁻¹					

¹Sum over three crops

²At 12% estimated recoverable crystal, such that 8.33 t cane ha⁻¹ produced 1 t ERC ha⁻¹. [€]Tin and string method

3.3.4. Effect of silicon on sugarcane harvest characteristics

3.3.4.1. Oxisol

3.3.4.1.1. Stalk yield, sucrose percent and sucrose yield

The effect of the application of 300 kg Si ha⁻¹ was limited (results not shown). Stalk yield was significantly increased by Si application in the second ratoon crop only, from 77.7 t cane ha⁻¹ (control), to 81.8 t cane ha⁻¹. Silicon had no effect on %ERC in any of the harvested crops. Silicon application does not commonly result in increased sucrose concentration in sugarcane. According to Korndörfer and Lepsch (2001), the yield increases associated with Si application are generally due to longer stalks, larger stalk diameters and an increase in stalk number, rather than an increase in pol (apparent sucrose) percent. The increase in stalk yield in the second ratoon translated into a significant increase in sucrose yield in the same crop, from 11.09 t ERC ha⁻¹ (control) to 11.65 t ERC ha⁻¹.

The inconsistent response to Si was unexpected, considering the extensive literature reporting improvements in sugarcane stalk and sucrose yield after application of Si (as calcium silicate slag, in most cases) (Clements, 1965; Ayres, 1966; Fox *et al.*, 1967; Gascho and Andreis, 1974; Ross *et al.*, 1974; Berthelsen *et al.*, 2001b (calcium silicate slag and cement); Korndörfer and Lepsch, 2001). Highly weathered

soils (Ross *et al.*, 1974; Korndörfer and Lepsch, 2001) and soils with low Si levels (Ross *et al.*, 1974; Berthelsen *et al.*, 2001b) were reported to be particularly likely to respond. Silicon deficiencies are common in sugarcane grown on the KwaZulu-Natal north coast (Van der Laan and Miles, 2010), where the trial was situated. The limited response to Si obtained may be due to poor plant uptake of Si on this soil, possibly due to its moderately high acid saturation (Table 3.2).

3.3.4.2. Inceptisol

3.3.4.2.1. Stalk yield, sucrose percent and sucrose yield

The effect of Si application was more pronounced on the Inceptisol than the Oxisol. Stalk yield on the Inceptisol was significantly increased by the application of Calmasil® or cement in all crops and crop combinations except the plant crop (Figure 3.14A). In the second ratoon and both crop combinations, the differences in stalk yield were highly significant (P < 0.01) and the average yield increase with Si application in the ratoon crops was 5.3 t cane ha⁻¹. Silicon application did not affect %ERC in any of the crops, but sucrose yield was significantly increased in all crops and combinations, by an average of 0.69 t ERC ha⁻¹ crop⁻¹ (Figure 3.14B).

This response in stalk and sucrose yield to Si application was expected although the Si in the Inceptisol topsoil, at 12.3 mg L⁻¹ (Table 3.2), was close to that of the Oxisol (13.8 mg L⁻¹), making the lack of response to Si on the Oxisol even more puzzling. The response to Si on the Inceptisol might be linked to slightly improved Si uptake by the plant (Section 4.3.3.1) which, in turn, could have been facilitated by lower acid saturation in the Inceptisol (26% mean acid saturation over 0-80 cm compared to 46% in the Oxisol).



Figure 3.14: Inceptisol: The effect of silicate application on (A) sugarcane yield and (B) sucrose yield for each individual crop (Plant Crop, Ratoon 1, Ratoon 2), the average over all three crops (All Crops) and the ratoon crops (Ratoons Only). Within sites and crops or crop combinations, columns with different letters are significantly different to each other (*P < 0.05; **P < 0.01; ***P < 0.001).

It is not clear whether the increase in sucrose yield obtained on the Inceptisol was due to the Si in the amendments, or the other constituents (Ca and Mg). Prior to establishment, soil Ca and Mg at both trial sites, at all soil depths, were greater than the FAS thresholds. Dolomitic lime, applied at each site before crop establishment, increased the soil Ca and Mg further. In addition, Ca and Mg in leaf samples collected at both sites, during each crop, were, in every case, greater than the FAS threshold levels (Section 4.3.3.3). It is likely therefore that the yield responses observed in the Calmasil® / cement treatments were due to Si, although it is also possible that the crop might have responded to the increased Ca, even though leaf concentrations were above the threshold.

3.3.4.3. Economic aspects

Costs were calculated based on Calmasil®, as cement is expensive and would be unlikely to be applied by a commercial agricultural enterprise. Applications of Si on the Oxisol yielded only 0.99 t ERC ha⁻¹ extra over three crops, generating a loss of R5 208 ha⁻¹ (Table 3.7A). On the Inceptisol, an application of 300 kg Si ha⁻¹ to the plant crop, and a second application prior to the second ratoon, would have resulted in a loss of R1 711 ha⁻¹ over the three crops combined (Table 3.7B). In practice, Calmasil® would usually be applied in place of (not in addition to) dolomitic lime. The difference in cost would therefore only be that between the cost of dolomitic lime and Calmasil® (R354 ton⁻¹ versus R308 ton⁻¹, respectively, excluding delivery costs), along with the harvest and transport costs associated with any increase in yields with Si application. Liming materials, and Calmasil® in particular, are not commonly topdressed during the ratoon crops, so any further yield increases accruing to Calmasil® application would be obtained without further application costs. **Table 3.7:** Partial budget for the application of 300 kg Si ha⁻¹ (as opposed to 0 kg Si ha⁻¹), as Calmasil®, in the plant crop and second ration crop on the A) Oxisol and B) Inceptisol

A: Oxisol:			
¹ Added income due to change:	(R ha⁻¹)	Added costs due to change:	(R ha ⁻¹)
Additional 0.99 t ERC ha ⁻¹ (R4 450 t ⁻¹ ERC)	4 406	² Cost of Si fertiliser (R7 kg ⁻¹ Si) (Calmasil®)	4 200
		² Cost of Si application (R165 t ⁻¹)	990
		² Cost of Si incorporation (R1598 ha ⁻¹)	3 196
		¹ Harvest cost (manual harvest) (³ R88.90 t ⁻¹ cane)	733
		¹ Transport cost (³ R60.03 t ⁻¹ cane)	495
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)
None		None	
Subtotal	4 406	Subtotal	9 614

Net change: R4 406 – R9 614 = - R5 208 ha⁻¹

B: Inceptisol:				
¹ Added income due to change:	(R ha⁻¹)	Added costs due to change:	(R ha ⁻¹)	
Additional 2.08 t ERC ha ⁻¹ (R4 450 t ⁻¹ ERC)	9 256	² Cost of Si fertiliser (R7 kg ⁻¹ Si) (Calmasil®)	4 200	
		² Cost of Si application (R165 t ⁻¹)	990	
		² Cost of Si incorporation (R1598 ha ⁻¹)	3 196	
		¹ Harvest cost (manual harvest) (³ R88.90 t ⁻¹ cane)	1 541	
		¹ Transport cost (³ R60.03 t ⁻¹ cane)	1 040	
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)	
None		None		
Subtotal	9 256	Subtotal	10 967	
Net change: R9 256 – R10 967 = - R1 711 ha ⁻¹				

¹Sum over three crops ²Sum over two applications ³At 12% estimated recoverable crystal, such that 8.33 t cane ha⁻¹ produced 1 t ERC ha⁻¹.

3.3.5. Effect of nutrient interactions on sugarcane harvest characteristics

3.3.5.1. Oxisol

There were generally few interactions between nutrients in terms of their effect on harvest characteristics. One exception was the interactive effect of N and Si on %ERC in the second ration crop on the Oxisol (Figure 3.15).



Figure 3.15: Oxisol: Second ratoon crop. The effect of nitrogen (N) application rate on percent estimated recoverable crystal (%ERC) at different rates of silicon (Si). Within and between Si rates, points with different letters are significantly different to each other (P < 0.05).

With zero applied Si, increasing N had no effect on %ERC. Where Si was applied, however, increasing N caused a significant decrease. However, this was most likely to be an effect of the Si-containing liming material, rather than the Si itself. Calmasil® contains Ca and Mg carbonates, along with Si. Cement similarly contains a large amount of Ca, and some Mg. It is therefore likely that the increased soil N release that often accompanies liming (Lyngstad, 1992; Garbuio *et al.*, 2011) increased the amount of available N to such an extent in the highest N treatment, that sucrose content was significantly compromised. Without Calmasil® or cement application, soil N was insufficient to reduce the sugarcane quality in this way. This effect was only seen in the second ratio crop, which had cement applied during early growth.

There was also a significant interaction between the effect of K, N and Si on %ERC in the second ratoon crop, as well as when ratoon crops, and all crops, were combined. The highest level of N, with zero K, lowered %ERC in the crop. At a higher rate of K (200 kg K ha⁻¹), however, the effects of high (160 kg ha⁻¹) N were mitigated and, in that treatment, %ERC was significantly higher than where no K was applied (Figure 3.16). This effect was not evident in the zero Si plots. These results show that, under certain conditions, the deleterious effect of high N rates on %ERC can be mitigated by the application of sufficient K. Qualifying conditions may include soils where a low N response is likely, i.e., those with high organic matter, such as the Oxisol. Similar effects were reported by Meyer and Wood (2001), and are likely linked to the role that K plays in the stimulation of phloem transport of sugars (Wood and Schroeder, 2004).



Figure 3.16: Oxisol: Ratoon crops only. The effect of nitrogen (N) and potassium (K) application rates on percent estimated recoverable crystal (%ERC) at 300 kg Si ha⁻¹. All treatment values marked with an asterisk* are significantly higher (P < 0.05) than the circled treatment. No other significant differences were found.

3.3.5.2. Inceptisol

On the Inceptisol, there were no significant interactions between any of the applied nutrients in terms of cane yield, cane quality or sucrose yield.

3.4. Conclusions

Different sugarcane crop yield and quality responses were obtained to the nutrients applied to the Oxisol and the Inceptisol. On the Oxisol, K application increased overall sucrose yield, driven by increased %ERC, without affecting stalk yield. This effect has not been widely reported in the industry. On the Inceptisol, however, the increase in sucrose yield with K application was due to increases in cane yield, without affecting %ERC. At both sites, the increased sucrose yields occurred only up to 200 kg K ha⁻¹.

The current K recommendation given by the FAS is linked to the attainable yield and soil clay percent, and would have been 250 kg K ha⁻¹ in the plant crop at both sites, while the ratoon recommendations would have been 200 kg K ha⁻¹ (Oxisol) and 190 kg K ha⁻¹ (Inceptisol). Potassium application rates similar to those recommended by the FAS, significantly increased sucrose yields but were not economically justifiable in the short term (three crops). It is possible, however, that yield response to K would have become greater in later ratoons, particularly on the Oxisol.

Sucrose yield in the ratoons at both sites responded favourably up to intermediate levels of N (80 kg ha⁻¹ on the Oxisol, 105 kg ha⁻¹ on the Inceptisol). Nitrogen application increased sugarcane yields at both sites in the ratoons, but not in the plant crops. Application of N depressed %ERC in all crops on the Oxisol, but not on the Inceptisol, with its lower organic matter levels.

The current FAS recommendation for N is linked to the attainable yield and soil organic matter and clay contents. Nitrogen recommendations for the Oxisol would have been 70 kg N ha⁻¹ in the plant crop, and 110 kg N ha⁻¹ in the ratoons; for the Inceptisol 115 kg N ha⁻¹ in the plant crop, and 155 kg N ha⁻¹ in the ratoons. Application of N at, in general, slightly lower levels than the FAS recommendations, proved economically beneficial, particularly on the Inceptisol. Given the range of environmental and varietal conditions that would affect results at each site, the current FAS recommendations for N appear valid. The exception to this was the plant crop on the Oxisol, where a zero N application would have resulted in the highest sucrose yields. This finding is of importance, given that the FAS currently recommends applying N to every plant crop. The economic analysis conducted in this study indicates that, because N application resulted in reduced sucrose yields on

the Oxisol, it was financially detrimental to apply N. Nitrogen application did not significantly reduce sucrose yields on the Inceptisol, probably due to lower organic N reserves. A general recommendation to reduce N application to zero in the plant crop on all soils is thus unlikely to be advisable. Such a recommendation should, however, be considered for humic soils, similar to the Oxisol in this study, and perhaps other soils with a high organic matter content.

On the Oxisol, crops responded very poorly to applied Si, with few significant responses in stalk yield, %ERC or sucrose yield. This may have been due to limited Si uptake on the acidic Oxisol. On the Inceptisol, however, application of 300 kg Si ha⁻¹ led to significantly higher sucrose yields. This increase was a result of significantly greater stalk yields, rather than an increase in %ERC. On both soils Si application was not economically justified over the three crops harvested, though it may have become more economically viable in the later ratoons.

While no nutrient interactions were recorded on the Inceptisol, K, N and Si interacted significantly on the Oxisol. In general, the application of siliceous liming materials appeared to increase the amount of N available to the crop, enhancing the depressive effect of N on %ERC. Potassium, however, served to mitigate this effect on this high organic matter soil, likely due to its role in phloem transport of sugars.

CHAPTER 4

The effect of applied potassium, nitrogen and silicon on leaf nutrient concentrations and relationships with sugarcane yield

4.1. Introduction

Characterising the impact of fertiliser applications on plant nutrient concentrations, and the subsequent use of plant sample analyses to establish nutrient requirements of the crop, is key to successful crop production. In sugarcane, leaf samples are routinely analysed to determine crop nutrient status, and this plays an important role in nutrient management (Schroeder *et al.*, 1993). In South Africa, approximately 6 000 sugarcane leaf samples are processed annually by the Fertiliser Advisory Service (FAS) laboratory at the South African Sugarcane Research Institute (SASRI).

Two important aspects of leaf analysis interpretation need to be considered. Firstly, the pattern and extent of uptake of applied nutrients needs to be understood in order for leaf analyses to be of reliable diagnostic value. The extent of uptake is complicated by interactions between nutrients, both in the soil and in the plant itself, such as that between nitrogen (N) and potassium (K). Increasing N supply generally results in an increase in K uptake by the plant (De Beaucorps, 1980; Miles, 2010), which can result in significantly higher yields than when either N or K are applied alone (Gething, 1993). Phosphorus (P) availability may also increase with silicon (Si) application (Smyth and Sanchez, 1980), with possible yield implications if P were deficient. Other interactions may be antagonistic. Calcium (Ca), magnesium (Mg) and K tend to exhibit cationic antagonism, and high levels of one or more of these nutrients can result in decreased uptake of another, despite sufficient soil levels (Gosnell and Long, 1971; Marschner, 1995; Garcia *et al.*, 1999).

Application of nutrient sources which have a liming effect (e.g. Calmasil®) may induce deficiencies of other nutrients such as zinc (Zn) (Ahmad *et al.*, 2012) or copper (Cu) (Nachtigall *et al.*, 2007) due to changes in soil pH. These deficiencies may lead to yield reduction. Interpretation of the results of leaf analysis therefore depends on analysing a range of nutrients and not just those applied in trials.

Nutrient interactions need to be taken into account when considering the effects of applied nutrients as they can help to explain yield responses and highlight possible future deficiencies or excesses. The dynamics of nutrient interactions and uptake therefore have a bearing on soil threshold levels and nutrient application rates.

Secondly, results should provide some indication of potential yield (or yield loss) associated with sufficiency (or deficiency) of a particular nutrient. Leaf analysis is therefore only meaningful if reliable nutrient concentration thresholds can be determined, below which some loss of yield or quality can be expected.

This study therefore aimed to determine (i) the effect of various application rates of K, N and Si, along with their interactions, on leaf nutrient concentrations of sugarcane, and (ii) the appropriateness of the current leaf threshold values in determining nutrient sufficiency or deficiency, as determined by corresponding yields. The ultimate goal was to assist in the refinement of industry nutrient recommendations.

4.2. Materials and methods *4.2.1. Trial details*

These are given in Sections 3.2.1 to 3.2.4.

4.2.2. Leaf sampling

Leaf samples were collected at both sites when each crop was 4 and 6 months old (Figure 3.2). These crop ages fall within the prescribed range (4-7 months) suitable for leaf sampling in these areas and during the correct time from spring to late summer (Miles and Titshall, 2020), with the exception of two May samplings for the plant and first ration crops on the Oxisol. These autumn samplings were included to maintain the 4- and 6-month schedule. The plant crop on the Inceptisol was sampled when the crop was 4 months old only, in April 2010 (Figure 3.2); the 6-month sampling was cancelled due to the extremely dry conditions in winter 2010 (Section 3.2.1.1). While there are currently no rainfall threshold values below which leaf sampling is discouraged, SASRI stipulates that "The crop must have received enough well-distributed rainfall/irrigation to ensure that there is no moisture stress in the crop prior to sampling" (Miles and Titshall, 2020). The rainfall distribution at the

trial sites (Figure 3.2) indicates that prior to four of the Oxisol and two of the Inceptisol sampling times rainfall may not have been sufficient to ensure reliable leaf sample results.

At each sampling event, 30 leaves were collected per plot. The third leaf from the top was taken in each case, counting from the first leaf which was more than half unfurled (Miles, 2010). The base and tip of the leaves were cut off, leaving 20-30 cm of the central part of the leaf blade as the sample. The leaf midrib was removed within 15 minutes of sampling and discarded. Blade samples were analysed by the FAS. Nitrogen was determined by near-infrared reflectance (NIR; Bran + Luebbe InfraAlyzer 2000). X-ray fluorescence spectrometry (XRF; PW 2400 PANalytical Xray) was used to determine Ca, Cu, iron (Fe), K, Mg, manganese (Mn), P, sulphur (S), Si and Zn, and a small subset of samples from each site was analysed colorimetrically for boron (B) (Gaines and Mitchell, 1979). Mean leaf Fe and Mn (data not shown) were within the SASRI-recommended ranges of 75-99 and 15-99 mg kg⁻¹, respectively (Miles and Titshall, 2020), apart from the second ration crop on the Inceptisol, where mean Fe was below this range (69 and 54 mg kg⁻¹ at 4 and 6 months, respectively). Treatment effects on Fe and Mn were, in most cases, neither significant nor consistent, and are not reported on further. Mean leaf B was 2 and 4 mg kg⁻¹ on the Oxisol and the Inceptisol, respectively. Though the current threshold is 10 mg kg⁻¹ (Miles and Titshall, 2020), the recommended threshold at the time of the trials was 2 mg kg⁻¹ (SASRI, 2013) and so no basal B was applied to the trials.

4.2.3. Statistical analysis

Mean leaf nutrient concentrations per treatment were analysed using analysis of variance (three-way ANOVA) and, in the case of the unbalanced dataset from the Inceptisol second ration harvest, residual maximum likelihood (REML) (Genstat, 14^{th} Edition, VSN International, 2011). Data which were not normally distributed were transformed using $log_{10}(x+1)$ or square root before conducting ANOVA. Where treatments showed significant effects, the Holm Sidak multiple comparison separation or Fisher's protected least significant difference (LSD) test was used to identify where the differences between treatment means lay.

4.3. Results and discussion

4.3.1. Effect of applied potassium on leaf nutrient content

Potassium fertiliser significantly affected leaf Ca, Cu, K, Mg, S and Zn at most sampling times. Leaf sample data from the second ratoon crop on the Inceptisol, at 4 months of age, illustrate the effects of applied K on these nutrients (Table 4.1). They are discussed in the sections below.

Table 4.1: Inceptisol: Effect of applied potassium (K) on mean \pm s.e. (n = 24) third leaf K, sulphur (S), calcium (Ca), magnesium (Mg), copper (Cu) and zinc (Zn) in the second ration crop, at 4 months' crop age

K applied	К	S	Са	Mg	Cu	Zn
(kg ha⁻¹)		(%	b)		(mg	kg ⁻¹)
0	$1.08 \pm 0.12 a^{\mp}$	0.16 ± 0.00 a	0.23 ± 0.01 a	0.15 ± 0.01 a	5.32 ± 0.13 a	13.86 ± 0.33 a
100	1.31 ± 0.04 b	0.15 ± 0.00 b	0.20 ± 0.01 b	0.12 ± 0.00 b	4.97 ± 0.13 b	12.72 ± 0.20 b
200	1.40 ± 0.08 c	0.14 ± 0.00 b	0.19 ± 0.01 b	0.12 ± 0.00 b	4.95 ± 0.13 b	12.62 ± 0.20 b
300	1.46 ± 0.14 c	0.14 ± 0.00 b	0.18 ± 0.01 b	0.10 ± 0.00 c	4.91 ± 0.12 b	12.58 ± 0.20 b
Sig.	P<0.001	P<0.001	P < 0.001	P < 0.001	P<0.05	P<0.001

^{*}Within columns, values with different letters are significantly different to each other.

4.3.1.1. Potassium

Potassium fertiliser significantly increased leaf K concentrations at all sampling times at each site (Appendix 4.1). On the Oxisol, in the zero K treatment, leaf K was below the SASRI 1.05% threshold at three of the six sampling times (first ratoon, 4 and 6 months; second ratoon, 4 months). At most Oxisol sampling events, leaf K tended to reach a maximum at 100 or 200 kg K ha⁻¹; the highest K application rate (300 kg ha⁻¹) never increased leaf K significantly above that at 200 kg ha⁻¹ (Appendix 4.1). On the Inceptisol, leaf K fell below 1.05% at one of the five sampling times (first ratoon, 4 months), in the zero K treatment only. Leaf K on the Inceptisol tended to reach a maximum at 200 or 300 kg K ha⁻¹ (as in Table 4.1).

4.3.1.2. Sulphur, zinc and copper

Applied K significantly suppressed S, Zn and Cu uptake at most sampling events (as in Table 4.1). Increasing K rates were associated with significantly reduced leaf S on the Oxisol at 5 out of 6 sampling events, and on the Inceptisol at 4 out of 5 events. Sulphur is taken up by sugarcane roots as SO_4^{2-} (Meyer, 2013). Increased soil and plant concentrations of the K⁺ ion might thus be expected, by maintaining charge balance, to facilitate increased uptake of the sulphate anion, or at least not reduce it. The reduction in S with the application of KCI might, however, be linked to the chloride (Cl⁻) ions. At high soil concentrations, nonspecific competition for uptake between ions of the same charge can occur (Marschner, 1995). Potassium chloride fertiliser caused *in planta* Cl levels to increase in potatoes (James *et al.*, 1970) and tomatoes (Nukaya *et al.*, 1991) and, in sugarcane plants, Cl deficiency can be corrected by the application of KCl (Anderson and Bowen, 1990). Although not measured in the present study, it is probable that application of KCl would have increased leaf Cl. Despite suppression, however, at no sampling time did leaf S fall below the critical threshold level of 0.12% (Miles and Titshall, 2020) at either site.

Both Zn and Cu are taken up by the plant as cations (Zn^{2+}, Cu^{2+}) (Mengel and Kirkby, 2001) and the decrease in leaf Zn (4/6 events on the Oxisol; 3/5 on the Inceptisol) and Cu (3/6 on the Oxisol; 2/5 on the Inceptisol) could have been caused by ionic antagonism with the applied K. Smith (1975) reported a reduction in alfalfa leaf Zn and Cu with K application. Bowen (1969), on the other hand, reported that K stimulated their uptake by sugarcane, though this effect did not occur consistently.

At all sampling times, leaf Cu was greater than the SASRI threshold of 3.0 mg kg⁻¹ (Miles and Titshall, 2020). Leaf Zn was generally higher than the threshold of 13.0 mg kg⁻¹ (Miles and Titshall, 2020), though it was below this threshold at one of the two sampling events at each site in the second ratoon crop, where K was applied (results not shown).

4.3.1.3. Calcium and magnesium

At both sites, increased leaf K significantly suppressed leaf Ca and Mg at most sampling events. This effect is illustrated, for the second ratoon crop on the Oxisol, in Figure 4.1.



Figure 4.1: Oxisol: Relationship between leaf potassium (K) and calcium (Ca) and magnesium (Mg) in the second ratoon crop. Leaf samples collected at 6 months' crop age. Within nutrients (Ca or Mg), points with different letters are significantly different (P < 0.001). Each point represents the mean (n=24), across all replicates, at each of the four K application rates (0 kg K ha⁻¹: clear symbol; 100 kg K ha⁻¹: grey symbol; 200 kg K ha⁻¹: hatched symbol; 300 kg K ha⁻¹: black symbol).

Applied K had a more consistent suppressive effect on Mg on both soils (reduced at every sampling event) than Ca that was reduced at four out of six events and four out of five events on the Oxisol (Table 4.2) and Inceptisol (Table 4.3), respectively.

K applied	Ca and Mg					
(kg ha⁻')		(%)				
	°PC - 4	months	PC - 6	months		
	Са	Mg	Са	Mg		
0	0.308 ± 0.001	0.150 ± 0.002 a [∓]	0.316 ± 0.002	0.167 ± 0.003 a		
100	0.308 ± 0.002	0.145 ± 0.001 b	0.310 ± 0.002	0.160 ± 0.002 ab		
200	0.305 ± 0.001	0.140 ± 0.001 c	0.313 ± 0.002	0.162 ± 0.002 ab		
300	0.306 ± 0.001	0.143 ± 0.002 bc	0.311 ± 0.003	0.158 ± 0.003 b		
Sig.	NS	P<0.001	NS	P<0.05		
	^b R1 – 4	months	R1 – 6 months			
	Са	Mg	Са	Mg		
0	0.308 ± 0.002 a	0.162 ± 0.003 a	0.314 ± 0.001 a	0.167 ± 0.002 a		
100	0.300 ± 0.001 b	0.153 ± 0.003 ab	0.307 ± 0.001 b	0.150 ± 0.002 b		
200	0.302 ± 0.001 ab	0.150 ± 0.002 b	0.306 ± 0.001 b	0.147 ± 0.002 bc		
300	0.300 ± 0.002 b	0.146 ± 0.002 b	0.303 ± 0.001 b	0.143 ± 0.002 c		
Sig.	P<0.01	P<0.001	P<0.001	P<0.001		
	^c R2 – 4	months	R2 – 6 months			
	Са	Mg	Са	Mg		
0	0.383 ± 0.010 a	0.293 ± 0.008 a	0.240 ± 0.006 a	0.218 ± 0.004 a		
100	0.333 ± 0.009 b	0.235 ± 0.005 b	0.205 ± 0.005 b	0.175 ± 0.004 b		
200	0.323 ± 0.010 b	0.217 ± 0.004 bc	0.185 ± 0.004 c	0.159 ± 0.004 c		
300	0.290 ± 0.004 c	0.203 ± 0.004 c	0.183 ± 0.004 c	0.153 ± 0.004 c		
Significa	P<0.001	P<0.001	P<0.001	P<0.001		

Table 4.2: Oxisol: Effect of applied potassium (K) on mean \pm s.e. (n=24) third-leaf calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

⁺Within sampling events and columns, values with different letters are significantly different to each other.

K applied	Ca and Mg					
(kg ha⁻¹)		(%)				
	^a PC - 4	months	PC - 6	months		
	Ca	Mg	Са	Mg		
0	0.316 ± 0.002 a [∓]	0.186 ± 0.003 a	nd [#]	nd		
100	0.310 ± 0.002 ab	0.170 ± 0.003 b	nd	nd		
200	0.308 ± 0.002 bc	0.163 ± 0.002 b	nd	nd		
300	0.303 ± 0.001 c	0.162 ± 0.003 b	nd	nd		
Sig.	P < 0.001	P < 0.001	nd	nd		
	[⊳] R1 - 4	months	R1 - 6 months			
	Ca	Mg	Са	Mg		
0	0.340 ± 0.005 a	0.171 ± 0.003 a	0.333 ± 0.003	0.203 ± 0.003 a		
100	0.310 ± 0.007 b	0.156 ± 0.003 b	0.329 ± 0.002	0.199 ± 0.003 ab		
200	0.316 ± 0.004 b	0.152 ± 0.002 bc	0.331 ± 0.002	0.198 ± 0.002 ab		
300	0.309 ± 0.003 b	0.143 ± 0.002 c	0.329 ± 0.002	0.193 ± 0.001 b		
Sig.	P < 0.001	P < 0.001	NS	P < 0.05		
	°R2 - 4	months	R2 - 6 months			
	Ca	Mg	Са	Mg		
0	0.227 ± 0.006 a	0.148 ± 0.006 a	0.184 ± 0.004 a	0.141 ± 0.003 b		
100	0.198 ± 0.005 b	0.124 ± 0.004 b	0.158 ± 0.003 b	0.122 ± 0.003 b		
200	0.189 ± 0.006 b	0.117 ± 0.003 b	0.152 ± 0.003 bc	0.113 ± 0.002 c		
300	0.179 ± 0.006 b	0.103 ± 0.003 c	0.144 ± 0.003 c	0.110 ± 0.002 c		
Sig.	P < 0.001	P < 0.001	P < 0.001	P < 0.001		

Table 4.3: Inceptisol: Effect of applied potassium (K) on mean \pm s.e. (n=24) third-leaf calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

⁺Within sampling events and columns, values with different letters are significantly different to each other.

[#]nd – not determined.

Interactions between K, Ca and Mg as indicated by plant uptake are well documented. Grunes *et al.* (1992) found that concentrations of Mg and Ca in three cool-season grass species were significantly reduced after K fertilisation. Garcia *et al.* (1999) reported a reduction in grapevine leaf K with increased Ca application, and a reduction in leaf Ca and Mg with higher K application rates. Gosnell and Long (1971) attributed decreased sugarcane uptake of Ca and Mg with K fertilisation to cationic antagonism. Potassium depresses root uptake of Mg when rhizosphere K is high (Marschner, 1995) and increased K concentrations in roots have been reported to depress Mg translocation from roots to shoots, leading to low

leaf Mg (Karlen *et al.*, 1978; Wilkinson *et al.*, 2000). Decreases in Ca and Mg have also been attributed to the 'dilution effect' caused by higher yields in response to K application (Dibb and Thompson, 1985).

Most studies of K x Ca x Mg interactions have considered a two- or even three-way interaction with increased (or reduced) rates of any of the three nutrients leading to a concurrent reduction (or increase) in the other two, resulting in a more or less stable cationic total (Mengel and Kirkby, 2001). In the present study, however, K fertiliser application consistently reduced leaf Ca and Mg, while increased rates of Ca and Mg (as applied in the Calmasil® / cement treatments) seldom affected leaf K (Section 4.3.3.3). This indicates one-way antagonism between K and Ca or K and Mg. Potassium ions are generally taken up more efficiently than Ca or Mg due to the effective H⁺/K⁺ symport, and so K uptake may be less affected by a rise in soil Mg concentration (Mengel and Kirkby, 2001). Similar results were reported for alfalfa (Omar and El Kobbia, 1966), where increasing K led to reduced plant Mg, but increasing Mg had little or no effect on plant K. Miles (2010) reached a similar conclusion for sugarcane. Marschner (1995) reports that Ca²⁺ stimulates net uptake of K⁺ at low pH, by counteracting the negative effects of high H⁺ concentrations on uptake mechanisms.

4.3.1.4. Sum of cations

The sum of cation equivalents within plant tissue often remains nearly constant when the concentration of one cation is increased in the soil (Mengel and Kirkby, 2001), largely due to ionic antagonism. This pattern was shown, to a large extent, in the leaf samples from both sites (Table 4.4). Although the sum of cations (equivalent charge basis) differed between, it remained fairly consistent within, sampling times, across the different K rates (Table 4.4).

Table 4.4: Effect of applied potassium (K) on the mean \pm s.e. (n=16) third-leaf sum of potassium, calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age, on the Oxisol and Inceptisol. Cation sum calculated for the 0 kg Si ha⁻¹ and intermediate N rate (80 kg ha⁻¹ [Oxisol]; 105 kg ha⁻¹ [Inceptisol]) treatments

Kapplied	Sum of K, Ca and Mg (cmol _c kg ⁻¹)				
r applied					
(kg ha⁻¹)	Oxi	sol	Ince	otisol	
	^a PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months	
0	70.59 ± 0.63	64.51 ± 1.24	61.02 ± 1.10	nd [#]	
100	71.42 ± 2.78	63.37 ± 0.49	62.73 ± 0.45	nd	
200	72.39 ± 0.34	65.42 ± 1.32	64.34 ± 0.55	nd	
300	72.81 ± 0.79	67.05 ± 1.38	64.20 ± 0.72	nd	
Significance	NS	NS	NS	nd	
	^b R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months	
0	54.57 ± 0.49	56.91 ± 0.29	57.47 ± 0.89	66.14 ± 0.74 a [∓]	
100	54.77 ± 0.43	57.93 ± 0.35	55.24 ± 2.41	67.98 ± 0.27 b	
200	55.61 ± 0.69	58.64 ± 0.56	60.44 ± 0.45	67.93 ± 0.38 b	
300	55.65 ± 0.79	58.64 ± 0.60	61.15 ± 0.64	68.21 ± 0.60 b	
Significance	NS	NS	NS	P<0.05	
	^c R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months	
0	67.46 ± 1.30	60.38 ± 1.97 a	51.01 ± 0.57 a	51.78 ± 0.65 a	
100	67.21 ± 1.45	62.99 ± 0.99 ab	54.32 ± 0.99 ab	53.08 ± 0.35 ab	
200	68.52 ± 1.49	62.93 ± 1.32 ab	55.38 ± 1.21 ab	53.60 ± 0.40 ab	
300	66.15 ± 0.86	63.93 ± 0.63 b	56.95 ± 1.26 b	55.07 ± 0.80 b	
Significance	NS	P<0.05	P<0.01	P<0.05	

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

[#]nd – not determined.

^{*}Within sampling events, values with different letters are significantly different to each other.

While some significant differences in the sum of cations were found between different K rates, the differences themselves were not large (~5-8%), and similar to the range (~4-6%) reported by Mengel and Kirkby (2001). It is possible that inclusion of ammonium and sodium, which were not measured in the leaf at either trial site, might have made the sum of cations even more similar across the K rates. Increasing K supply, therefore, increased leaf K but did not result in greatly increased cation totals, illustrating the suppression of Ca and Mg uptake.

4.3.2. Effect of applied nitrogen on leaf nutrient content

4.3.2.1. Nitrogen

As expected, N application caused a highly significant increase in leaf N at all sampling events at both sites (Appendix 4.2). On the Oxisol, the highest rate of N did not lead to an increase in leaf N above that of the intermediate rate. The highest N rate did, however, increase leaf N further on the Inceptisol at times, even when leaf N was high (>2.0%) (Appendix 4.2).

4.3.2.2. Phosphorus, potassium, sulphur, zinc, copper and silicon

Leaf concentrations of P, K, S, Zn and Cu all increased with N application at most sampling events at both sites. Leaf sample results from one sampling event each on the Oxisol and Inceptisol illustrate typical effects of applied N on nutrient concentrations (Table 4.5).

Nitrogen plays a critical role in increasing plant utilisation of available P, due to mechanisms such as N-induced increases in root growth, and enhanced ability of the roots to absorb and translocate P (Wilkinson *et al.*, 2000). Poor uptake of K under conditions of severe N deficiency, despite plentiful soil K resources (as reported by Miles, 2010), was evident at both trial sites. In the zero N treatment, leaf K was always lower than at the highest applied N levels, and this effect was significant at five out of six sampling events on the Oxisol and four out of five events on the Inceptisol.

Nitrogen and S uptake must be balanced for efficient protein synthesis and optimal crop quality (Wilkinson *et al.*, 2000). It is evident that the soil organic matter, as well as the gypsum applied at both sites at planting, provided sufficient S to counterbalance the increased N uptake (Table 4.5).

N			A. Oxisol: plar	nt crop, 4 months	S	
applied	Р	К	Si	S	Cu	Zn
(kg ha⁻¹)			(%)		(mg	∣ kg ⁻¹)
0	0.207 ±0.001 a [∓]	1.603 ±0.017 a	0.519 ±0.012 a	0.298 ±0.003 a	5.75 ±0.02 a	15.96 ±0.08 a
80	0.220 ±0.001 b	1.720 ±0.023 b	0.464 ±0.010 b	0.329 ±0.003 b	5.82 ±0.02 b	16.74 ±0.07 b
160	0.219 ±0.001 b	1.761 ±0.023 b	0.463 ±0.012 b	0.325 ±0.002 b	5.85 ±0.02 b	16.87 ±0.08 b
Sig.	P<0.001	P<0.001	P<0.001	P<0.001	P<0.01	P<0.001
N		B. In	ceptisol: second	l ratoon crop, 4	months	
applied	Р	К	Si	S	Cu	Zn
(kg ha⁻¹)			(%)		(mg	ı kg⁻¹)
0	0.175 ±0.001 a	1.256 ±0.091 a	0.459 ±0.014 a	0.129 ±0.003 a	4.63 ±0.08 a	12.10 ±0.13 a
105	0.182 ±0.001 b	1.326 ±0.182 b	0.383 ±0.018 b	0.153 ±0.003 b	5.13 ±0.09 b	12.88 ±0.21 b
210	0.183 ±0.001 b	1.353 ±0.221 b	0.368 ±0.018 b	0.160 ±0.002 c	5.35 ±0.13 b	13.85 ±0.21 c
Sig.	P<0.001	P<0.001	P<0.001	P<0.001	P<0.001	P<0.001

Table 4.5: Effect of applied nitrogen (N) on mean \pm s.e. (n = 32) third-leaf nutrient status on the (A) Oxisol and (B) Inceptisol, each at one sampling event

^{*}Within columns and sites, values with different letters are significantly different to each other.

Micronutrient interactions with N often occur as a result of the acidifying effect of applied N (Wilkinson *et al.*, 2000), leading to increased solubility of metal micronutrients (Meyer, 2013). Although the pH(CaCl₂) was already low in both soils (Oxisol: 4.3; Inceptisol: 4.5), it is possible that localised acidification, through the application of urea, could have rendered Zn and Cu more available in the higher N treatments, resulting in their greater plant uptake.

Leaf Si decreased with N application at all six sampling events on the Oxisol, and at two of the five events on the Inceptisol (as in Table 4.5). These reductions may be linked to a decrease in soil pH associated with increasing N application, rendering phytolith Si less soluble (Haynes, 2014). Additionally, N application, especially to N-deficient plants, leads to more efficient water use by the plants, allowing greater dry matter production for each unit of water and Si absorbed (Jones and Handreck, 1967).

4.3.2.3. Calcium and magnesium

Calcium uptake was seldom affected by N application at either site, decreasing at only two out of six sampling events on the Oxisol, and not at all on the Inceptisol (Table 4.6). On the Inceptisol, Mg uptake was increased by N application at four of the five sampling events (Table 4.6). Nitrogen deficiency can cause marked reductions in uptake of Mg (amongst other nutrients) (Gosnell and Long, 1971), and the significant increases in Mg uptake with N application (Table 4.6) may reflect this. In the zero N treatments, three of the four Inceptisol sampling times with significant Mg effects, had below-threshold N concentrations (Appendix 4.2). This may indicate that N deficiency in the zero N plots had led to reduced Mg uptake. This effect was not consistent, however, on the Oxisol, which showed only two instances of reduction in Mg uptake, of which only one coincided with a below-threshold N concentration in the zero-N treatment (Table 4.6). This soil's inherently higher N status, due to high organic matter, does not explain the fact that at two of the sampling times (plant crop -6 months and first ration -4 months, Appendix 4.2), below-threshold N concentrations in the zero N treatments did not correspond with a significant reduction in Mg uptake.

N applied	Ca (%)					
(kg ha⁻')	Ox	isol	Ince	ptisol		
	^a PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months		
0	0.311 ± 0.001 a [∓]	0.311 ± 0.002	0.311 ± 0.002	nd [#]		
^{\$80/Ψ} 105	0.306 ± 0.001 b	0.314 ± 0.002	0.309 ± 0.002	nd		
[¢] 160/ ^Ψ 210	0.303 ± 0.001 b	0.313 ± 0.002	0.308 ± 0.002	nd		
Significance	P<0.001	NS	NS	nd		
	^b R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months		
0	0.303 ± 0.002	0.309 ± 0.001 a	0.314 ± 0.004	0.330 ± 0.002		
80/105	0.302 ± 0.002	0.306 ± 0.001 b	0.320 ± 0.006	0.329 ± 0.002		
160/210	0.303 ± 0.001	0.307 ± 0.001 ab	0.323 ± 0.004	0.333 ± 0.002		
Significance	NS	P<0.05	NS	NS		
	°R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months		
0	0.333 ± 0.009	0.209 ± 0.005	0.204 ± 0.004	0.162 ± 0.004		
80/105	0.336 ± 0.011	0.203 ± 0.006	0.199 ± 0.006	0.160 ± 0.004		
160/210	0.328 ± 0.008	0.198 ± 0.006	0.192 ± 0.006	0.156 ± 0.004		
Significance	NS	NS	NS	NS		
N applied		Mg	(%)			
(kg ha⁻')	Ox	isol	Inceptisol			
_	PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months		
0	0.145 ± 0.001	0.159 ± 0.002	0.174 ± 0.003	nd		
[₩] 80/ [₩] 105	0.145 ± 0.002	0.165 ± 0.002	0.168 ± 0.002	nd		
^v 160/ ^v 210	0.143 ± 0.002	0.160 ± 0.002	0.168 ± 0.003	nd		
Significance	NS	NS	NS	nd		
-	R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months		
0	0.156 ± 0.003	0.158 ± 0.002 a	0.149 ± 0.003 a	0.192 ± 0.002 a		
80/105	0.151 ± 0.003	0.150 ± 0.002 b	0.158 ± 0.002 b	0.199 ± 0.002 b		
160/210	0.152 ± 0.002	0.148 ± 0.003 b	0.159 ± 0.002 b	0.204 ± 0.002 b		
Significance	NS	P<0.001	P<0.05	P<0.001		
-	R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months		
0	0.243 ± 0.009	0.185 ± 0.005 a	0.107 ± 0.003 a	0.111 ± 0.002 a		
80/105	0.233 ± 0.007	0.177 ± 0.005 a	0.128 ± 0.004 b	0.126 ± 0.003 b		
160/210	0.235 ± 0.007	0.167 ± 0.006 b	0.134 ± 0.004 b	0.128 ± 0.003 b		
Significance	NS	P<0.001	P<0.001	P<0.001		
^a PC = Plant crop: ^b R1 = First Ratoon crop: ^c R2 = Second Ratoon crop						

Table 4.6: Effect of applied nitrogen (N) on mean ± s.e. (n=32) third-leaf calcium (Ca) and magnesium (Mg), at 4- and 6-months' crop age, on the Oxisol and Inceptisol

 $^{\phi}$ = rate applied on the Oxisol; $^{\Psi}$ = rate applied on the Inceptisol

[#]nd = not determined.

[†]Within sampling events, values with different letters are significantly different to each other.

4.3.3. Effect of applied silicates on leaf nutrient content

4.3.3.1. Silicon

In the South African sugar industry, a minimum of 0.75% leaf Si is currently considered satisfactory (Miles and Titshall, 2020); this limit was 0.50% at the time of the trials. In Florida, a yield response to Si is likely if leaf Si falls below 1.00%

(Meyer, 2013). An Australian study suggested that 95% of maximum yield could be obtained when Si in the top visible dewlap leaf (usually the third leaf) was 0.55% (Berthelsen *et al.*, 2003). In the present study, leaf Si was low on both soils at all sampling times, with minima of about 0.20%, and maxima of only 0.57% (Oxisol) and 0.82% (Inceptisol). Hence, although Calmasil® and cement application caused a significant increase in leaf Si at all sampling events at both sites (Table 4.7), concentrations very seldom increased to above the 'satisfactory' FAS threshold (0.75%) on the Inceptisol, and never on the Oxisol.

Si applied (kg ha⁻¹)	Si (%)					
	Oxisol		Inceptisol			
	^a PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months		
0	0.46 ± 0.01 a [∓]	0.19 ± 0.00 a	0.21 ± 0.01 a	nd [#]		
300	0.51 ± 0.01 b	0.20 ± 0.00 b	0.25 ± 0.02 b	nd		
Significance	P<0.001	P<0.05	P<0.05	nd		
	[▶] R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months		
0	0.37 ± 0.01 a	0.48 ± 0.01 a	0.62 ± 0.01 a	0.78 ± 0.01 a		
300	0.38 ± 0.00 b	0.50 ± 0.01 b	0.68 ± 0.01 b	0.82 ± 0.01 b		
Significance	P<0.05	P<0.001	P<0.001	P<0.01		
	°R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months		
0	0.50 ± 0.01 a	0.44 ± 0.01 a	0.34 ± 0.01 a	0.26 ± 0.01 a		
300	0.57 ± 0.01 b	0.52 ± 0.01 b	0.47 ± 0.02 b	0.33 ± 0.02 b		
Significance	P<0.001	P<0.001	P<0.001	P<0.001		

Table 4.7: Effect of applied Calmasil® / cement on mean ± s.e. (n=48) third-leaf silicon (Si) content, at 4- and 6-months' crop age, on the Oxisol and Inceptisol

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

^{*}Within sites and sampling months, values with different letters are significantly different to each other.

[#] nd – not determined.

The naturally low soluble Si of acid soils (Section 2.4.2), such as those used for the present trials, resulted in the highly weathered Oxisol being unable to support adequate levels of plant Si, despite additions of this nutrient. On the relatively young Inceptisol, weathering has apparently not advanced sufficiently to break down the Si-containing minerals in the resistant sandstone parent material. Leaching of the applied Si, and adsorption by aluminium (AI) and Fe oxides, may have exacerbated

Si deficiencies at both sites. Keeping *et al.* (2013) similarly reported that sugarcane leaf Si seldom exceeded 0.50% on a rainfed Inceptisol in KwaZulu-Natal.

4.3.3.2. Nitrogen and phosphorus

Silicon application did not affect N uptake at any sampling event on the Oxisol, and significantly reduced it at only one out of five sampling times on the Inceptisol (results not shown). Although Si application has been reported to increase (Cuong *et al.*, 2017) and decrease (Greger *et al.*, 2018) plant N uptake, its effects in this study were limited.

Phosphorus uptake was improved with Si application at the Oxisol site (3/6 events), and once at the Inceptisol site (results not shown). Obihara and Russell (1972) reported that there is competition between silicate and phosphate for adsorption sites in the soil. Phosphate tends to be displaced by Si from such sites, leading to increased P in the soil solution (Smyth and Sanchez, 1980), facilitating greater uptake of P by the plant. This effect was not consistent but occurred more often on the Oxisol, where a higher concentration of Al and Fe oxides was likely (with a correspondingly greater anion exchange capacity (AEC); Gu and Schulz, 1991). A greater AEC may offer increased scope for replacement of P anions with Si, while the liming effect of Calmasil® and cement would decrease AEC (Gu and Schulz, 1991), possibly leading to release of some P into the soil solution.

4.3.3.3. Calcium, magnesium and potassium

Calcium and Mg uptake was significantly improved by the application of Si at two of the six sampling events on the Oxisol. On the Inceptisol Ca and Mg uptake was increased at three out of five and one out of five events, respectively (Table 4.8). At all sampling events, with or without the Si sources, leaf Ca and Mg were above the 'satisfactory' thresholds of 0.15% (Ca) and 0.08% (Mg) (Miles and Titshall, 2020), at both trial sites.

Si	Ca, Mg and K (%)							
applied (kg ha⁻¹)	Oxisol							
	^a PC - 4 months			PC - 6 months				
	Ca	Mg	ĸ	Ca	Mg	κ		
0	0.306 ± 0.001	0.144 ± 0.001	1.714 ± 0.021 a [∓]	0.314±0.002	0.163±0.002	1.394 ± 0.020		
300	0.307±0.001	0.144 ± 0.001	1.676±0019b	0.311±0001	0.160 ± 0.002	1.425±0.016		
Sig.	NS	NS	P<0.05	NS	NS	NS		
	^b R1 – 4 months			R1 – 6 months				
	Ca	Mg	ĸ	Ca	Mg	ĸ		
0	0.302 ± 0.001	0.152 ± 0.002	1.071 ±0011	0.307±0001	0.152 ± 0.002	1.162±0.014		
300	0.303 ± 0.001	0.154±0.002	1.064 ±0014	0.308±0.001	0.152 ± 0.002	1.147±0.012		
Sig.	NS	NS	NS	NS	NS	NS		
	°R2 – 4 months			R2 – 6 months				
•	Ca	Mg	K	Ca	Mg	K		
0	0.317±0.006 a	0.231±0.005a	1.266±0.023	0.200±0.005a	0.1/3±0.005a	1.477±0.032		
300	0.348±0.008b	0243±0.007b	1.235±0030	0.208±0005b	0.180±0.004b	1.455 ± 0.030		
Sig.	P<0.001	P<0.05	NS	P<0.05	P<0.05	NS		
applied	Inceptisol							
(kg ha⁻¹)	^a PC - 4 months			PC - 6 months				
	Ca	Mg	ĸ	Ca	Mg	κ		
0	0.307±0.001 a	0.167±0.002a	1.317±0022a	nd [#]	nd	nd		
300	0.312±0.002b	0.174±0.002b	1250±0022b	nd	nd	nd		
Sig.	P<0.01	P<0.01	P<0.001	nd	nd	nd		
	[™] R1 – 4 months		R1 – 6 months					
_	Ca	Mg	ĸ	Ca	Mg	K		
0	0.314 ± 0.004	0.158 ± 0.002	1.167±0016	0.329 ± 0.002	0.201 ± 0.002 a	1.359 ± 0.014		
300	0.324 ± 0.004	0.154±0.002	1.146±0018	0.332±0002	0.196±0.002b	1.358±0.011		
Sig.	NS	NS	NS	NS	P<0.05	NS		
	R2 – 4 months			R2 – 6 months				
0	Ca	Mg	K		Mg	K		
0	0.188±0.004 a	0.124 ± 0.004	1.322±0025	0.152±0003a	0.122±0.003	1.3/0±0.018a		
300	0209±0.005b	0.122±0.003	1.301±0026	0.16/±0003b	0.121±0.002	1.339±0.01/b		
Sig.	P<0.001	115	NS	PS0.001	NS	P<0.001		

Table 4.8: Effect of applied silicon (Si) as Calmasil® / cement on mean \pm s.e. (n=48) third-leaf calcium (Ca), magnesium (Mg) and potassium (K) content, at 4- and 6-months' crop age, on the Oxisol and Inceptisol

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop. [†]Within sites, sampling months and columns, values with different letters are significantly

different to each other.

[#] nd – not determined.

Potassium showed a significant decrease at only three of the 11 sampling times, once on the Oxisol (when there was no corresponding increase in either Ca or Mg) and twice on the Inceptisol (Table 4.8). Those on the Inceptisol both coincided with an increase in Ca; Mg was increased at only one of the times. There was no apparent effect of the sum of Ca and Mg (cation equivalent basis) on the K decrease (results not shown). The data indicate that there was very little antagonistic effect of Ca and Mg on K, but that which did occur appeared to be driven by Ca. It is possible,

however, that reduced leaf K may have been caused by a dilution effect following increased growth and yield on the Si treatments (Section 3.3.4.2.1) rather than ionic antagonism.

4.3.4. Leaf potassium, nitrogen and silicon status and crop yield 4.3.4.1. Potassium

Sugarcane yield, sucrose percent (%ERC) and sucrose yield (t ERC ha⁻¹) tended to increase with increasing leaf K, whether this was measured at 4 or 6 months' crop age. This was consistent across trial sites and crops (Figure 4.2A and B), with the exception of the first ratoon crop on the Inceptisol, when increased leaf K associated with the highest K rate resulted in a reduction in sucrose yield (Figure 4.2B). Despite these relatively consistent increases, it would not be possible to use these results to establish a reliable threshold value for K. For instance, on the Oxisol (Figure 4.2A), maximum sucrose yields were obtained at 1.8% K (plant crop), 1.1% K (first ratoon), and 1.4% K (second ratoon). In a later crop, a result of 1.3% would be difficult to interpret as it is not clear whether application of additional K fertiliser would be advisable. It is possible that inadequate moisture status of the crop at some of the sampling times may have contributed to the difficulty in interpreting the leaf sample results.

Similarly, on the Inceptisol (Figure 4.2B), maximum yield was achieved at \sim 1.4-1.5% K in the plant and second ratoon crops. In the first ratoon crop, however, yield was significantly reduced at 1.3% leaf K. The leaf K threshold used by SASRI is 1.05%, in line with that in many countries (Anderson and Bowen, 1990). In the present trials, maximum yield was unlikely to have been obtained in any of the crops if the leaf K levels were as low as 1.05% (Figure 4.2A and B).



Figure 4.2: The effect of leaf potassium (K) concentration on tonnes estimated recoverable crystal (t ERC ha⁻¹), as a percent of the maximum for each harvest, on the (A) Oxisol; and (B) Inceptisol. Leaf samples were collected at 4 months' crop age. Within soil types and crops, each point represents the mean (n=24), across all replicates, at each of the four K application rates (0 kg K ha⁻¹: clear symbol; 100 kg K ha⁻¹: grey symbol; 200 kg K ha⁻¹: hatched symbol; 300 kg K ha⁻¹: black symbol).

4.3.4.2. Nitrogen

Leaf N and sucrose yield showed a negative association in the Oxisol plant crop (Figure 4.3). As with K, leaf analysis did not provide an accurate estimate of the potential efficacy of applying more N. For example, a leaf N concentration of 2.0% would have been 'too high' in the plant crop indicating reduced sucrose yield. In the second ratoon, 2.0% would have been 'too low', and applying more N would have been advantageous. Current FAS leaf N threshold limits range from 1.6-1.9%, depending on month of sampling, area and crop (plant vs ratoon). In Mauritius, this threshold is 1.95% (at 3-5 months) (Halais, 1962), and in Guyana, thresholds range from 1.9-2.5%, at 2-6 months (Evans, 1965). In Australia, the third leaf N threshold ranges from 1.7-1.9%, depending on the sampling month (Calcino et al., 2018). On the Inceptisol, leaf N showed a more predictable pattern of increasing sucrose yield with greater leaf N in the ratoons (data not shown). Again, however, threshold values would be difficult to predict, as yield increases did not level off. Further increases may therefore have occurred beyond the leaf N values recorded (which were as high as 2.3% in the second ratoon crop). Again, moisture status may have impacted these results.



Figure 4.3: Oxisol: The effect of leaf nitrogen (N) concentration on tonnes estimated recoverable crystal (t ERC ha⁻¹), as a percent of the maximum for each harvest. Leaf samples were collected at 4 months' crop age. Within crops, each point represents the mean (n=32), across all replicates, at each of the three N application rates (0 kg N ha⁻¹: clear symbol; 80 kg N ha⁻¹: hatched symbol; 160 kg N ha⁻¹: black symbol).

4.3.4.3. Silicon

Leaf Si levels were not related to %ERC, supporting previous research (Korndörfer and Lepsch, 2001). Increased leaf Si in the ratoon crops coincided with increased stalk (and hence sucrose) yields (Appendix 4.3). As with K and N, leaf analysis could not accurately predict the sufficiency (or otherwise) of measured leaf Si concentrations. The generally low uptake of Si (Section 4.3.3.1) may have contributed to this phenomenon. It is noteworthy, however, that the greatest percentage increases in sucrose yield (t ERC ha⁻¹) were obtained with the greatest percentage increases in leaf Si concentration (Appendix 4.3).

4.4. Conclusions

Increasing rates of applied K suppressed the leaf concentration of Ca, Mg, S, Zn and Cu, with no impact on sugarcane yields. Potassium suppressed Mg uptake more consistently than Ca. Calcium and Mg, on the other hand, seldom suppressed K uptake, and when they did so, Ca appeared to play the dominant role. Cationic antagonism between these three nutrients has been reported in sugarcane, though previous reports have stressed the suppressive effect of Ca and Mg on K. The finding, in this study, that K appears to reduce Ca and Mg uptake far more strongly than *vice versa*, is of considerable significance to the South African sugar industry. Currently, FAS fertiliser K recommendations take soil Ca and Mg into account. For the northern irrigated areas of the industry, when high-clay (>40%) soils have (Ca + Mg) levels greater than 4 000 mg L^{-1} , markedly higher K threshold values are used, as such high Ca and Mg levels are assumed to reduce K uptake (Donaldson et al., 1990). Meyer (2013) also refers to reports of leaf K suppression by increased Ca and/or Mg. No allowance is made in the current FAS recommendations, however, for the reverse situation and soil K levels are not taken into account when determining Ca or Mg thresholds, which remain constant. It is therefore recommended that higher Ca and Mg thresholds be implemented where soil K is high. The details of such a change need to be elucidated with further trials.

There was no consistent effect of K application on leaf N or Si at either trial site. Potassium application generally led to a reduction in leaf S, Zn and Cu, but they did not fall below threshold levels. They should nonetheless be monitored regularly to ensure that deficiency (and hence yield reduction) does not occur.

Nitrogen application increased plant uptake of P, K, S, Zn and Cu. In contrast, Si uptake was often suppressed. Application of Si seldom increased leaf Si to above the 'satisfactory' level of 0.75%. It is possible that higher rates of Calmasil® could have increased leaf Si to a greater extent, though such rates may not increase yield sufficiently to be economically profitable. Over a longer term, however, more than three crops would likely be produced following a single application of Calmasil®, perhaps leading to improved economic feasibility. Although the required additional cane yield might not be achieved, even over future rateons, continued soil
acidification and depletion of Si reserves would ultimately necessitate applications of these liming sources to allow the continued production of sugarcane.

Relationships between leaf nutrient concentrations and crop yields were poorly defined. Although harvest characteristics showed patterns of increase or decrease with increasing leaf K, N and Si, the leaf data did not allow for the reliable determination of nutrient threshold levels, above which it would not be economic to apply more of a specific nutrient. The data therefore indicated that leaf analysis was not a reliable means of establishing nutrient sufficiency in the crops from the present trials. It is suggested that variable crop moisture status at the time of leaf sampling may have been a major contributor to these poor relationships. The FAS gives instructions to growers of when to collect leaf samples and the analytical reports state that results are only meaningful if the samples were collected under optimal conditions of age, season and rainfall distribution. However, crop moisture status is likely to be variable in the entire rainfed portion of the South African sugarcane industry where seasonal rainfall distribution is often unreliable. Thus, it is likely that a portion of the samples submitted to the laboratory is affected by conditions of limited moisture, as was probably the case from the present trials.

Chapter 5

The effect of potassium, nitrogen and silicon application on relationships between soil nutrient dynamics and yield of sugarcane

5.1. Introduction

Exchangeable or readily-available soil nutrients are supplemented by more slowlyavailable nutrient reserves. Release of these reserves to the available pool may depend, directly or indirectly, on the amount of the nutrient supplied as fertiliser.

In sugarcane, top- and subsoil samples are routinely analysed to determine soil nutrient status (Schroeder et al., 1993). Although readily-available potassium (K) is routinely measured and used for K fertiliser recommendations by many soil laboratories, a number of authors (e.g. Wood and Burrows, 1980; Wood and Schroeder, 1991; Miles and Farina, 2014) have identified the slowly-available K as an important determinant of actual plant K availability. This may represent a considerable source of K, which can become available to the crop over a season or more. Boiling nitric acid has been used to extract this slowly-available soil K (nitric K) (Haysom, 1971). Haysom (1971) concluded that, in the Australian sugarcane industry, soils with nitric K reserves greater than 2.5 cmol_c kg⁻¹ do not require K fertiliser to maintain high productivity, and Australian recommendations have taken both readily-available and nitric K into account (Moody et al., 2007; Schroeder, 2007). Australia's sugarcane soils currently receive a reduced K fertiliser recommendation when nitric K is greater than 0.7 meg 100 g⁻¹ (Calcino *et al.*, 2018). Miles and Farina (2014) postulated that non-exchangeable K reserves contribute significantly to crop K requirements in many southern African soils.

The availability of applied K is determined by, amongst others, the extent to which it is immobilised by soil minerals. Illite and vermiculite are strongly K-fixing clay minerals that result in reduced availability (Goli-Kalanpa *et al.*, 2008), even following K application (Wood and Meyer, 1986). Higher applications of K fertiliser are thus required on soils that contain these minerals in order to supply the crop with sufficient available K (Kingston, 2000). Over time the K from these soil minerals may

become available to plants. On the other hand, soils dominated by, for example, hydroxy-aluminium and iron interlayer groups, exhibit low fixation of K, thereby allowing a larger proportion of added K to remain in solution (Rich and Obenshaim, 1955). The presence of only small amounts of 2:1 layer silicate clays influences the supply of K to plants, even in highly weathered soils (Arkcoll *et al.*, 1985).

Applied nitrogen (N) and silicon (Si) are affected by different soil, management and environmental conditions. Applied N is typically ephemeral in soil, so its dynamics are difficult to track, and its uptake by plants, and hence effect on crop growth, are challenging to predict. Thus, no attempt was made, in this study, to use soil N data to predict sugarcane yield response to applied N.

Silicon application is recommended in the rainfed areas of the South African sugar industry, as available soil Si is typically below the threshold (15 mg L⁻¹) required by sugarcane (Miles, 2014). However, researchers have reported limited or inconsistent plant uptake of the Si applied in these areas, despite significant increases in soil Si concentrations (Keeping and Meyer, 2002; Berry *et al.*, 2011). The possible reasons for such poor Si uptake have been discussed in Section 2.4.2.

In addition to the effect of nutrient applications on the concentration of soil nutrients, the impact of nutrient levels on crop yield and quality needs to be investigated in order for robust fertiliser recommendations to be made. Fertiliser application rates which only raise soil nutrients to below established thresholds will lead to yield loss, while excessive recommendations will be uneconomic and may lead to environmental problems. Further complexity is introduced when subsoil nutrients are also considered.

The aims of this investigation on two different soils of the South African sugarbelt were to determine the effects of (a) applied K, N and Si on soil nutrients and other soil characteristics; (b) varying soil nutrient levels on crop yield and quality; (c) subsoil nutrient levels on crop yield; and (d) the extent to which nitric K contributed to readily-available K.

5.2. Materials and methods

5.2.1. Trial details and pre-trial soil sampling

These are given in Sections 3.2.1 to 3.2.4.

5.2.2. Topsoil and depth sampling

Topsoil samples (0-20 cm) were collected using an auger from each of the four replicates approximately 6 to 8 weeks after application of N fertiliser to each crop. Fifteen samples were collected from each nett plot (central five rows at each trial site), in the middle of the interrows. These samples were bulked per plot and thoroughly mixed prior to being analysed.

Soil samples were also collected from three of the four replicates in 20 cm increments from 0-100 cm (Oxisol) and 0-80 cm (Inceptisol) between 2 and 7 weeks after each harvest. The 0-20 cm samples were collected as before and for the subsoil samples, four cores were collected diagonally across each nett plot from the middle of the interrows. The upper 20 cm of soil was discarded, as this had already been collected, and subsequent 20 cm depth increments were combined across the four core sites within each plot and thoroughly mixed prior to analysis.

5.2.3. Chemical analysis

All soil samples were dried at 40°C, crushed and then sieved to pass a 1 mm mesh. Exchangeable K, calcium (Ca), magnesium (Mg), copper (Cu), iron (Fe), manganese (Mn) and zinc (Zn) were extracted using 'Ambic 2' multinutrient extractant (Van der Merwe *et al.*, 1984) and phosphorus (P), Si, pH and exchangeable acidity were determined using the methods given in Appendix 3.1. Ammonium and nitrate were extracted with 2 *M* KCI and measured colorimetrically by segmented flow autoanalyser (Seal Analytical Flow Injection Analyser).

5.2.4. Mineralogy and total potassium

At each trial site one soil core was collected with an auger in 20 cm increments to 100 cm (Oxisol) and 80 cm (Inceptisol) from an unfertilised pathway between trial plots. Part of each sample was separated into clay and the (sand + silt) fractions by decanting, using the application of Stokes' Law.

The mineralogical composition of the clay fraction was characterised by X-ray diffraction (XRD) on oriented specimens on glass slides. Prior to preparation of the specimens, half of the separated clay fraction was saturated with Mg^{2+} and the other half with K⁺. The Mg-saturated specimens were air-dried, treated with ethylene glycol at 60°C overnight, and with glycerol at 85°C overnight, and the K-saturated specimens were air-dried and heated at 550°C for four hours (Bühmann *et al.*, 1985). All XRD analyses were carried out on a Philips PW1050 X-ray diffractometer using monochromated CoK α radiation at 40 kV and 40 mA. Oriented specimens were scanned at 1° min⁻¹ with a step-scan of 0.02° between 3 and 45° 2 Θ . All data were captured by a Sietronics 122D automated micro-processor attached to the diffractometer.

The (sand + silt) samples were dried (110°C for 24 hours), milled and then random powder preparations were examined by XRD using monochromated CoK α radiation from 3 to 90° 2 Θ on a PANalytical Empyrean X-ray diffractometer.

The remaining whole soil samples were dried (110°C for 24 hours), milled, formed into a fused bead, and analysed for total K on a PANalytical Axios Max X-ray fluorescence (XRF) spectrometer.

Rock fragments were collected using an auger from the Oxisol at depths between 240 and 300 cm, while decomposing rock was collected at 80 to 100 cm from the Inceptisol. One rock sample from each trial site was dried (110°C for 24 hours), crushed, milled, formed into a fused bead, and analysed for total K by XRF.

5.2.5. Slowly-available potassium

Subsoil samples from each depth collected from the zero K, zero N and zero Si treatment plots following the first ration crop harvest at each site were extracted with boiling 1 M HNO₃ for 30 min (Haysom, 1971). The K was measured using inductively coupled plasma optical emission spectrometry (ICP-OES, Varian 720ES), and the mean nitric K calculated for each soil and soil depth. Reported nitric K values were corrected for readily-available ('Ambic 2') K.

5.2.6. Statistical analysis

Mean results of each determination were subjected to analysis of variance (ANOVA) (Genstat – 14th Edition; VSN International, 2011), where the data sets were balanced (i.e., the topsoil samples). For the unbalanced depth sample data (where four replicates were sampled at 0-20 cm and only three at greater depths), residual maximum likelihood (Genstat – 14th Edition; VSN International, 2011) was used.

Where data were not normally distributed, log_{10} transformations were performed to normalise the data. Where treatments showed significant effects, values were analysed using the Holm Sidak multiple comparison separation or Fisher's protected least significant difference test to identify where the differences between treatment means lay. Differences were considered to be significant at P < 0.05.

5.2.7. Potassium balance

To examine the fate of K in the cropping system, a K balance was calculated for each harvest (Equation 5.1).

K balance (kg ha⁻¹) = (K in harvested crop + Remaining soil K) – (Applied K + Initial soil K) Equation 5.1

where:

K in harvested crop is the amount of K present in each above-ground plant part, determined within two weeks prior to harvest. Treatments including all rates of K and N with applied Si were sampled for this purpose. At each sampling event, 1 m of one guard row per plot was cut at ground level and partitioned into the stem, dead leaves, green leaf blades and the combined meristem and green leaf sheaths. Each

component was weighed fresh, dried (80°C for 48 hours), reweighed and analysed for K by XRF (leaf material) or ashing and ICP-OES (other plant components).

Remaining soil K is the readily-available ('Ambic 2') soil K in the top 100 cm (Oxisol) or 80 cm (Inceptisol), sampled two weeks after crop harvest. This is assumed to reflect the available K which remained in the soil after the crop was removed.

Applied K is the amount of fertiliser K (kg ha⁻¹) applied at the start of each crop.

Initial soil K is the readily-available ('Ambic 2') soil K in the top 100 cm (Oxisol) or 80 cm (Inceptisol), sampled two weeks after harvest of the *previous* crop. This is the amount of K available in the soil at the start of the crop's growth.

Potassium in root tissue, slowly-available K and leaching were not taken into account.

5.3. Results and discussion

5.3.1. Initial soil properties

These are presented in Section 3.3.1 (Table 3.2).

5.3.2. Potassium

5.3.2.1. Mineralogy and total potassium content

The parent material at both sites was sandstone, though there were differences in mineralogy (Table 5.1). The Oxisol sandstone was found at depth (240-300 cm) and was highly weathered. It was chiefly composed of quartz and kaolinite with subsidiary K-feldspar, goethite and hematite. Kaolinite was the dominant mineral in the clay fraction of the Oxisol which also contained some hydroxyl AI interlayered vermiculite (HIV) and illite, along with traces of hematite, quartz and K-feldspar (Table 5.1). In the (sand + silt) fraction, quartz was dominant, with trace amounts of K-feldspar and anatase.

In contrast, the Inceptisol sandstone was more coherent and found at 80-100 cm depth. It was, however, weathered to the extent that some root penetration was possible. The rock was a mixture of quartz, kaolinite, K-feldspar and mica and/or illite, with trace amounts of goethite, hematite and anatase (Table 5.1). Again,

kaolinite was the dominant mineral in the clay fraction. Illite and quartz were present as subsidiary components, along with trace amounts of vermiculite, goethite and Kfeldspar (Table 5.1). In the (sand + silt) fraction, quartz was dominant, with Kfeldspar present, as well as trace amounts of anatase. The Inceptisol, being less weathered, contained more K-feldspar than the Oxisol. The K-feldspars present in both soils could be responsible for some K release (Sadusky *et al.*, 1987; Sanz Scovino and Rowell, 1988; Parker *et al.*, 1989a, b; Pal *et al.*, 2001).

of the Oxisol and Inceptisol topsoils										
Fraction	Depth	Kao	Goe	Hem	Qtz	ΗIV	III / Mca	Vrm	K-Fel	Ant
	(

Table 5.1: Mineralogy of the sandstone parent rock, and the clay and (sand + silt) fractions

							INICa			
	(cm)									
				Oxisol						
Clay	0-20	+++	-	tr	tr	+++	+	-	tr	-
(Sand + silt)	0-20	-	-	-	++++	-	-	-	tr	tr
Rock	240-300	++	+	+	++++	-	-	-	+	tr
				Inceptisc	ol 🛛					
Clay	0-20	++++	tr	-	+	-	+	tr	tr	-
(Sand + silt)	0-20	-	-	-	++++	-	-	-	+	tr
Rock	80-100	+	tr	tr	++++	-	++	-	+	tr

++++ = > 60%; +++ = 40–60%; ++ = 20–40%; + = 5–20%; tr = <5%; - = absent; Kao = kaolinite; Goe = goethite; Hem = hematite; Qtz = quartz; HIV = hydroxyl Al interlayered vermiculite; III = illite; Mca = mica; Vrm = vermiculite; K-Fel = K-feldspar; Ant = anatase

The differences in the parent material at the two sites, both in the degree of weathering and the mica present in the Inceptisol, are reflected in the Inceptisol containing an order of magnitude greater total K than the Oxisol at all depths (Table 5.2). The Inceptisol thus has a much larger pool of potentially available K than the Oxisol.

Soil	Sampla	Depth	К
5011	Sample	(cm)	(%)
		0 – 20	0.096
		20 – 40	0.080
Oxisol	Whole soil	40 - 60	0.076
		60 - 80	0.078
		80 – 100	0.092
	Rock	240 - 300	0.252
		0 – 20	0.564
	Whole soil	20 – 40	0.592
Inceptisol	VVIIOle SUI	40 - 60	0.607
		60 - 80	0.847
	Rock	80 - 100	2.704

Table 5.2: Total potassium (K) in the whole soil and rock from the Oxisol and Inceptisol

5.3.2.2. Effect of potassium application on topsoil (0-20 cm) readily-available potassium

The application of K led to an increase in topsoil readily-available K in both soils (Figure 5.1). At both sites, there was a greater rate of increase in the ration crops (where the K was surface-applied) than in the plant crop (when the K was incorporated into the soil).

An increase in topsoil K in the zero plots, between the pre-trial sampling and the plant crop's sampling, was evident at both sites. On the Oxisol, pre-trial analysis returned a value of 55 mg K L⁻¹ (Table 3.2), while the zero plots had 125 mg K L⁻¹ three months after trial planting (Figure 5.1A). On the Inceptisol, K increased from 44 mg L⁻¹ (Table 3.2) to 96 mg L⁻¹ (Figure 5.1B) over four months. It is possible that K release from slowly-available pools accounted for this additional K (McLean and Watson, 1985).



Figure 5.1: The effect of potassium (K) application to each crop on topsoil (0-20 cm) readilyavailable K in the (A) Oxisol and (B) Inceptisol. Each point represents the mean of 24 plots per K treatment.

Also noteworthy is the extent to which topsoil readily-available K increased in each soil, in response to fertiliser applications. At the plant crop sampling event, the zero plots' topsoil K at the two sites were within 30% of each other. In the plant crop

300 kg K ha⁻¹ treatment, however, (where each soil had received the same amount of K), the Oxisol's K concentration (226 mg L⁻¹) was 65% greater than that in the Inceptisol (137 mg L⁻¹) and by the second ratoon, it was 82% greater. There was thus a greater build-up of readily-available K over the duration of the three crops on the Oxisol, where second ratoon soil K concentrations were 40% greater than after the plant crop (at 300 kg K ha⁻¹), compared with 26% greater in the Inceptisol.

The amount of organic matter in the soils (Table 3.2) may have affected the behaviour of applied K. High soil organic matter content may favour the initial fast adsorption of K^+ ions (Wang and Huang, 2001). However, monovalent ions tend to be more loosely held by organic matter than divalent ions (Wang and Huang, 2001), and, depending on the constitution of the organic matter, may be more easily released or exchanged. The higher organic matter in the Oxisol may thus have adsorbed a portion of the applied K, but retained it in a more readily-available form.

The clay mineralogy of the two soils could also have influenced the degree to which applied K became part of the non-exchangeable K pool. Of the major clay minerals responsible for K fixation, vermiculite was present in the Inceptisol in trace amounts, but absent in the Oxisol (Table 5.1) and this may have increased K fixation in the Inceptisol to a limited extent. Hydroxyl-aluminium interlayered vermiculite was an important constituent of the Oxisol's clay mineral fraction, but was absent in the Inceptisol (Table 5.1). This could have been a contributing factor in the reduced fixation of K, and hence build-up of readily-available K, in the Oxisol in comparison with the Inceptisol.

The inverse of the regression slopes in Figure 5.1 represents the K requirement factor (KRF) of each soil, defined as the amount of K (kg ha⁻¹) required to raise the soil K test by 1 mg L⁻¹ (Johnston *et al.*, 1999). The KRFs of the Oxisol were 2.84, 1.33 and 1.24 kg K ha⁻¹ mg⁻¹ L⁻¹ soil in the plant, first and second ratoon crops, respectively. The corresponding KRF values of the Inceptisol were 6.77, 2.41 and 2.28. These KRF values fall within the range (1.50 to 8.80 kg K ha⁻¹ mg⁻¹ L⁻¹ soil) reported for KwaZulu-Natal soils by Johnston *et al.* (1999). Two Oxisols (Inanda soil forms) reported by Johnston *et al.* (1999) had KRF values of 1.68 and 1.64, within the same range as the Oxisol in the present study; a number of Inceptisols ranged from 2.27 to 3.40. No similar Inceptisols in the Johnston *et al.* (1999) study, however,

exhibited KRF values as high as that at the plant crop in the present study. Thus, almost twice the amount of K fertiliser was required to raise the Inceptisol K concentration than that of the Oxisol in the present study.

5.3.2.3. Effect of potassium application on mean soil profile characteristics following harvests

Following each crop of sugarcane, treatments with applied K showed an increase in average soil profile K concentrations, on both soils. Both trial sites responded similarly and showed the same trend as given for the topsoil samples alone (Figure 5.1). Potassium application also, at times, significantly decreased soil pH and significantly increased Zn, exchangeable acidity and acid saturation (results not shown), though these effects did not occur consistently across trial sites or sampling times.

There was, however, a significant decrease in Si, with K application, after two of the three crop cycles on the Inceptisol (Figure 5.2). This was likely due to the higher yield from the plots which had received K, leading to increased Si uptake by the crop. Ashraf *et al.* (2010) reported that Si uptake by sugarcane was significantly increased by the addition of K (in a hydroponic medium), and Tahir *et al.* (2010) found that Si application to wheat seedlings significantly increased K concentration in the shoots. This mutually positive relationship between K and Si was not, however, reflected in the Inceptisol leaf analyses (Chapter 4).



Figure 5.2: Inceptisol: Effect of potassium (K) application on soil profile (0-80 cm) available silicon (Si) after each of the three crops (n=31). Within each crop, points with different letters are significantly different to each other (P < 0.05).

5.3.2.4. Potassium movement down the profile

There was some evidence of K movement down the profile at both trial sites. Movement was most marked in the Oxisol (Figure 5.3A, B, C) as shown after the harvest of the second ratoon crop, when there were significant differences between the readily-available soil K at the different K application rates at the 40-60 cm depth interval (Figure 5.3C).



Figure 5.3: Oxisol: Mean readily-available potassium (K) per soil depth interval following (A) plant crop, (B) first ration and (C) second ration harvests. n=24 (0-20 cm) and n=18 (20-100 cm). Within each crop, points with different letters are significantly different to each other (P < 0.001). Where no significant differences occurred between points within a depth interval, the points are circled.

The Inceptisol, however, did not show significant differences between K rates at this depth (Figure 5.4A, B, C), indicating either that K application had not affected the soil to this depth, or that the applied K may have become non-available at depth. Although this might indicate slightly greater leaching of applied K in the Oxisol, K movement may also have been facilitated by the higher rate of gypsum applied to the Oxisol (5 t ha⁻¹) compared to the Inceptisol (3 t ha⁻¹). Gypsum has been reported to cause some leaching of K through the soil profile (Ernani *et al.*, 2006).



Figure 5.4: Inceptisol: Mean readily-available potassium (K) per soil depth interval following (A) plant crop, (B) first ration and (C) second ration harvests. n=24 (0-20 cm) and n=18 (20-80 cm). Within each crop, points with different letters are significantly different to each other (P < 0.001). Where no significant differences occurred between points within a depth interval, the points are circled.

5.3.2.5. Slowly-available (nitric) potassium

The nitric K in both soils, at all but two of the sampled soil depths, was $< 1 \text{ cmol}_c \text{ kg}^{-1}$ (Table 5.3). This corresponds with mean values reported for the KwaZulu-Natal north coast by Miles and Farina (2014) and indicates low slowly-available K reserves (Miles, 2016). The Inceptisol contained between three and five times the amount present in the Oxisol at each depth interval (Table 5.3). This finding and the trends with depth are consistent with the amount of total K found in the soil and parent material at each site (Table 5.2). In the Oxisol, between 7.2 and 9.0% of the total K was made up by nitric K, depending on the sample depth (Table 5.3), while these figures were 4.7-6.7% in the Inceptisol. Corresponding figures in the literature range from 0.7-2.2% on a loamy sand (Sadusky *et al.*, 1987) to 3.6%

on a laterite soil (Lalitha and Dhakshinamoorthy, 2014), illustrating a greater percentage of slowly-available K reserves in the soils of the present study.

Table 5.3: Mean \pm s.e. (n=5, Oxisol; n=4, Inceptisol) slowly-available (nitric) potassium (K) per soil depth interval at each trial site, following the first ration crop harvest, and as a percentage of total K

Soil	Depth	Nitric K	Nitric K	% of Total K made
3011	(cm)	(cmol _c kg⁻¹)	(kg K ha⁻¹)	up by nitric K
	0-20	0.22 ± 0.02	189 ± 20	9.03 ± 0.86
	20-40	0.17 ± 0.02	148 ± 14	8.33 ± 0.76
Oxisol	40-60	0.14 ± 0.01	125 ± 7	7.21 ± 0.39
	60-80	0.16 ± 0.01	143 ± 5	7.97 ± 0.27
	80-100	0.19 ± 0.01	169 ± 8	7.92 ± 0.37
	0-20	0.70 ± 0.13	661 ± 118	4.65 ± 0.87
Inceptisol	20-40	0.97 ± 0.15	976 ± 130	6.54 ± 0.99
	40-60	1.06 ± 0.12	1009 ± 107	6.70 ± 0.75
	60-80	1.29 ± 0.12	1238 ± 103	6.30 ± 0.57

5.3.2.6. Potassium balance

On the Oxisol, between 33 and 112 kg ha⁻¹ more K was available at the end of the three crop cycles than at the beginning (Table 5.4). This represents a recovery of 104-129% of the original K. On the Inceptisol, K recovery was even greater, between 110 and 194% of the original K, or 63-202 kg K ha⁻¹ more. This recovery was despite the fact that neither K in the crop roots, nor K leaching below the root zone, were taken into account, both of which are likely to have resulted in even greater K recovery. It is, however, not uncommon for more K to be 'recovered' than was present at the start of cropping (e.g. Madaras and Lipavský, 2009).

In each soil there was a greater over-recovery of readily-available K in the lower-K treatments than in those with higher K rates (Table 5.4). Where less fertiliser K was applied, therefore, a larger proportion of K became available from other sources. It is possible that in the low K-input treatments, the sugarcane roots extracted additional K from deeper in the soil profile, while the higher K treated plots had sufficient in the upper portion of the profile.

Table 5.4: Potassium (K) balance for the Oxisol and the Inceptisol soil profiles. Values are mean \pm s.e. (kg K ha⁻¹) per crop at each site (n=24). 'K removed by crop' refers to K present in the total above-ground biomass, per crop. Calculated K recovery compares the mean amount of readily-available K remaining at the end of each crop with that at the start

Start of Crop		End of Crop		K Balance	
K applied	K originally present in soil profile	K remaining in soil profile	K removed by crop	Difference end-start	Calculated K recovery
		(kg K ha⁻¹) -			(%)
Oxisol pro	ofile (0-100 cm)				
0	480 ± 23	361 ± 21	230 ± 39	112 ± 44	129 ± 11
100	557 ± 27	465 ± 31	245 ± 22	53 ± 42	110 ± 6
200	645 ± 20	607 ± 23	271 ± 38	33 ± 41	104 ± 5
300	712 ± 26	791 ± 29	272 ± 24	51 ± 44	106 ± 5
Inceptisol	profile (0-80 cm)				
0	331 ± 19	336 ± 10	197 ± 19	202 ± 23	194 ± 26
100	348 ± 18	389 ± 12	213 ± 20	153 ± 24	141 ± 9
200	397 ± 19	469 ± 23	265 ± 25	137 ± 29	125 ± 6
300	418 ± 21	541 ± 24	240 ± 20	63 ± 28	110 ± 5

Another possible explanation for the over-recovery of K may be that greater amounts of slowly-available K became available during the growing season, possibly due in part to efficient root extraction, as well as in response to lower levels of readily-available K. Poss *et al.* (1997) similarly concluded that if K exports exceeded inputs, some K must have been released from the slowly-available soil K pool. Based on the nitric K values (Table 5.3), K could become slowly available to the crop over a period of years. It is feasible that with the low readily-available K in the zero K plots, diffusion dynamics favoured faster release of slowly-available K than in the high K treated plots. The results indicate that despite these soils having 'low' K reserves (Section 5.3.2.5; Haysom, 1971), they may be able to release more K than such a classification suggests. The greater K recovery on the Inceptisol is supported by the greater reservoir of nitric K in this soil than in the Oxisol (Table 5.3).

5.3.2.7. Effect of topsoil potassium on sugarcane yield and quality

There were no significant differences in stalk yield between the different rates of applied K at any harvest on the Oxisol (Section 3.3.2.1.1). Yields did, however, increase slightly with increasing topsoil readily-available K in each harvested crop (Figure 5.5A). Although not statistically significant, the increase in yield between the lowest and highest readily-available soil K values was almost 8 t cane ha⁻¹ or 6% in the first rateon crop, a yield increase significant in practical terms.

Sucrose content (% estimated recoverable crystal, %ERC) of the cane on the Oxisol was significantly increased by increasing K rates only in the second ratoon harvest, and then not by a large amount (Section 3.3.2.1.2). The total sucrose tonnage (t ERC ha⁻¹) was, however, notably increased by increased soil K (Figure 5.5B), with increases in the second ratoon crop being significant.

The readily-available topsoil K threshold currently recommended for this soil appears to be accurate in terms of both stalk and sucrose yield. Figure 5.5 indicates that above approximately 150 mg K L⁻¹, both stalk and sucrose yield response curves begin to flatten off. Above the South African Sugarcane Research Institute's Fertiliser Advisory Service (FAS) threshold of 166 mg K L⁻¹ (Miles, 2014), yield responses to further readily-available K were limited, and there would therefore be minimal additional yield benefit of applying more K fertiliser.



Figure 5.5: Oxisol: Mean (n=24) sugarcane (A) stalk and (B) sucrose yield, as a percentage of the maximum yield per harvest in relation to readily-available topsoil potassium (K). For the second ratoon crop in B, points with different letters are significantly different to each other (P < 0.05). Each point is shaded to represent one of the four K application rates (0 kg K ha⁻¹: clear symbol; 100 kg K ha⁻¹: grey symbol; 200 kg K ha⁻¹: hatched symbol; 300 kg K ha⁻¹: black symbol).

Only the first ration crop on the Inceptisol showed significant yield differences between different K application rates (Section 3.3.2.2.1). However, a general pattern of yield increase was evident with increasing topsoil readily-available K, with, for example, a difference of 7 t cane ha⁻¹ (7%) between the lowest and highest amounts

of soil K in the plant crop (Table 5.5). No significant differences were evident in the %ERC between different K rates on the Inceptisol (Table 5.5). The t ERC ha⁻¹ was, however, significantly increased by increased topsoil K in both the plant and first rateon crops (Table 5.5).

Table 5.5: Inceptisol: Mean (n=24) sugarcane stalk yield, sucrose percent and sucrose yield, as a percentage of the maximum yield per harvest in relation to readily-available topsoil potassium (K)

K applied	Readily- available K	Stalk yield (t cane ha ⁻¹)	Sucrose percent (%ERC)	Sucrose yield (t ERC ha ⁻¹)
(kg ha⁻¹)	(mg L ⁻¹)	(as %	of the maximum yield pe	er harvest)
			Plant crop	
0	96	93.3	98.6	92.0 a [∓]
100	104	94.4	98.7	93.2 a
200	129	94.4	99.9	94.2 a
300	137	100.0	100.0	100.0 b
	Significance	NS	NS	P<0.05
			First ratoon crop	
0	57	91.2 a	99.3	91.3 a
100	94	94.1 ab	100.0	95.0 ab
200	135	100.0 b	98.8	100.0 b
300	182	93.7 ab	98.5	93.0 ab
	Significance	P<0.05	NS	P<0.05
			Second ratoon crop	
0	45	94.7	99.5	93.9
100	80	98.0	97.1	94.9
200	130	100.0	99.4	98.9
300	174	99.9	100.0	100.0
	Significance	NS	NS	NS

⁺ Within each harvest and yield component, values with different letters are significantly different to each other.

The topsoil readily-available K threshold recommended for the Inceptisol (average stalk yields obtained ~90 t cane ha⁻¹ over the three crops), was 135 mg K L⁻¹. Statistically, with no significant differences between the stalk yields associated with the three highest topsoil K values in the first ratoon crop, and no differences in either of the other two crops, this threshold could have been reduced to 94 mg L⁻¹, the amount associated with the 100 kg K ha⁻¹ application rate (Table 5.5, stalk yield, first ratoon crop). However, from the total sucrose tonnage (Table 5.5), sucrose yield would have been significantly reduced below 137 mg L⁻¹ topsoil K in the plant crop, and by approximately 1 t ERC ha⁻¹ (though not significant) in the first and second

ratoon crops. Given that sugarcane payments are based on sucrose yield, the current recommended local threshold appears appropriate for this soil.

Mean sugarcane stalk yields over the three harvests were greater on the Inceptisol than on the Oxisol (Table 5.6). In contrast, the Inceptisol had lower readily-available topsoil K over the duration of the trial, and lower stalk K concentrations. This may suggest luxury uptake of K on the Oxisol, which had more topsoil (0-20 cm) readily-available K, and lower cane yields, but greater stalk K. Luxury uptake is less evident on the Inceptisol, where stalk K percent values did not increase substantially upon addition of fertiliser K (Table 5.6). Wood and Schroeder (2004) reported luxury uptake of K in sugarcane, though a variety x K rate interaction was evident. It is also possible that the lower stalk K on the Inceptisol (in comparison with the Oxisol) was due to dilution following greater crop growth.

Table 5.6: Mean readily-available topsoil (0-20 cm) potassium (K) (n=27), mean yield (n=27) in harvested stalks and mean K percent in stalks (n=27) per season, and total K removed (n=36), as a sum over the three crops at each trial site

K applied	Mean availa	i readily- ble soil K	Mean stalk yield		Mean percent K in stalks		Total K removed in the harvested stalks	
$(ka ha^{-1})$	(m	ng L⁻¹)	(t cane ha ⁻¹) (%) (kg		(%)		g ha⁻¹)	
(kg na)	Oxisol	Inceptisol	Oxisol	Inceptisol	Oxisol	Inceptisol	Oxisol	Inceptisol
0	107	66	81.9	90.7	0.32	0.26	270	298
100	153	93	84.5	93.0	0.34	0.36	285	317
200	234	132	85.5	95.4	0.44	0.35	378	446
300	292	164	86.4	95.5	0.46	0.33	369	332

Also notable is the large amount of K removed in the harvested stalks at both sites (Table 5.6). Even with a low concentration (66 mg L^{-1}) of readily-available K, stalks on the Inceptisol removed almost 300 kg K ha⁻¹ in the zero K treatment. Potassium removal on the Oxisol was also high, albeit with a larger pool of readily-available soil K. This indicates that, in the absence of high available K, less available sources of soil K supplied sufficient K to support high uptake, particularly on the Inceptisol (Section 5.3.2.6).

5.3.2.8. Inclusion of subsoil potassium data into topsoil calibration

In order to ascertain the extent to which subsoil (>20 cm) K affected sugarcane yields, subsoil K concentrations were included with the topsoil regressions against harvest yields (t cane ha⁻¹). Although samples were collected to 80 cm (Inceptisol) and 100 cm (Oxisol), sugarcane growers do not have standardised procedures for depth sampling, and hence sample to varying depths. If subsoil data are to be included in routine advisory calibrations, it may be most useful to include only two subsoil depths (20-40 cm and 40-60 cm), since in practice samples are likely to be taken only to these depths. Mean soil K from 0-60 cm was thus plotted against sugarcane stalk yield (Figure 5.6).

When only topsoil (0-20 cm) K concentrations were plotted against stalk yield (Figure 5.6A), an R^2 value of 0.42 was obtained when the soils were combined. When the Oxisol and Inceptisol values were separated, R² values of 0.61 and 0.27 were obtained, respectively (Figure 5.6A). Including the subsoil values improved the relationships ($R^2 = 0.59$, 0.78 and 0.48 for the combined, Oxisol and Inceptisol soils, respectively) (Figure 5.6B). The threshold used by a laboratory to make fertiliser recommendations would depend upon the targeted yield. While the topsoil threshold, based on the maximum value of the trendlines in Figure 5.6A, would range from 145-250 mg K L^{-1} (depending on the threshold curve chosen), the profile (0-60 cm) threshold K for these soils (Figure 5.6B), would be approximately 78-95 mg L^{-1} . If 95% of maximum yield is considered satisfactory, then these thresholds could be considerably reduced to approximately 100 mg K L⁻¹ (for topsoils, Figure 5.6A), and 45 mg K L⁻¹ (for soil profiles, Figure 5.7). Regardless of which soil's curve and yield threshold are used, the results indicate that inclusion of subsoil K values into the recommendation package would improve the accuracy of the soil K threshold curve, ultimately leading to more precise fertiliser recommendations. However, the responses to K in these trials were small; even the lowest yields were, in all cases, > 90% of the maximum (Figure 5.6A and B). More precise response curves would need to be determined on soils where the K response is greater.

The better relationship between readily-available soil K and sugarcane stalk yield obtained for the Oxisol ($R^2 = 0.61$) than the Inceptisol ($R^2 = 0.27$) may have been due to the greater amounts of nitric K present in the Inceptisol (Table 5.3). Release



of K from this pool may have rendered the relationship between yield and readily-available K less consistent.

Figure 5.6: Mean sugarcane stalk yield, as a percentage of the maximum yields obtained at each harvest, in relation to (A) topsoil (0-20 cm) and (B) top- and subsoil (0-60 cm) readily-available potassium (K) in the Oxisol and Inceptisol. The trendlines are also plotted for the two soils combined. Each point comprises a mean of 24 (0-20 cm) or 18 (0-60 cm) plot yields and soil K values.

5.3.3. Nitrogen

Results from these measurements and analyses showed few significant findings and so are not included here but for completeness are given in Appendix 5.1 - 5.4.

5.3.4. Silicon

5.3.4.1. Effect of silicate application on topsoil (0-20 cm) properties

Similar results were obtained at all sampling events on both soils; the significant results from one sampling time at each trial site are thus used to illustrate these effects (Table 5.7). The application of Calmasil® and cement led to an increase in pH, and a concomitant decrease in acid saturation and exchangeable acidity and, at a number of sampling events, decreases in some micronutrients such as Zn and Cu (Table 5.7). Total cations were significantly increased by the addition of the siliceous materials, a phenomenon associated with the increase in variable charge on colloids resulting from pH increases (Wooldridge, 1989; Lemire *et al.*, 2006).

Si applied	C	a	N (mg	lg L⁻¹)	Si*	
(kg ha⁻¹)	Oxisol	Inceptisol	Oxisol	Inceptisol	Oxisol	Inceptisol
0	1246 ± 34 a [∓]	582 ± 21 a	159 ± 6 a	134 ± 5 a	15 ± 0 a	15 ± 1 a
300	1740 ± 35 b	894 ± 59 b	189 ± 5 b	156 ± 7 b	27 ± 1 b	27 ± 2 b
Sig.	P<0.001	P<0.001	P<0.001	P<0.01	P<0.001	P<0.001
Si applied	pH (CaCl ₂)		Zn (mg		Cu L ⁻¹)	
(kg ha⁻¹)	Oxisol	Inceptisol	Oxisol	Inceptisol	Oxisol	Inceptisol
0	4.35 ± 0.02 a	4.07 ± 0.02 a	6.8 ± 0.3 a	3.0 ± 0.1	1.8 ± 0.0 a	0.7 ± 0.0
300	4.60 ± 0.02 b	4.48 ± 0.06 b	5.8 ± 0.2 b	2.7 ± 0.1	1.7 ± 0.0 b	0.7 ± 0.0
Sig.	P<0.001	P<0.001	P<0.001	NS	P<0.001	NS
Si	Exchangea	able acidity	Acid saturation		Total cations	
applied	(cmo	l _c L ⁻¹)	(%	%)	(cmo	l₀ L ⁻¹)
(kg ha⁻¹)	Oxisol	Inceptisol	Oxisol	Inceptisol	Oxisol	Inceptisol
0	2.75 ± 0.11 a	1.26 ± 0.06 a	25.7 ± 1.2 a	23.1 ± 1.2 a	10.9 ± 0.1 a	5.6 ± 0.1 a
300	1.44 ± 0.09 b	0.68 ± 0.08 b	12.1 ± 0.8 b	11.7 ± 1.4 b	12.3 ± 0.2 b	6.8 ± 0.3 b
Sig.	P<0.001	P<0.001	P<0.001	P<0.001	P<0.001	P<0.001

Table 5.7: Effect of applied silicates (Calmasil® and blastfurnace cement) on selected topsoil (0-20 cm) properties in the second ratoon crop, 6 weeks (Oxisol) and 8 weeks (Inceptisol) after application of cement. Data are presented as mean ± s.e. (n=48)

*Local recommended threshold is 15 mg L^{-1} .

[†] Within soils and columns, values with different letters are significantly different to each other.

5.3.4.2. Effect of silicate application on mean soil profile characteristics following harvest

The effects of silicate application on soil profile properties were similar at each site and sampling time, and only those for the Oxisol (second ratoon) are presented (Table 5.8). The application of Calmasil® and cement led to significant increases in soil Ca throughout both profiles at all sampling times, along with increases in Mg, Si and pH at almost every sampling time. Exchangeable acidity and acid saturation decreased at two out of three sampling times at each site, while Cu, Zn and Fe were sometimes reduced, presumably also due to the increased pH (Meyer, 2013). The application of silicates significantly increased soil profile K (P<0.05) at only one sampling time (second ratoon) on the Inceptisol, and never on the Oxisol. Even nonsignificant effects of silicates on K at both sites were inconsistent, with K being slightly increased at times, and decreased at others (data not shown).

Si rate	Са	Mg	Si	Zn
(kg ha⁻¹)		(mg	L ⁻¹)	
0	650 ± 23 a [∓]	121 ± 3 a	16 ± 0 a	3.0 ± 0.2 a
300	856 ± 23 b	132 ± 3 b	19 ± 0 b	2.6 ± 0.2 b
Sig.	P<0.001	P<0.01	P<0.001	P<0.05
Si rate	pH (CaCl₂)	^a Exch. acidity	^b Acid sat.	
(kg ha⁻¹)		(cmol _c L ⁻¹)	(%)	
0	4.33 ± 0.01 a	2.84 ± 0.06 a	40.72 ± 0.98 a	
300	4.46 ± 0.01 b	2.20 ± 0.07 b	32.19 ± 1.29 b	
Sig.	P<0.001	P<0.001	P<0.001	

Table 5.8: Oxisol: Effect of applied silicates (Calmasil® and blastfurnace cement) on selected soil profile (0-100 cm) properties following the second ration crop harvest. Data are presented as mean ± s.e. (n=36)

^aExch. acidity = exchangeable acidity

^bAcid sat. = acid saturation

 $^{\intercal}$ Within columns, values with different letters are significantly different to each other

5.3.4.3. Silicon, calcium and magnesium movement down the profile

There was some movement of Si down the soil profile at both trial sites. Though Si had not moved deeper than 20 cm following the first ration harvest, by the second ration harvest, both soils had significantly more Si to a depth of 60 cm in the 300 kg Si ha⁻¹ treatment than in the 0 kg Si ha⁻¹ treatment (Figure 5.7). Increases in available Si in both the top- and subsoil, and Si movement were more pronounced in the lower-clay Inceptisol than in the Oxisol (Figure 5.7). Lower acid saturation in the Inceptisol (Table 3.2) may also have been partially responsible for the increased Si availability.



Figure 5.7: Mean extractable silicon (Si) per soil depth interval following the second ration crop in the (A) Oxisol and (B) Inceptisol. n=24 (0-20 cm) and n=18 (>20 cm). Within each trial site and depth interval, points with different letters are significantly different to each other (P < 0.001). Where no significant differences occurred between points within a depth interval, the points are circled.

Calcium movement down the soil profiles was limited, as expected. In both soils, Ca was significantly increased by silicate application for the first two crops in only the 0-20 cm soil layer. By the second ratoon harvest, this difference had extended to the 20-40 cm layer (Appendix 5.5).

Significant differences in Mg in the Oxisol between the two Si rates only extended to 20 cm (plant crop), 40 cm (first ratoon) and 20 cm (second ratoon; Appendix 5.6A). More movement of Mg occurred in the Inceptisol (Appendix 5.6B). This downward movement was most likely in response to the gypsum applied at the start of the trial (Alva *et al.*, 1998; Ernani *et al.*, 2006; Blum *et al.*, 2011). Magnesium leaching is generally greater on sandier soils (Ernani *et al.*, 2006), and thus the Inceptisol experienced greater Mg leaching than the Oxisol, despite a lower rate of gypsum application.

5.3.4.4. Effect of topsoil silicon on sugarcane yield and quality

Sugarcane sucrose yields were consistently highest where Si was applied. In four of the total six harvests across the two trial sites (all three on the Inceptisol and one on the Oxisol), these increases were significant (Section 3.3.4). Although the FAS recommended threshold for soil Si (across all soils) is 15 mg L⁻¹, increases above this level at the two trial sites always resulted in greater stalk and sucrose yields (Table 5.9), indicating that this threshold may be too low for these soils. In the Australian sugarcane industry, on soils with low Si reserves (H₂SO₄-extractable Si), a response to Si application is considered "possible" when available (CaCl₂-extractable) Si is between 10 and 20 mg kg⁻¹ (Calcino *et al.*, 2018). A response is considered "unlikely" when available Si is > 20 mg kg⁻¹.

	Oxisol			Inceptisol	
Available Si	Stalk yield	Sucrose yield	Available Si	Stalk yield	Sucrose yield
(mg L ⁻¹)	(0	%)	(mg L ⁻¹)	(%	b)
15 ± 0	95 a [∓]	95 a	15 ± 1	94 a	94 a
27 ± 1	100 b	100 b	27 ± 2	100 b	100. b
Significance	P<0.05	P<0.05		P<0.05	P<0.05
TARGE 1					(1

Table 5.9: Mean (n=48) sugarcane stalk yield (t cane ha⁻¹) and sucrose yield (t ERC ha⁻¹), as a percentage of the maximum yields obtained in the second ration harvests, in relation to available silicon (Si) (mean \pm s.e.) in the Oxisol and Inceptisol topsoil (0-20 cm)

⁺Within columns, values with different letters are significantly different to each other.

For the Inceptisol, raising soil Si to 27 mg L⁻¹ would have been economically beneficial in terms of the extra sucrose yields realised. On the Oxisol, however, significant yield increases were only obtained in the second ratoon crop (Table 5.9). Inconsistent yield increases, therefore, would not make application of 600 kg Si ha⁻¹ (over the three crops) economically beneficial on the Oxisol, though future ratoons may affect this conclusion.

5.3.4.5. Inclusion of subsoil silicon data into topsoil calibration

Stalk yield plotted against topsoil (0-20 cm) Si concentration measured after three harvests gives a strong (R^2 =0.95) relationship across the two soils (Figure 5.8). This indicates that a topsoil Si concentration of approximately 20 mg L⁻¹ would be an improvement on the current 15 mg L⁻¹ threshold. Inclusion of the subsoil (20-60 cm) data weakened the relationship (R^2 = 0.23; results not shown), possibly due to roots accessing Si, Ca and Mg at depths greater than 60 cm.



Figure 5.8: Mean sugarcane stalk yield, as a percentage of the maximum yield in each harvest, in relation to topsoil (0-20 cm) extractable silicon (Si) in the Oxisol (second ratoon harvest) and Inceptisol (first and second ratoon harvests). Each point comprises a mean of 48 plot yields and soil Si values.

5.4. Conclusions 5.4.1. Potassium

There were low concentrations (<1 cmol_c kg⁻¹) of slowly-available K (nitric K) in both soils and yet K release contributed substantially to readily-available soil K. Two findings illustrate this point. Firstly, readily-available K increased considerably, in both soils, between pre-trial sampling and the first sampling event following plant crop fertiliser application, in the zero-K plots. Secondly, the K balance over all of the crops at each site showed that, on average, more K was accounted for at the end of each crop, than was present at the start. It was noteworthy that the zero-K plots gave a higher percentage of K recovery than the treatments which received K. This may have been due to release of more slowly-available K levels. Based on these results, it is possible that nitric K values do not present a true picture of the amount of slowly-

available K in these soils, and that more K may become available to a crop than this measure suggests.

Readily-available soil K increased more and there was greater build-up of K over the duration of the study in the Oxisol than in the Inceptisol. As both soils received the same amount of K fertiliser, the increased readily-available K in the Oxisol may have been the result of greater quantities of organic matter in this soil retaining K in a form more readily available for exchange and thus measurement than in the Inceptisol. In addition, greater K fixation by the 2:1 clay minerals present in the Inceptisol may have limited readily-available K following K fertiliser application.

In the Oxisol, increasing topsoil K did not translate into statistically significant stalk yield increases. The total tonnes of sucrose obtained were, however, increased by increasing topsoil K, significantly so in some cases. Similar results were obtained on the Inceptisol, and though lower mean readily-available K, stalk and sucrose yield increases were measured, they were more often significant than on the Oxisol. Importantly, while much reliance is placed on statistical significance, in this study small increases in yield (and hence profitability) may prove economically significant to the farmer, even when not statistically significant. The results obtained confirmed that the local recommended threshold values used for both soils were appropriate.

The inclusion of subsoil K values improved the strength of the relationship between soil K and yield. The FAS recommendations should take this finding into account, and growers should be encouraged to have subsoil samples analysed in order to improve the accuracy of K fertiliser recommendations.

Potassium application significantly decreased topsoil Si in the Inceptisol at times. This was possibly due to increased plant uptake following higher yields in the higher K treatments on this soil.

5.4.2. Nitrogen

Conclusions are presented in Appendix 5.4.

5.4.3. Silicon

Application of silicate products led to the expected significant increases in soil pH, Ca, Mg, Si and total cations, along with significant decreases in exchangeable acidity and acid saturation. Zinc, Fe and Cu were also decreased at some sampling times. Silicate application significantly increased soil profile K once, on the Inceptisol. The effects of silicate application on K were non-significant at all other times. Downward Si movement through the soil profile was more pronounced on the Inceptisol than on the Oxisol. Downward Ca movement was limited in both soils, while Mg leaching occurred, especially in the Inceptisol. Increased soil Si was generally associated with increased stalk (and hence sucrose) yield. Inclusion of subsoil Si values weakened the relationship with yield. The results indicate that the current local recommended threshold of 15 mg L⁻¹ is too low for both soils and a higher threshold of approximately 20 mg L⁻¹ should be considered.

CHAPTER 6

Stratification of potassium and silicon in two sugarcane soils and the effect of sampling depth on fertiliser recommendations and costs

6.1. Introduction

Sugarcane soils undergo relatively little tillage in comparison with some other crops. In South Africa, sugarcane is replanted, on average, every eight cropping seasons (Lecler and Tweddle, 2010), which total between eight and sixteen years, depending on the climate of the area. Many sugarcane growers also use minimum tillage at replanting, whereby the old crop is sprayed with herbicide, and new rows are drawn using a ridging implement, without full ploughing (Meyer and van Antwerpen, 2010). Conventional tillage is illegal in South Africa on slopes steeper than 12-20%, depending on the soils' erodibility (Anonymous, 1983; McFarlane and Maher, 1993), rendering minimum tillage the only means of replanting on such slopes. Lime is incorporated, typically to a depth of 20 cm, and fertiliser applied in the furrows, at planting only. For each ratoon crop, fertiliser is generally applied to the soil surface.

Infrequent tillage, coupled with surface or topsoil nutrient application, may lead to stratification of nutrients in the soil, with implications for nutrient retrieval by crop roots. Organic matter (Franzluebbers, 2002; Sá and Lal, 2009) and less mobile nutrients such as potassium (K), phosphorus (P) and zinc (Zn) are most likely to become concentrated at or near the soil surface (Wright *et al.*, 2007; Fernández *et al.*, 2012; Cade-Menun *et al.*, 2015), especially in clay soils where downward movement of nutrients is more restricted (Askegaard *et al.*, 2004). Under certain conditions, where sugarcane surface feeder roots are absent or limited (due to management and environmental conditions), or if the surface soil should dry, these nutrients will be largely unavailable to the crop.

In Chapter 5, both the topsoil and the whole soil profile (down to 100 cm (Oxisol) or 80 cm (Inceptisol)) were studied, and movement of nutrients down the soil profile was discussed with a focus on the consequences for nutrient management and soil sampling decisions. However, nutrient stratification can also occur at a finer scale,

such as within the 'topsoil'. Most routine analytical laboratories make nutrient recommendations based on a certain standardised sampling depth. The Fertiliser Advisory Service (FAS) at the South African Sugarcane Research Institute (SASRI), for example, gives recommendations based on an assumed sampling depth of 0-20 cm (SASRI, 2004), as this is the depth at which samples have been collected in crop nutrition trials. Collecting samples at different depths from within the 'topsoil', in soils where nutrients are markedly stratified, may affect nutrient and lime recommendations. Samples collected from shallower depths, where soil is nutrient-rich, may result in recommendations lower than those required for good crop performance, leading to reduced yields. Samples collected from too deep, on the other hand, where nutrients are less concentrated (unless leaching and redeposition have occurred), will result in recommendations higher than required, leading to excessive expenditure and possible environmental pollution.

This chapter, therefore, reports on an investigation to determine (a) the extent to which K and silicon (Si) are stratified within the upper part of two contrasting soil profiles, and (b) the effect this stratification has on K and Si recommendations and costs based on samples collected at different depths.

6.2. Materials and methods

6.2.1. Trial details

These are given in Sections 3.2.1 to 3.2.3.

6.2.2. Incremental soil depth and topsoil (0-20 cm) sampling

Incremental soil samples were collected following application of fertiliser to the second ratoon crop at each site. This crop on the Oxisol started to grow in winter. Potassium and Si (cement) were applied 5 and 13 weeks, respectively, into the crop's cycle, while nitrogen (N) was applied when spring regrowth started, 17 weeks into the crop's cycle. Incremental soil sampling was conducted in October 2012, 3, 4 and 7 weeks after N, Si and K application, respectively.

On the Inceptisol (which started to grow in summer), N and K were applied 7 weeks into the crop's cycle, and Si, 10 weeks. Sampling was carried out in February 2013, 8 weeks after Si application and 11 weeks after K and N were applied.

Due to the intensity of sampling, not all plots were sampled. The first three replicates were sampled from plots including all four K rates, both Si rates and one (intermediate) N rate (80 kg N ha⁻¹ and 105 kg N ha⁻¹ on the Oxisol and Inceptisol, respectively) giving a total of 24 plots at each trial site. The soil samples were collected using four Beater augers (Beater, 1955), each with a hollow tubular metal tip that was inserted into the soil to depths of 2.5, 7.5, 15 and 28 cm. The first (2.5 cm) auger was used to collect 3-4 soil cores per row, in the middle of each interrow and a total of approximately 18-20 cores were collected from each plot. The 7.5 cm auger followed directly behind the first auger, collecting samples from the same holes, so that the depth increment 2.5-7.5 cm was collected. The 15 cm auger followed immediately, collecting soil from 7.5 to 15 cm, and finally the 28 cm auger collected soil from the 15 to 28 cm depth increment from the same holes as the previous augers. Samples from each depth were bulked per plot, thoroughly mixed, and analysed for exchangeable K, calcium (Ca), magnesium (Mg) and Si, and pH and exchangeable acidity at the FAS (Appendix 3.1).

Within one day of the incremental sampling, topsoil (0-20 cm) samples were collected using a Beater auger from the middle three interrows of each plot, with five soil cores collected from each. The fifteen cores from each plot were bulked, thoroughly mixed, and analysed, as for the incremental depth samples, at the FAS.

6.2.3. Statistical analysis

Differences within the mean determinants for each sampled depth interval were determined using analysis of variance (ANOVA) (Genstat – 14^{th} Edition, VSN International, 2011). Where treatments showed significant effects, values were analysed using the Holm Sidak multiple comparison separation or Fisher's protected least significant difference test to identify where the differences between treatment means lay. Differences were considered to be significant at P < 0.05. Mean K, acid saturation, organic matter and clay were determined for each soil depth interval, and for the depths 0-7.5 cm, 0-15 cm, 0-28 cm and 0-20 cm.

6.2.4. Fertiliser recommendations

Using the current FAS crop nutrition guidelines, fertiliser recommendations were calculated for the 0-7.5, 0-15, 0-28 and 0-20 cm depths. The K fertiliser requirement is calculated at the FAS by subtracting the soil test value (readily-available K) from a threshold value (based on the clay content, base status and predicted yield), and multiplied by a K requirement factor (the amount of fertiliser K required to raise the soil test value by 1 mg L^{-1}). This value is then modified according to whether the crop is burnt or not, and adjusted according to the nitric K value.

Recommended Si rates are not usually calculated based on soil Si concentrations. The amount of Calmasil® is determined by the lime recommendation, with Calmasil® being substituted in a 1:1 ratio for dolomitic lime. Each ton of Calmasil® contains approximately 100 kg Si. The Ca, Mg, exchangeable acidity and total cations in each depth interval (Appendix 6.1) were used to calculate lime (and hence Calmasil®) requirements for the same depth intervals as for K. Lime requirements are calculated at the FAS using a factor that takes into account lime quality and depth of incorporation, along with threshold acid saturation (usually 20%), exchangeable acidity and effective cation exchange capacity (Miles, 2014). In situations where no lime is required but extractable soil Si is $< 15 \text{ mg L}^{-1}$, Calmasil® may be recommended by sugarcane extension specialists (Miles, 2014), often at a rate of 3 t ha⁻¹. Although blastfurnace cement was applied to supply Si during the field trial, it is too expensive for commercial farm use and recommendations for this product were therefore not calculated. Current prices of potassium chloride (KCI) and Calmasil® were used to establish fertiliser costs and partial budgets calculated (Section 3.2.6).

6.3. Results and discussion

6.3.1. Stratification of potassium and silicon

6.3.1.1. Potassium

Potassium exhibited marked stratification within the upper 28 cm of both soils. The surface layers (0-2.5 and 2.5-7.5 cm) of all the plots, including the zero K treatments, contained the greatest amounts of readily-available K (Figure 6.1). This stratification was particularly evident in the Oxisol (Figure 6.1A), where differences between depth
intervals were significant even in the zero K plots where no K had been applied for the preceding three years. This was likely a residual effect from pre-trial fertiliser applications, together with returns of K from leaf material and K deposition during burning. In the Inceptisol, treatment differences became less marked at 7.5-15 cm, and non-significant at 15-28 cm (Figure 6.1B). Potassium is typically not very mobile in soil (McKenzie and Pauly, 2013), and remained near the soil surface even in the sandy Inceptisol. By the time the second ratoon crops were harvested, 11 months (Oxisol) and 14 months (Inceptisol) after this incremental sampling was conducted, readily-available K had shown some movement to 40-60 cm (Oxisol, Figure 5.3C) and 20-40 cm (Inceptisol, Figure 5.4C).

There was greater build-up of readily-available K in the Oxisol than in the Inceptisol, with over 600 mg K L⁻¹ measured in the top 2.5 cm of the Oxisol at the highest K rate, and only 312 mg L⁻¹ in the Inceptisol (Figure 6.1). As discussed in Chapter 5, this may be accounted for by the presence of vermiculite in the Inceptisol, and its absence in the Oxisol. Higher organic matter content in the Oxisol is also likely to have played a role in K retention.



Figure 6.1: Readily-available soil potassium (K) across both rates of applied silicon (0 and 300 kg ha⁻¹), in the (A) Oxisol and (B) Inceptisol, at four K application rates, following fertiliser application at the start of the second rate rate rate or crop. Within each soil, points with different letters are significantly different to each other (P < 0.001).

6.3.1.2. Silicon

Stratification of Si was far less marked than that of K. In the Oxisol, there were significant differences between the zero Si and 300 kg Si ha⁻¹ treatments at all depths (Figure 6.2), indicating downward movement of Si through the soil. While available Si was highest in the shallowest depth intervals (in the 300 kg Si ha⁻¹ treatment), it was only significantly greater than at one other depth in this treatment. The blastfurnace cement was incorporated to a depth of approximately 3 cm (Section 3.2.3) but the Si did not remain at this depth, and downward movement was evident despite the cement being applied only 4 weeks before sampling. Movement may have been promoted by the above-average rainfall experienced during September and October 2012 (Figure 3.2), between application and sampling. Leaching is an important process determining soil Si concentration (Berthelsen and Korndörfer, 2005; Tubana *et al.*, 2016) and may have occurred to a sufficient extent to move applied Si into the deepest depths sampled.



Figure 6.2: Oxisol: Extractable soil silicon (Si) at two Si application rates following fertiliser application at the start of the second ratoon crop (n=4). Points with different letters are significantly different to each other (P = 0.001).

On the Inceptisol, Si from the Calmasil® had not reached a depth of 15-28 cm by the second ratoon, 3 years after application (Figure 6.3). There had been some

downward movement of Si, however, as indicated by significant differences between the zero and 300 kg Si ha⁻¹ treatments to a depth of 7.5-15 cm. The trends in Figure 6.3 appear to indicate considerable movement over a relatively short period, as indicated by the sharp increase in Si at 7.5-15 cm (but no deeper). It is possible that this Si 'bulge' could have been Si from the Calmasil®, but a more likely cause was movement of Si from the blastfurnace cement, applied 8 weeks before sampling. Downward movement of this Si may have been greater than that from Calmasil® due to its smaller particle size enhancing solubility. Approximately 33% of Calmasil® particles are greater than 200 μ m (Keeping, 2017), while less than 5% of cement particles are typically larger than 90 μ m (Thomas and Jennings, 2014).



Figure 6.3: Inceptisol: Extractable soil silicon (Si) at two Si application rates following fertiliser application at the start of the second ratoon crop (n=4). Points with different letters are significantly different to each other (P = 0.001).

6.3.2. Fertiliser recommendations based on sampling depth 6.3.2.1. Potassium

Increasing sampling depth significantly affected K fertiliser recommendations and costs (Figure 6.4). Comparison of the 0-20 cm K recommendation (from the 100 kg K ha⁻¹ treatments) with those from the other depth increments illustrated different points at the two trial sites.



Figure 6.4: The Fertiliser Advisory Service potassium (K) recommendation and K fertiliser purchase cost, based on samples collected at different depths from the 100 kg K ha⁻¹ plots (n=6) during the second ration crop on the Oxisol and Inceptisol. Within each soil, columns with different letters are significantly different to each other (P < 0.05).

On the Oxisol, the standard 0-20 cm sample K recommendation was not significantly different to those for 0-7.5 and 0-15 cm. The 0-28 cm recommendation was, however, significantly higher than that for any other depth (Figure 6.4). Fertiliser costs were R0, R978, R644 and R1 838 ha⁻¹ for the 0-7.5, 0-15, 0-20 and 0-28 cm depths, respectively (Figure 6.4). Sampling to 28 cm instead of 20 cm would, therefore, result in fertiliser costs being almost trebled. Increasing the K application above 100 kg K ha⁻¹ led to no further (statistical) increase in yield (Figure 3.4). Thus applying 150 kg K ha⁻¹ (the recommended amount based upon the 0-28 cm sample) would have caused unnecessary expenditure of approximately R600 ha⁻¹, with no extra income likely (Table 6.1).

 Table 6.1: Partial budget: Economic effect of basing potassium (K) fertiliser application rate

 on soil samples collected at 0-28 cm instead of 0-20 cm (recommended depth) in the second

 ratoon crop on the Oxisol

Added income due to change: No additional yield for > 100 kg K ha ⁻¹ application	(R ha ⁻¹)	Added costs due to change: *Cost of extra K fertiliser (50 kg K ha ⁻¹ @ R12 kg ⁻¹ K)	(R ha⁻¹) 600		
		[•] Cost of extra K fertiliser application ([€] R0.26 kg ⁻¹ KCI)	26		
Reduced costs due to change:	(R ha ⁻¹)	Reduced income due to change:	(R ha⁻¹)		
None		None			
Subtotal	0	Subtotal	626		
Not obspace B0 $= B626 = B626 hs^{-1}$					

Net change: $R0 - R626 = -R626 ha^{-1}$

*Though the FAS recommendation for 0-20 cm was 53 kg K ha⁻¹, this study did not include this K rate, so the 'extra' K fertiliser for 0-28 cm sampling was based on the difference between 100 kg K ha⁻¹ (closest treatment to the 0-20 cm recommendation) and 150 kg K ha⁻¹ (0-28 cm).

[€] Tin and string' method (see Glossary)

On the Inceptisol, the 0-20 cm recommendation was not significantly different to that from 0-28 cm, while the two shallowest depths returned K recommendations about 8% and 22% of that from 0-20 cm (Figure 6.4). In this soil, too little K would have been recommended had the sample been taken at a shallower depth than the calibrated 0-20 cm. Assuming linear interpolation between yield points in Figure 3.6, a loss of approximately R1 200 ha⁻¹ would have been incurred if fertiliser recommendations were based on 0-15 cm samples rather than the recommended 0-20 cm samples (Table 6.2).

Table 6.2: Partial budget: Economic effect of basing potassium (K) fertiliser application rate on soil samples collected at 0-15 cm instead of 0-20 cm (recommended depth) in the second ratoon crop on the Inceptisol

Added income due to change:	(R ha ⁻¹)	Added costs due to change:	(R ha ⁻¹)
None		None	
Reduced costs due to change:	(R ha⁻¹)	Reduced income due to change:	(R ha⁻¹)
*Reduction in K fertiliser (78 kg K ha ⁻¹ @ R12 kg ⁻¹ K)	936	0.5 t sucrose ha ⁻¹ yield reduction (R4 450 t ⁻¹ ERC)	2 250
*Cost of extra K fertiliser application ([€] R0.26 kg ⁻¹ KCI)	41		
Subtotal	977	Subtotal	2 250
Net change: R977 - R2 250 = -R1 27	'3 ha ⁻¹		

*Though the FAS recommendation for 0-15 cm was 40 kg K ha⁻¹, this study did not include this K rate, so the reduction in K fertiliser for 0-15 cm sampling was based on the difference between 100 kg K ha⁻¹ (closest treatment to the 0-15 cm recommendation, rounding up) and 178 kg K ha⁻¹ (0-20 cm).

[€]Tin and string' method

6.3.2.2. Silicon / Lime

There were no significant differences between the lime recommendations (based on a threshold of 20% acid saturation) for any of the depths on the Oxisol (Figure 6.5). On average, on this soil, Calmasil®/cement application did not affect yield (Section 3.3.4.1.1).

On the Inceptisol, sampling to 28 cm would have returned a significantly greater recommendation than that of the 0-20 cm depth (Figure 6.5). This would come at a significantly higher cost to the grower. For instance, the 0-20 cm sample returned a lime recommendation of $1.3 \text{ t} \text{ ha}^{-1}$, at a cost of R1 297 ha⁻¹ (applied) (Calmasil®), while the recommendation for a 0-28 cm sample was $2.8 \text{ t} \text{ ha}^{-1}$, at a cost of R4 020 ha⁻¹ (applied), a 148% increase in cost. However, results in Chapter 3 (Figure 3.14B) showed that application of $3 \text{ t} \text{ ha}^{-1}$ lime (300 kg Si ha⁻¹) resulted in a significant increase in sucrose yield over the 0 t ha⁻¹ treatment. The data do not indicate whether application of $1.3 \text{ t} \text{ ha}^{-1}$ lime (based on the 0-20 cm depth sample) would have been sufficient to increase sucrose yield.



Figure 6.5: The Fertiliser Advisory Service lime recommendation and lime purchase cost, based on samples collected at different depths from the zero silicon plots (n=12) during the second ratoon crop on the Oxisol and Inceptisol. Within each soil, columns with different letters are significantly different to each other (P < 0.05).

6.4. Conclusions

Potassium exhibited marked stratification within the upper 28 cm of both soils. Readily-available K was greatest at the surface and decreased with depth. This stratification was particularly evident on the Oxisol that had higher clay and organic matter than the Inceptisol. As a result, soil samples collected at different depths resulted in different K fertiliser recommendations. Samples collected from close to the soil surface indicated that little K fertiliser was required, possibly leading to reduced yields. Samples collected deeper than the 20 cm recommended by the FAS resulted in recommendations that were likely to be too high, and therefore significantly more expensive.

Although Si stratification was less marked than that of K, lime recommendations were nonetheless significantly affected by sampling depth on the Inceptisol. When samples were collected at a depth greater than that recommended by the FAS, lime

recommendations were substantially higher than for samples collected from the recommended depth.

These results highlight important management considerations. Although deep rooting systems are important, especially for accessing water and nutrients, surface rooting should be encouraged on both soils to take advantage of the nutrient-rich topsoil layers. This can be done by mulching in the form of a cover crop, or the retention of the crop residue after sugarcane harvest. These findings underline the importance of the recommended best management practice of maintaining a surface cover.

Secondly, soil samples for nutrient recommendations must be collected at the depth specified by the advisory service to which the samples are sent. Collecting samples at other depths will likely lead to either under- or over-fertilisation, with considerable yield and economic consequences.

CHAPTER 7

The effect of potassium and silicon application on sugarcane root distribution in two soils

7.1. Introduction

In the South African sugarcane industry, there is concern that sugarcane roots are often stunted, blackened and unhealthy, and apparently not reaching the depths reported in earlier studies (6 m, Evans, 1936; 5 ft / 1.52 m, Wood and Wood, 1967). Consequently, research at the South African Sugarcane Research Institute (SASRI) includes investigations into the extent and possible causes of this phenomenon. The present research project therefore included a study of root distribution as part of the field study (Sections 1.3.2, 2.6.1 and 2.6.3).

Potassium (K) encourages healthy root development and reduces susceptibility to root rots (Melis and Farina, 1984). Cavalcante *et al.* (2015) reported a 53% decrease in root density and reduced root dry matter production in sugarcane grown in a K-deficient nutrient solution, compared with a K-replete solution.

Silicon (Si) application has been reported to improve water use efficiency and plant performance under conditions of drought stress in a range of plants, including sorghum (Hattori *et al.*, 2005; Ahmed *et al.*, 2011), maize (Gao *et al.*, 2004), wheat (Gong *et al.*, 2005), soybean (Shen *et al.*, 2010), rye (Hattori *et al.*, 2009), and sugarcane (Savant *et al.*, 1999). Sonobe *et al.* (2011) investigated the root responses of sorghum to Si application with and without drought stress. They found that the reduction in seedling dry weight due to water stress was alleviated by Si application, and water uptake was increased. Silicon nutrition, therefore, may also play a role in root development.

Due to the arduous nature of root sampling and observation, however, information on the effects of nutrition on root growth and density in the field is limited. This study was therefore conducted to (a) determine the effects of K and Si on sugarcane total root length and distribution, and (b) establish whether different rates of these nutrients affect root growth.

7.2. Materials and methods 7.2.1. Trial details

These are given in Sections 3.2.1 to 3.2.4.

7.2.2. Root sampling

Root samples were collected 5.5 months (Inceptisol) and 6 months (Oxisol) after harvest of the first ration crop, i.e., during the growth of the second ration crop. A sampling date between 5 and 6 months after harvest was selected to coincide with sufficient maturity of the new crop's root system, at a time when the old root system would have died off (considered to be within approximately 4 months of harvest; Smith *et al.*, 2005). Samples were collected during the summer months, in a period of favourable growth conditions. On the Oxisol, sampling occurred 6 weeks after nitrogen (N) application, 2.5 months after blastfurnace cement (Si) application and 4.5 months after application of K and phosphorus (P). On the Inceptisol, sampling occurred 3 months after cement application, and 4 months after application of N and K. Roots were sampled from intermediate N treatments (80 kg N ha⁻¹ on the Oxisol; 105 kg N ha⁻¹ on the Inceptisol), on both Si treatments (0 and 300 kg Si ha⁻¹), and on three of the K treatments (0, 100 and 300 kg K ha⁻¹). All four replicates were sampled, resulting in a total of 24 plots being sampled at each trial site.

Corers (stainless steel pipes, 3.8 cm internal diameter and approximately 1.2 m in length) were constructed in-house at SASRI and used to collect bulk soil and root samples. The end of the pipe inserted into the soil was tapered slightly to assist soil penetration, and the other end was reinforced to allow it to be hammered into the soil using a heavy mallet. The sampling pipes were marked at 20 cm increments, to 100 cm. In each sampled plot, eight holes were made between the stools (clumps) of sugarcane, within the five nett rows. Each depth core (0-20, 20-40, 40-60 and 60-80 cm and, on the Oxisol only, 80-100 cm) from each hole was placed in a separate plastic sampling bag. Practical difficulties, including extremely hard soil at

depth and soil getting stuck in the corers, resulted in some plots being sampled to only 60 or 80 cm on the Oxisol.

To separate the roots, the samples were placed into a 1.5 L bucket and soaked in tap water for 30 minutes before being poured through a 1 mm sieve and all soil washed off. The roots remaining on the sieve were collected with tweezers, placed into an airtight 50 mL container and scanned on a flatbed document scanner (HP Scanjet 4570c). The scanned images were analysed using APS Assess 2.0 software (Lamari, 2008), yielding total root length per sample.

7.2.3. Statistical analysis

The total root length per depth interval dataset was unbalanced due to the fact that some holes had been sampled to only 60 cm, while others were sampled to greater depth. The effect of soil type and depth interval on the mean total root length per depth interval was therefore analysed using Residual Maximum Likelihood (REML) (Genstat – 14^{th} Edition, VSN International, 2011). Total root length per hole to 60 cm was calculated. Differences in total root length between K and Si treatments, and soil type, were established using three-way analysis of variance (ANOVA) (Genstat – 14^{th} Edition, VSN International, 2011). For both REML and ANOVA, where treatments showed significant effects, values were analysed using the Holm Sidak multiple comparison separation test to identify where the differences between treatment means lay. Differences were considered to be significant at P < 0.05.

7.3. Results and discussion

7.3.1. Profile root distribution

In the Oxisol, 46% of the total root length per sample hole (0-100 cm) was found in the top 20 cm, and 69% in the top 40 cm. In the Inceptisol, which was sampled to 80 cm, 49% of total root length was found in the top 20 cm, and 73% in the top 40 cm (Figure 7.1). Most sugarcane root biomass is typically found close to the surface, declining, approximately exponentially, with increasing depth (Smith *et al.*, 2005). Root length density follows a similar pattern (Smith *et al.*, 2005). Blackburn (1984) reported that approximately 50% of sugarcane root biomass is found in the top 20 cm of soil, and 85% in the top 60 cm. This equates to an

'extinction coefficient' for root distribution, *b*, of ~0.967, where the cumulative proportion of roots (Y) to a certain depth, *d* (cm), can be calculated as $Y = 1 - b^d$ (Gale and Grigal, 1987). Using this relationship, approximately 74% of the roots should have been found in the top 40 cm of soil, which is very close to the value found in both soils.



Figure 7.1: Mean (\pm s.e.) total length of roots per sampled depth interval in the Oxisol (n = 24) and Inceptisol (n = 24), across all potassium and silicon treatments sampled. Bars with different letters are significantly different to each other (P < 0.001).

Van Antwerpen (1999) reported that 57% of total sugarcane roots (sampled to 195 cm) were found in the top 45 cm of a non-irrigated clay soil (32-44% clay), while in a sandy soil (12% clay), this figure was 38%. The lower percentage of roots in the latter was due to the need for more roots to access the subsoil to obtain water (Van Antwerpen, 1999). This distribution was not evident in the present study, where a greater proportion of measured roots was found in both topsoils. It is probable that some roots in the current study were not sampled due to growth being below the lowest sampling depth leading to some overestimation of the proportion found in the topsoil. The lithocutanic B horizon (Soil Classification Working Group, 1991) of the Inceptisol may also have restricted downward root growth, in contrast to the deep sands in the Van Antwerpen (1999) study. The higher proportion of roots found in the Oxisol topsoil than that found in the high clay soil of the Van Antwerpen study may have been due to variations in root growth under dryland (non-irrigated) conditions.

Van Antwerpen (1999) lists rainfall as an important determinant of root growth. At the Oxisol site, above-average rainfall was recorded in the 3 months prior to root sampling, which may have led to slightly lower rates of root extension into the subsoil as the roots had access to sufficient moisture in the topsoil.

Root density in the top 20 cm ranged from 0.08 to 1.25 cm cm⁻³ in the Oxisol, and from 0.05 to 1.39 cm cm⁻³ in the Inceptisol. These values are within and below the lower range of the root density reported by Van Antwerpen (1998), of 0.5 to 2.7 cm cm⁻³. Again, above-average rainfall during the months prior to root sampling on both the Oxisol and Inceptisol may have resulted in less need for root proliferation.

7.3.2. Effect of soil type and applied nutrients on total root length

In both soils the amount of K applied had no effect on the total root length to a depth of 60 cm (Table 7.1). While deficiencies in P or N tend to foster preferential partitioning of photosynthetic carbon to the roots, leading to an increase in the root : shoot dry weight ratio (Cakmak *et al.*, 1994), the opposite is true for K and magnesium (Mg) deficiencies (Marschner *et al.*, 1996).

Treatment		Soil
(kg ha⁻¹)	Oxisol	Inceptisol
K	Root	ength (cm)
0	$^{+}$ 200.2 ± 9.1	316.1 ± 11.3
100	213.0 ± 10.4	306.7 ± 9.5
300	187.4 ± 9.6	306.2 ± 12.7
Significance	NS	NS
Si		
0	204.7 ± 8.4	327.3 ± 8.4 a
300	195.7 ± 7.5	292.0 ± 9.5 b
Significance	NS	P<0.01
Soil	Mean (± s.e.) total root length (cm) per soil type, over all treatments
Oxisol	200	.2 ± 5.6 a
Inceptisol	273	.5 ± 5.9 b
Significance	Р	<0.001

Table 7.1: Mean (\pm s.e.) total root length in samples collected from 0-60 cm, 6 months (Oxisol) and 5.5 months (Inceptisol), after harvest of the first ration crop from potassium (K) and silicon (Si) treatments. n=8 (K rates) and n=12 (Si rates) at each site

[†] Within soils and treatments, values with different letters are significantly different at the P level specified.

Potassium concentrations in the 0-20 cm depth of the 0 kg K ha⁻¹ treatments were 63 mg L⁻¹ and 47 mg L⁻¹ in the Oxisol and Inceptisol, respectively (Table 7.2), both well below the SASRI Fertiliser Advisory Service's 90 t cane ha⁻¹ thresholds of 155 mg L⁻¹ and 93 mg L⁻¹, respectively (Miles, 2014). In the 300 kg K ha⁻¹ treatment, 188 mg K L⁻¹ and 123 mg K L⁻¹ were present in the Oxisol and Inceptisol, respectively (Table 7.2). In this treatment, K status of the soils was well above the K thresholds, yet root length was not greater than that on the zero-K treatments.

Table 7.2: Readily-available potassium (K) and extractable silicon (Si) to 100 cm (Oxisol) and 80 cm (Inceptisol), in the 0, 100 and 300 kg K ha⁻¹ and the 0 and 300 kg Si ha⁻¹ treatments, in samples collected 4 weeks (Oxisol) and 6 weeks (Inceptisol) after harvest of the first ratio crop. n=8 (K rates) and n=12 (Si rates) on each soil

Nutrient	Application rate	Sample depth	Readily-available K and extractable Si $(mg L^{-1})$		
	(kg ha⁻¹)	(cm)	[∓] Oxisol	Inceptisol	
		0-20	63 de	47 ab	
		20-40	37 abc	37 ab	
	0	40-60	23 a	34 a	
		60-80	20 a	38 ab	
		80-100	14 a		
		0-20	94 f	70 c	
		20-40	57 bcde	41 ab	
	100	40-60	24 ab	34 a	
Potassium		60-80	22 a	44 ab	
		80-100	18 ab		
	300	0-20	188 g	123 e	
		20-40	90 ef	53 b	
		40-60	39 abcd	41 ab	
		60-80	31 ab	51 b	
		80-100	28 abc		
	Significance		P<0.001	P<0.001	
		0-20	15 e	14 abc	
	0	20-40	13 d	13 abc	
		40-60	10 c	12 ab	
		60-80	8 b	12 a	
		80-100	6 a		
		0-20	19 f	16 c	
Silicon	300	20-40	14 de	15 bc	
		40-60	10 c	14 abc	
		60-80	8 b	13 abc	
		80-100	7 ab		
	Significance		P<0.001	P<0.001	

⁺Within soils and nutrients, values with different letters are significantly different at the P level specified.

It is possible that the sugarcane roots may have obtained some K from depths below those measured. In the Inceptisol, for instance, mean readily-available K from 80-420 cm (i.e., below the depths routinely measured in this study) was 30 mg L⁻¹ (data not shown), with some depth layers up to 57 mg K L⁻¹. This may explain the fact that the zero-K plots did not show a significant reduction in root length in comparison with the K-treated plots. Less readily-available K pools may also have contributed towards supplying the K required to maintain rooting in the zero-K treatments. Other results indicated that less readily-available K became available during the growing season, possibly providing more K than that indicated by the nitric K measurements (Section 5.3.2.6).

Silicon application on the Inceptisol led to a significant decrease in total root length (0-60 cm), in comparison to the zero-Si treatment (Table 7.1). Silicon was significantly higher in the topsoil of the 300 kg Si ha⁻¹ treatment than in the zero-Si treatment (Table 7.2). Application of Si may affect sugarcane rooting in a number of ways. Firstly, the application of Calmasil® and cement led to a significant increase in soil pH and a decrease in exchangeable acidity and acid saturation (Section 5.3.4.1; Table 5.7). Application of such siliceous materials should, therefore, encourage root growth (Sumner, 2011). Secondly, their application led to slight decreases in leaf N, this decrease being significant at one sampling time (data not shown). A model (Thornley, 1972) predicts increased biomass partitioning towards the shoot under conditions of increased N concentration. Decreased plant N, with Si application, would therefore be predicted to decrease shoot biomass, favouring root growth instead. Thirdly, the silicate materials applied contained Mg (Section 3.2.3), raising the soil's Mg status (Section 5.3.4.2; Table 5.8). This would overcome any Mg deficiency, and should therefore increase biomass partitioning to the roots. None of these effects, however, explain the measured decrease in root length following Si application. This may, however, be a result of the improved water use efficiency reported following Si application that reduces the need for the crop to establish an extensive root system. Sugarcane plants subjected to water deficit tend to grow deeper root systems (Smith et al., 2005), with rainfed plants typically having deeper root systems than irrigated crops (Van Antwerpen, 1998). In general, grasses subjected to water stress tend to prioritise root growth and in sugarcane an increase in root dry matter has been reported (Queiroz et al., 2011). It is possible, therefore, that the plants' reaction to water stress in the zero-Si treatments may have led to an increase in biomass accumulation in the roots, whereas the applied Si reduced the plants' stress reaction in the 300 kg Si ha⁻¹ treatment. It is also possible that application of Calmasil® and cement could have raised the soil electrical conductivity (not measured in this study), possibly restricting root growth to some extent.

The Inceptisol had significantly greater total root length than the Oxisol (Table 7.1). The Oxisol had better water-holding characteristics than the Inceptisol, including higher clay and organic matter contents (41% and 3.9%, respectively, at 0-20 cm in the Oxisol; 24% and 1.4% in the Inceptisol; Table 3.2). In addition, the Oxisol site had received higher rainfall (approximately 1 110 mm) during the second ratoon crop (from harvest of the previous crop to root sampling), than the Inceptisol (approximately 770 mm). It is therefore likely that the second ratoon crop (and possibly all crops) on the Inceptisol experienced greater water stress than on the Oxisol, leading to greater root extension in the Inceptisol. Soil strength can also inhibit root growth and clay soils may have significantly reduced root systems compared with their sandy or loam counterparts (Smith *et al.*, 2005). The higher clay percentage in the Oxisol may therefore have inhibited root growth. As mentioned in Section 7.2.2, difficulties were encountered during root sampling to depth on the Oxisol, highlighting the possibility of soil strength reducing root growth.

7.4. Conclusions

Approximately 70% of the roots sampled were found in the top 40 cm of the soil at each site. The concern within the South African sugar industry of roots not reaching depths reported by earlier workers does not seem to apply to the two trial soils where the root distribution appears to be normal according to the proportion of roots expected to be in the upper part of the soil (Blackburn, 1984; Gale and Grigal, 1987).

Unexpectedly, K application had no effect on total root length. It is possible that roots may have obtained some K from deeper in the soil profile, and also from less readily-available K pools, to maintain root growth in the zero-K treatment. Application of Si, on the other hand, decreased total root length to a depth of 60 cm, significantly so in the Inceptisol. In addition, total root length in the Oxisol was less than in the Inceptisol. Water stress may be implicated in both responses, with the effect of increasing total root length where water was limited, or where plants experienced water stress for other reasons such as soil texture. The lower total root length (in the Inceptisol) in the 300 kg Si ha⁻¹ treatment suggests that the reported amelioration of drought stress following Si application is likely not due to an increased root system, but rather to other mechanisms such as, amongst others, increased water uptake by the roots or decreased water loss from the leaves (Zhu and Gong, 2014).

CHAPTER 8

Potassium exhaustion study of an Oxisol and an Inceptisol, using Italian ryegrass (*Lolium multiflorum* L.)

8.1. Introduction

As with most crops, sugarcane responds to a shortage of essential nutrients through a reduction in crop performance or yield. Low readily-available (exchangeable) potassium (K) can inhibit sugarcane germination (Anderson and Bowen, 1990), and lead to a decrease in cane yield and quality (Wood and Schroeder, 2004). Soil K thresholds have been established in many countries (Anderson and Bowen, 1990) and when soil K is below this threshold, sugarcane yields will be compromised, unless fertiliser K is applied.

For some soils, measures of readily-available K provide an accurate estimate of their K-supplying power. In other soils, however, less available (non-exchangeable) K may become available to crops, sometimes within a single growing season (Adamo et al., 2016). High crop yields can then be maintained even where readily-available soil K is below the accepted threshold. In the present field trials, sugarcane did not show a marked yield response to K application, even on the Inceptisol, where the pre-trial readily-available K was below threshold. A K balance (Section 5.3.2.6) indicated that more K was taken up by the crop than was originally present in readilyavailable form in the soils. To examine this further a K 'exhaustion' pot trial, using soil from the two trial sites, was established. Italian ryegrass (Lolium multiflorum) was grown. It is commonly used in K exhaustion trials because it has a very high K requirement per unit dry matter produced, re-grows rapidly after harvest, has an intensive root system, can be planted densely, and can be grown in small containers (Wood and Burrows, 1980; Samadi, 2011). The objectives of this pot trial were to determine a) whether K could be exhausted in these soils, as evidenced by a marked reduction in ryegrass yield and K content; b) if the K was not exhausted, whether more K was being released, and taken up by the plants, than was originally

readily available in the soil; and c) information regarding the K-supply potential of the two soils.

8.2. Materials and methods *8.2.1. Pot trial procedure*

Topsoil (0-10 cm) was collected with a spade at four locations in two replications from the 0 kg K ha⁻¹, 300 kg Si ha⁻¹, 80 kg N ha⁻¹ (Oxisol) or 105 kg N ha⁻¹ (Inceptisol) treatment plots. The soil was collected following harvest of the second ratoon crop on the Oxisol, and 12 months into the growth of the second ratoon crop on the Inceptisol. The soils were bulked at each site, air-dried for 8 days at room temperature, milled and passed through a 1 mm sieve.

The mass of 500 mL of each soil was determined and used to calculate the amount of each nutrient to apply to the pots, assuming a plough depth of 20 cm. The equivalent of 5 t ha⁻¹ dolomitic lime (ryegrass is acid-sensitive) and 100 kg P ha⁻¹ (as calcium phosphate) were thoroughly mixed into each topsoil. Six 500 mL plastic pots, each with four 5 - 6 mm diameter drain holes were filled with each treated topsoil, to within 1.5 cm of the top. Filter paper was placed inside each pot before filling with soil, in order to prevent soil loss. Sub-samples of the potting soils were analysed by the South African Sugarcane Research Institute's Fertiliser Advisory Service (FAS) for pH, clay, organic carbon, P, K, calcium (Ca) and magnesium (Mg) (Appendix 3.1).

Twenty-five long-duration Italian ryegrass (*Lolium multiflorum* L., cv. Feast II) seeds were spread evenly over the soil in each pot, and covered with a further 0.5 cm of the fertilised soils. Ten millilitres of the trace element mixture Trelmix® was added to 2 L of distilled water, the rate recommended by the suppliers. Undiluted Trelmix® contains boron (1.0 g L⁻¹), copper (3.0 g L⁻¹), iron (21.3 g L⁻¹), manganese (2.7 g L⁻¹), Mg (0.3 g L⁻¹), molybdenum (0.3 g L⁻¹) and zinc (2.3 g L⁻¹).

Mottram *et al.* (1981) and Van Antwerpen *et al.* (1994) derived equations to relate field capacity to topsoil clay percent, and the mean value of these equations was used to calculate the volume of water required to bring each soil to field capacity. The appropriate volumes of the Trelmix® solution were applied to each pot, and any

seeds exposed during watering were gently pushed back into the soil, to a depth of 0.5 cm.

The pots were randomly placed into two 43 cm x 30.5 cm x 8.8 cm plastic trays. At one end of each tray, a 7.2 cm length of 7.6 cm diameter PVC pipe was placed vertically. At the top and bottom of each of these pipes, four slots, each 1.0 cm x 0.3 cm wide, had been cut, one in each quadrant of the circular pipe. This allowed a 1.75 L bottle to be filled with distilled water, turned upside down and placed so that the neck of the bottle rested on the top of the PVC pipe, while the slots in the pipe allowed a depth of 1 cm of water to be maintained within the trays at all times. This method of irrigation followed that outlined by Portch and Hunter (2002) and allowed capillary action and root demand to regulate the uptake of water by the plants. The trays were placed in a glasshouse, where minimum and maximum temperatures were recorded for the duration of the trial. Temperatures ranged from 17.0 to 33.0°C in summer, and 14.5 to 28.0°C in winter.

Initial germination of the ryegrass was poor due to crusting of the soils. The soil surfaces were therefore disturbed and the pots reseeded at a rate of 50 seeds pot⁻¹, and removed from the water trays for all but 10 to 15 minutes per day. Subsequent germination was favourable and a dense stand was obtained in each pot. Plants were not thinned. One month after the initial planting, limestone ammonium nitrate (LAN, 28% N) was crushed and broadcast over the surface of each pot at an equivalent rate of 100 kg N ha⁻¹. The Trelmix® solution was applied, at the same rates as before, to the newly fertilised pots. For the remainder of the trial, LAN was applied at the same rate every 21 days and the Trelmix® solution every 42 days.

Sixty days after sowing, the ryegrass was harvested by cutting at 5 cm above the soil surface. At harvest, the grass was at the 'three-leaf stage', where three leaves had grown on the primary tiller of each plant. Because the amount of ryegrass harvested per pot was insufficient for analysis, all replicates were combined at each harvest event. The material was weighed immediately (data not shown) and then dried at 70°C for 48 hours. The material was re-weighed, ground and stored. Harvests were again conducted each time the primary tillers reached the three-leaf stage (112, 161, 207, 249, 291 and 340 days after sowing), and the material treated as before. At the last harvest, the stubble (the 5 cm of aerial plant material remaining after harvest)

was cut at ground level and treated in the same way as the leaf samples. A subsample of soil (± 50g) from each pot was analysed at the FAS as before. To separate the roots, the remaining soil from each pot was placed into a plastic beaker and covered with tap water to soak for 4 hours before being washed under running tap water over a 1 mm sieve. The clean roots were dried at 70°C for 72 hours, weighed and stored for analysis. The root, stubble and leaf samples from all harvests were analysed at the FAS for total K by inductively coupled plasma-OES spectrometry (Varian ICP 720-ES) following acid digestion.

8.2.2. Data analysis

The total amount of K removed by the plants for the duration of the pot trial was calculated. This total included the cumulative total K in the leaves from each harvest, as well as the stubble and root K measured after the final destructive harvest. Combining the replicates to obtain enough plant material for analysis precluded statistical analysis of the data.

8.3. Results and discussion

The Oxisol and Inceptisol differed in texture and nutritional status. The Oxisol had higher clay, organic carbon, P and Ca contents and readily-available K was 60% higher than in the Inceptisol (Table 8.1). However, both soils were below the soil K threshold for Italian ryegrass of 100 mg kg⁻¹ (0.26 cmol_c kg⁻¹) (Manson, 1995).

Soil	pH (CaCl₂)	Clay	Organic carbon	Ρ	Readily- available K	Exch. Ca	Exch. Mg
		((g kg ⁻¹)	(mg kg⁻¹)	(cmol _c kg ⁻¹)	
Oxisol	5.2	340	51	59	0.24	10.00	1.37
Inceptisol	4.9	190	21	36	0.15	5.12	1.39

Table 8.1: Initial properties of soils used in the ryegrass pot trial (following the application of dolomitic lime and calcium phosphate basal treatments)

8.3.1. Ryegrass uptake of potassium and potassium balance

With the exception of the first harvest, the ryegrass grown on the Inceptisol had a higher leaf K content than that on the Oxisol (Figure 8.1). Leaf K was above 3% at the first harvest on each soil, but dropped to below 2% from the second harvest. By the final harvest, leaf K was 0.43% (Oxisol) and 0.90% (Inceptisol) (Figure 8.1).



Figure 8.1: Ryegrass leaf potassium (K) and cumulative K removed in harvested leaves over time on the (A) Oxisol and (B) Inceptisol.

Manson (1995) and Miles et al. (1986, cited in Manson, 1995) reported a critical leaf

K threshold of 3.0% for optimal ryegrass production in KwaZulu-Natal, South Africa, while McDonnell *et al.* (2018) gave a satisfactory range of 1.6 - 2.0% K in south-western Australia, the higher concentration targeting the first two ryegrass harvests. Both soils, therefore, provided sufficient K for the first harvest, but leaf K dropped below the sufficiency level of 1.6% for the second harvest. For the remainder of the pot trial, leaf K in the ryegrass grown on the Oxisol was below 1% (Figure 8.1A), while that on the Inceptisol exceeded 1.6% for the third, fourth and fifth harvests, before dropping below this for the last two harvests (Figure 8.1B). Cumulative leaf K removal appeared to be leveling off in both soils (Figure 8.1A and B) – more particularly so in the Oxisol. This, coupled with the declining leaf K concentration, indicates that the soils may have been nearing K exhaustion.

Although the Oxisol ryegrass leaf (Table 8.2) and stubble (Table 8.3) had a higher dry mass than that grown on the Inceptisol (possibly due to higher N mineralisation on the higher organic matter Oxisol), the K concentration (K%) was greater on the Inceptisol (Table 8.3). The net result was that a greater amount of K was removed by the ryegrass from the Inceptisol (269 kg K ha⁻¹) than from the Oxisol (155 kg K ha⁻¹) (Table 8.3). Figure 8.1A and B, where only leaf K is shown, illustrates a similar pattern. A considerable amount of K was thus removed from these soils. Interestingly, root mass was greater in the Inceptisol than the Oxisol (Table 8.3). The same pattern was found with sugarcane root length (Table 7.1) and may be a function of soil texture and consequent water or nutrient availability.

Days after planting	Dry mass of harvested leaf material (g pot ⁻¹)			
	Oxisol	Inceptisol		
60	0.57	0.49		
112	1.76	1.54		
161	1.13	0.81		
207	1.58	1.33		
249	1.27	1.05		
291	1.14	1.07		
340	1.27	1.08		
TOTAL	8.72	7.37		

Table 8.2: Ryegrass leaf dry matter at each harvest on the Oxisol and Inceptisol

Soil	Plant part	Dry mass of harvested plant material	K content	
	•	(g pot⁻¹)	(%)	(kg ha⁻¹)
Oxisol	Stubble	4.82	0.21	17.04
	Roots	5.98	0.09	8.98
Inceptisol	Stubble	4.15	0.60	42.62
	Roots	9.68	0.08	12.57
TOTAL: kg K ha ⁻¹ equivalent, over all plant parts, at final harvest				
			Oxisol:	155.2
		I	nceptisol:	268.6

Table 8.3: Ryegrass stubble and root dry matter and potassium (K) content at final harvest(340 days after planting) and total K removed by leaf, stubble and roots

At the end of the pot trial, almost 94% of the original readily-available K in the Oxisol was accounted for, with almost 76% taken up by the ryegrass and 18% remaining in the soil (Table 8.4). By contrast, the ryegrass grown on the Inceptisol took up 181% (269 kg K ha⁻¹) of the original readily-available soil K (149 kg K ha⁻¹), while 41% (61 kg K ha⁻¹) remained in the soil after the pot trial. A total of 330 kg K ha⁻¹, or 222% of the original amount, was thus taken up by the ryegrass or remained in the soil after the pot trial.

Table 8.4: Potassium balance in the rye	grass pot trial on the Oxisol and Inceptisol
---	--

Property	Oxisol	Inceptisol
Pre-trial: readily-available K in soil (kg ha ⁻¹)	205.54	148.56
Post-trial: readily-available K in soil (kg ha ⁻¹)	36.68	61.15
Difference: readily-available K (pre-trial) – readily-available K (post trial) (kg ha ⁻¹)	168.86	87.41
% of pre-trial K remaining in soil after the trial	17.85	41.16
Readily-available K removed by ryegrass (kg ha ⁻¹)	155.21	268.63
% of original readily-available soil K removed by ryegrass	75.51	180.82
% readily-available K unaccounted for by ryegrass removal and remaining soil K	6.64	-121.98

More K was therefore taken up from the Inceptisol, despite having approximately 60 kg ha⁻¹ less readily-available K than the Oxisol before the trial started. This

apparent inconsistency may be explained by the mineralogy of the soils. The Inceptisol topsoil contained six times more total K and the parent material approximately ten times that of the Oxisol (Table 5.2). Slowly-available (nitric) K was also greater in the Inceptisol than in the Oxisol (Table 5.3). In the topsoil (0-20 cm), the Inceptisol contained 661 kg ha⁻¹ nitric K, approximately 3.5 times more than the Oxisol (Table 5.3). If the total potentially available pool of K is assumed to be the sum of the readily-available and nitric K fractions, the Oxisol had 395 kg K ha⁻¹ and the Inceptisol 810 kg K ha⁻¹. The ryegrass took up 39% of this total on the Oxisol, and 33% on the Inceptisol. All of the K taken up by the ryegrass on the Oxisol could be accounted for by the readily-available K (Table 8.4), while two thirds of the Inceptisol ryegrass K removal was supplied by the nitric K pool. This indicates that more K became available for slow release to the crop in the Inceptisol than in the Oxisol.

It has been shown that K⁺ ions can be extracted by plants from, particularly, illitic and smectitic clay minerals, within a single growing season (Adamo et al., 2016). Illite was present in the clay fraction of both topsoils, while vermiculite was present in the Inceptisol, but not the Oxisol (Table 5.1). The extent of the release from the Inceptisol may have been due to the large amount of total K in the topsoil, as well as the presence of illite, vermiculite and K-feldspars that facilitated the slow release of K. These results also raise the possibility that even though the Oxisol contained more organic matter than the Inceptisol, its ability to supply K was less than that of the minerals (and perhaps the organic matter) in the Inceptisol. The organic matter in humic soils is considered to be very stable (due, at least in part, to the presence of aluminium-organic matter complexes), inhibiting its activity with regard to K release. Roberts (1968) reported annual patterns of K release from soil organic matter and retention by the microbial fraction, indicating that the various organic matter fractions have different K dynamics. Although not analysed, it is possible that the organic matter in the Inceptisol - perhaps with a greater labile proportion - may have released more K than the Oxisol.

8.3.2. Soil potassium release: comparison between pot and field potassium balances

Direct comparison of the absolute amounts of K release and uptake between the pot and field trials is not possible, due to the very different trial conditions. However, patterns of supply and uptake by the ryegrass and sugarcane can be examined.

In the pot trial, 94% of the original, readily-available soil K in the Oxisol was recovered in the final soil and plant parts. The 6% that was not accounted for may have been leached, or lost in small amounts of root material that were not recovered during the harvesting and root-washing procedure. In the field K balance conducted on the Oxisol (Section 5.3.2.6), an average of 129% of the original, readily-available soil K was recovered after harvest in the plant parts and that remaining in the soil (Table 5.4). Considering that the field trial K balance did not take the K in crop roots into account, 129% was a conservative measure, likely under-estimating the actual K uptake. Both field and pot trials indicated that pre-trial readily-available soil K was used from slowly-available sources.

In the Inceptisol, 222% of the original readily-available soil K was recovered in the plant and that remaining in the soil after the pot trial. A large amount of K was therefore taken up by the ryegrass which was not readily available at the start of the trial. In the field K balance, 194% of the original soil K was recovered in the plants and remaining soil K. Again, as root K was not taken into account in the field K balance, 194% of the K recovery. Smith *et al.* (2005) reported that sugarcane root dry mass was approximately 15% of the aerial dry matter at 200 days after planting, though this proportion declines with age. At even half of this figure (7.5%), more than 200% of original soil K would likely have been recovered in the field trial had root K been measured. The recovery values obtained from both pot and field trials further indicate that K which was not initially readily-available, had become available to the crop, and to a much greater extent than in the Oxisol. These data confirm the results reported in Chapter 5 and again may explain the relatively limited sugarcane yield responses to K application (Chapter 3).

The rate (as opposed to just the total amount) of K release was also evidently greater in the Inceptisol than that in the Oxisol. The pot trial (11.2 months) appeared to give sufficient time for the Inceptisol to release slowly-available K, while this was not the case on the Oxisol.

8.3.3. Long-term potassium supply

While it is evident that the K supply had not been exhausted by the end of the pot trial, the data indicate that the soils were nearing this point. Figure 8.1A, in particular, shows a flattening of the cumulative K uptake curve on the Oxisol, and, to a lesser degree, Figure 8.1B indicates a similar trend for the Inceptisol. Although ryegrass leaf Κ concentration naturally decreases with increasing maturity (McDonnell et al., 2018), two facts indicate that K supply was becoming limited in the Oxisol, in particular, and possibly in the Inceptisol. Firstly, above-ground dry matter decreased by 17% (Oxisol) and 12% (Inceptisol) between the first half of the trial (mean of harvests two, three and four) and the second half (harvests five, six and seven). Harvested ryegrass yields were, therefore, not sustained to the same extent in the second half of the trial, though this may be partly explained by the season -Italian ryegrass thrives in winter, while the trial ended in mid-summer. Secondly, readily-available pre-trial K was 206 mg L⁻¹ in the Oxisol and 149 mg L⁻¹ in the Inceptisol; at the end of the pot trial, it was 37 and 61 mg L⁻¹, respectively (Table 8.4). Both soils, but particularly the Oxisol, therefore showed a marked reduction in readily-available K over the duration of the trial.

In considering the long-term K supplying power of the two soils, neither the field nor the pot trials ran for long enough to reach K exhaustion, but the percentage K removals in the trials give an indication of when this might theoretically occur. Approximately a third of the total potentially available (readily-available + nitric) K was taken up by the ryegrass on each soil (Section 8.3.1). If a new ryegrass crop were planted, as would typically be the case with annual (Italian) cultivars, and if it is assumed that similar K use would occur, it would take 869 days (2.4 years) for topsoil K to become completely exhausted in the Oxisol, and 1 017 days (2.8 years) in the Inceptisol. On both soils, however, the rate of nitric K release could prove a limiting factor in terms of K supply for crop growth. In the pot trial, the Oxisol supplied sufficient K, in addition to other nutrients such as N, to sustain yields above those on the Inceptisol. On the other hand, K concentration in the Inceptisol ryegrass was much higher than that on the Oxisol (Figure 8.1), apparently made possible by faster K release from slowly-available pools in this soil. This may then lead to faster K exhaustion on the Inceptisol although it could be more than counterbalanced by the much greater total K present in this soil (Table 5.2).

It is perhaps instructive, in light of the evidence from both the field and pot trials, to consider apparent total K reserves to rooting depth for the ryegrass and sugarcane. In the pots, the soil depth available for the ryegrass roots to exploit provided approximately 395 kg K ha⁻¹ in the Oxisol, and 810 kg K ha⁻¹ in the Inceptisol (total of readily-available and nitric K). Ryegrass roots can reach depths of over 100 cm when stressed (Steynberg *et al.*, 1994), but most are commonly found within 30-60 cm (Fessehazion *et al.*, 2012). To a depth of 60 cm, 766 kg K ha⁻¹ would be potentially available on the Oxisol, and 2 943 kg K ha⁻¹ on the Inceptisol. This amount of K would, in theory, take about 5 years and 11 years, respectively, for the ryegrass to use to exhaustion on each soil. Such estimates are very strongly influenced by the nitric K value in both soils.

A similar calculation for the field trials using total potentially-available soil profile K and sugarcane K removal data results in times to reach K exhaustion of about 17 years on the Oxisol and 63 years on the Inceptisol. If only the readily-available K values are used then these times reduce to about 7 years on the Oxisol and 5 years on the Inceptisol. Practical experience renders all these potential exhaustion times purely theoretical as over such extended periods K would need to be added in order to ensure adequate yields. Significantly, by the ratoon crops in the present field trials, both soils showed indications of yield reduction in the zero-K control treatments.

A portion of the potentially available K (readily-available + nitric K) would not be accessible to the plant, at least within a practically significant timeframe. There are a number of possible reasons for this:

 Potassium depletion will likely continue until a certain soil-specific minimum K level is reached, after which only very small amounts of K will be released (Øgaard *et al.*, 2002; Løes and Øgaard, 2003).

- The measurements of soil K, both readily-available and, especially, nitric K, may not be sufficiently sensitive to predict either its availability or accessibility to plants due to differences in the inorganic and organic fractions between soils, the ability of different plants to access the various K pools or incomplete exploitation of soil by the roots.
- If plants are unable to extract nitric K directly from its source, the replenishment of the readily-available pool from the slowly-available pool may be too slow to ensure adequate growth.

The calculations and discussion above indicate that the importance of nitric K is as an indicator of the relative size of the slowly-available K pool. In the Australian sugarcane industry, K recommendations are reduced by varying amounts (depending on soil texture) where nitric K exceeds $0.7 \text{ cmol}_c \text{ kg}^{-1}$ (Calcino *et al.*, 2018). Interestingly, the Inceptisol topsoil had $0.7 \text{ cmol}_c \text{ kg}^{-1}$ nitric K (Table 5.3), increasing into the subsoil. Its K release was markedly greater in both the pot and field trials than that in the Oxisol, with a topsoil nitric K of $0.22 \text{ cmol}_c \text{ kg}^{-1}$, decreasing into the subsoil.

In South Africa, a minimum threshold value of 1.8 cmol_c kg⁻¹ is used, above which K fertiliser recommendations are reduced (Miles, 2016). As the nitric K value increases, K recommendations are reduced further. The various (conservative) thresholds were calculated by adding the standard error of estimation in the mid infra-red calibration (0.9) to the thresholds proposed by Haysom (1971). When the slowly-available K measure was introduced by the FAS in 2016, it was anticipated that as more data became available, this figure could be amended. Evidence from both the pot and the field trials in the present study suggest that these current FAS limits should be reconsidered, with there being an urgent need for further field trial work on a range of soil types in order to obtain more definitive criteria for the interpretation of the nitric K test.

8.4. Conclusions

A greater amount of K was removed by the ryegrass from the Inceptisol than from the Oxisol, despite higher original levels of readily-available K and, at most harvests, higher ryegrass dry matter on the Oxisol. Most of the original K in the Oxisol could be accounted for at the end of the pot trial. On the Inceptisol, however, an extra 122%, over the original readily-available K, was present in the combination of the ryegrass dry matter and the soil. Thus, a large amount of K was released from 'non-exchangeable' sources in the Inceptisol during the course of the pot trial. This finding supports the differences in amounts of total K and slowly-available (nitric) K (Chapter 5) between the two soils. Furthermore, the indication from the pot trial is that considerable amounts of this K could be released within a single growing season in the field.

The field trials showed that the application of K fertiliser would have resulted in a financial loss on both soils, as it was evident that both soils contained sufficient K to support at least three harvested sugarcane crops (though there was some evidence of yield decline in the zero-K treatments in the ratoons). The pot trial results support the field trial findings, indicating that, particularly in the Inceptisol, a significant amount of K was released from the soil and used by the crop.

Both soils, however, were able to release sufficient K to allow continuous growth of ryegrass without reaching K exhaustion, although there was evidence to indicate that K supply was becoming more limited. The reasons for this may be found in differences in both the mineral and possibly, organic, fractions of the two soils. It is possible that the more stable and base-depleted organic matter present in the Oxisol was associated with reduced availability of K, compared to the situation in the Inceptisol. Although the mineralogy of the two soils was similar, as both contained illite and K-feldspars which are able to supply slowly-available K, the latter were more abundant in the Inceptisol. The presence of small amounts of vermiculite in the Inceptisol may also have facilitated greater K release. The indication is, therefore, that a better understanding of, firstly, the properties of the organic material in the different soils and, secondly, their mineralogy, would further aid in predicting the potential extent of K release, and hence the formulation of K fertiliser recommendations and the prevention of K mining.

CHAPTER 9

A summary conceptual model, general discussion, conclusions and future work

9.1. Introduction

As outlined in Sections 1.3 and 2.6, this study was undertaken to assist in the improvement of potassium (K), nitrogen (N) and silicon (Si) recommendations given by the Fertiliser Advisory Service (FAS) at the South African Sugarcane Research Institute (SASRI). This work resulted from an identified need at SASRI for more research into K and Si in particular. To achieve this the project proposal structure included factorial K x N x Si field trials, established on two contrasting soils, an Oxisol and an Inceptisol, to examine the effects of these applied nutrients and their interactions on sugarcane yield parameters and soil, leaf and rooting characteristics. Pot trial(s) were envisaged in the proposal to allow investigation into factors identified in the field trials which might require further elucidation.

The conclusions in each chapter have highlighted a number of aspects which may be used to improve the recommendations given by the FAS. This chapter presents a summary conceptual model and outlines the major findings of both the field trials and the pot trial and suggests where further work is necessary to either support recommendations or to increase understanding of the processes involved.

9.2. Conceptual model: Effects of potassium, nitrogen and slag-based silicate fertiliser on sugarcane yield and quality

A conceptual model was developed to summarise the possible effects of K, N and Si (slag-based) fertilisers on sugarcane yield and quality (Figure 9.1A and B). It includes suggestions for possible changes to nutrient recommendations for the South African sugar industry (pending, in some cases, further study). The following sections discuss further the aspects outlined in the model.

Α.



Β.



Figure 9.1: Conceptual model summarising possible effects of (A) potassium (K) and nitrogen (N), and (B) silicon (Si) (slag-based) fertilisers on sugarcane yield and quality, along with recommendations for soil sampling and management. Included are suggestions for possible changes to nutrient recommendations. %ERC = percent estimated recoverable crystal; t ERC = tonnes ERC; PC = plant crop; R2 = second ratoon crop; N.S. = non-significant differences; \downarrow = decrease; \uparrow = increase; Δ =change; ? = requires further research.

9.3. Nutrient recommendations 9.3.1. Potassium

In general, the K thresholds and recommendations currently used by the FAS appear appropriate for the Oxisol and Inceptisol under study. Sucrose yields were significantly increased by the application of 200 or 300 kg K ha⁻¹ in three of the six harvests. Depending upon the amount of K initially present in each soil, the FAS recommendations would have been 250 kg K ha⁻¹ for both soils' plant crops, and approximately 200 kg K ha⁻¹ for the ratoons. Despite the sucrose yield increases in some crops, at current fertiliser and sucrose prices, an application rate of 200 kg K ha⁻¹ was not economically justifiable over the first three crops at each site, but it is possible that the economics might have become more favourable in later ratoons, where response to K is usually more marked. This study also found that combining subsoil (20-60 cm) K data with those from the topsoil (0-20 cm) improved the relationship between soil K and crop yield.

Although K application led to a significant increase in sucrose yields in half of the harvests, the other three crops did not respond to K. This indicates that, at least in some seasons, soil K release was able to supply much of the crop's requirement. This K release was clearly seen in the field trial K balance analysis where, especially in the zero-K plots, large amounts of K were measured at the end of the crop cycle which were not accounted for by either fertiliser or original, readily-available soil K. The ryegrass pot trial showed a very similar effect, especially on the Inceptisol, supporting the field trial results.

In the K balance calculations from both the field and pot trials the Inceptisol showed greater K release than the Oxisol. Despite this, however, the nitric K reserves in both soils were low (< 1 cmol_c kg⁻¹). This may suggest that the K released was neither initially readily-available nor nitric extractable. This then focuses attention on the soil mineralogy as a major driver of K supply to the crop and points to the possibility of the nitric K test being relatively insensitive to the availability of the K reserves in some clay minerals and perhaps also from within primary minerals such as feldspars. In the context of routine soil testing, the characterisation of mineralogy in each grower sample is currently not an option. The findings in this study, however,

indicate a need for a re-evaluation of the currently-used nitric K soil test thresholds at the FAS to ensure appropriate advice is given on crop K requirements.

9.3.2. Nitrogen

This study confirmed the depressive effects of excess N on sucrose percent and that plant crops respond less to N than ration crops. Plant crop yields, sucrose percent and total sucrose yield did not respond positively to applied N and, on the humic Oxisol, N application served to significantly decrease sucrose yield in the plant crop. Reduced N application rates, as are currently recommended by the FAS on humic soils and in plant crops, were therefore confirmed by this study. However, the evidence also suggested that plant crops grown on humic soils, and perhaps other soils with a high organic matter content, may not require any N fertiliser at all. The economic analysis conducted in this study confirmed that on the Oxisol, it would be financially detrimental to apply N in the plant crop. This finding is of importance, given that the FAS currently recommends N application in every plant crop, on all soil types.

Nitrogen recommendations for the Oxisol would have been 70 kg N ha⁻¹ in the plant crop, and 110 kg N ha⁻¹ in the ratoons; for the Inceptisol 115 kg N ha⁻¹ in the plant crop, and 155 kg N ha⁻¹ in the ratoons. Application of N at, in general, slightly lower levels than the FAS recommendations, proved economically beneficial, particularly on the Inceptisol. The exception to this was, as stated above, the plant crop on the Oxisol, where zero N would have resulted in the highest sucrose yields.

9.3.3. Silicon

The results suggest that the current soil Si threshold of 15 mg L⁻¹ is too low to promote optimal yield. It is recommended that the FAS increase this threshold to an interim value of 20 mg L⁻¹ (pending confirmation), particularly in light of the fact that results from this study (and others) indicate poor uptake of Si by sugarcane. In contrast to K, inclusion of subsoil Si when calculating nutrient recommendations was not beneficial.

Application of Calmasil® at the rates used in this trial were uneconomic over the three crops measured. Over a longer term, however, continued soil acidification and
depletion of Si reserves would ultimately necessitate applications of these liming sources to allow the continued production of sugarcane.

The finding that high soil K may limit calcium (Ca) and magnesium (Mg) uptake by the plant is of considerable significance to the South African sugar industry and has implications for lime recommendations. Although this cationic antagonism is not directly related to Si application, when soil Si is below the threshold, it is recommended at present that a calcium silicate slag (such as Calmasil®) be used in place of the dolomitic lime requirement of the soil, as it similarly provides Ca and Mg (as well as Si) and raises soil pH. Currently, high soil exchangeable Ca and Mg lead to increased FAS K recommendations. Based on the outcomes of this study, it is necessary to consider the reverse situation and to increase Ca and Mg (and hence lime or Calmasil®) recommendations in soils with high available K.

Sugarcane grown on the Oxisol responded poorly to applied Si; on the Inceptisol, however, application of 300 kg Si ha⁻¹ led to significantly higher sucrose yields. This increase was a result of significantly greater stalk yields, rather than an increase in sucrose percent.

9.3.4. Nutrient interactions

There were generally few consistent interactions between the three main nutrients applied, though a few notable exceptions were found.

9.3.4.1. Nitrogen and silicon

On the second ration crop on the Oxisol, increasing N had no effect on sucrose percent, when zero Si was applied. Where Si was applied, however, increasing N caused a significant decrease in sucrose percent. This interaction was most likely an effect of the Si-containing liming material, leading to increased soil N release, rather than the Si itself.

Nitrogen application led to a significant increase in soil profile Si following each ratoon crop harvest on the Oxisol. This may have been linked to reduced crop removal of Si on these treatments.

9.3.4.2. Potassium and silicon

Potassium application significantly decreased topsoil Si in the Inceptisol at times, possibly due to increased plant uptake following higher yields in the higher K treatments on this soil.

Silicon application increased soil profile K only once, on the Inceptisol. The effects of silicate application on K were non-significant at all other times.

9.3.4.3. Nitrogen, silicon and potassium

There was a significant interaction between the effect of K, N and Si on sucrose percent in the combined ratoon crops on the Oxisol. Under certain conditions, the deleterious effect of high N rates on sucrose percent (exacerbated by high Si/lime rates) were mitigated by the application of sufficient K.

9.3.5. Sampling and crop nutrition

There was poor correlation between leaf nutrient levels and crop yields such that low leaf nutrient concentrations did not necessarily correspond with reduced sugarcane or sucrose yields. Moisture stress during the field trials probably led to this disparity, with insufficient moisture having been shown to render leaf sample results unreliable in terms of predicting sugarcane yields.

This study showed that the soil sampling depth recommended by the FAS should be strictly adhered to. Collection of samples at a shallower or deeper depth than that calibrated for in the laboratory will lead to incorrect recommendations, resulting in too little or too much of each nutrient being applied. Of the three nutrients studied, K and Si recommendations were affected to a greater extent by sampling depth than N. The stratification of nutrients, with higher concentrations of K (in particular) and Si in the upper few centimetres of soil, underlines the importance of encouraging surface rooting (along with the deep rooting system required), in order to take advantage of this nutrient-rich layer.

9.3.6. Rooting

The concern within the South African sugar industry about shallow rooting did not apply to the two trial soils, where the root distribution appeared normal. Unexpectedly, K application did not affect root length, indicating that roots may have obtained some K from deeper in the soil profile and perhaps from less readily-available K pools, to maintain root growth in the zero-K treatment. Silicate application decreased total root length to a depth of 60 cm. In addition, root length in the Oxisol was less than in the Inceptisol. Texture and water stress may be implicated in both responses, with the effect of increasing total root length where water was limited, or where plants experienced water stress for other reasons.

9.4. Future work 9.4.1. Potassium

To improve the K recommendations currently made by the FAS, the current study has highlighted the following two main areas that warrant consideration.

9.4.1.1. Subsoil

At present, the FAS uses only topsoil K results to formulate recommendations. Inclusion of subsoil K data improved the strength of the correlation between readilyavailable soil K and crop yield. Further trial work is necessary to refine this relationship and to determine optimal recommendations for different soil types.

Stratification of K (particularly in clay soils), along with utilisation of subsoil K by roots, may lead to high concentrations of K in the upper soil layers, and a corresponding lack of K at depth. If stunted, shallow rooting continues to be seen as a problem within the South African sugar industry, it may be necessary to increase subsoil K concentrations. Although incorporation of K to depth may pose practical difficulties, the downward movement of K following gypsum application may assist in this regard. Further investigation could elucidate a possible K requirement factor (KRF) for subsoils, based on mineralogy, clay percent or other soil characteristics, which could indicate the amount of K fertiliser required to raise the subsoil K test by one unit.

9.4.1.2. Non-exchangeable potassium release

Both the field and pot trials indicated that a substantial amount of K was released from non-exchangeable pools within a single growing season, even though the soils were classified as having 'low' K-supplying capability. These results and those from other work at SASRI (Miles and Farina, 2014; and work leading to the publication of Elephant *et al.*, 2018) resulted in 'reserve K' (determined by mid infra-red spectrometry (MIR)) being included in 2016 in the routine FAS suite of analyses. The MIR reserve K is a direct measure of nitric K and, at present, a value greater than 1.8 cmol_c L⁻¹ results in a reduction in the recommended K fertiliser rate, with higher reserve K values leading to further reductions. Inclusion of MIR reserve K has thus improved the value and accuracy of the FAS K recommendations by taking this release into account.

Further work should continue to include not only the nitric K test (or a proxy for it such as using infra-red technology), but also incubation tests and pot trials to investigate the amount of K released by soils within each range of nitric K results, as well as KRFs. Clay mineralogy has been shown in this study and others to have a marked effect on soil K reserves and KRFs (Elephant and Miles, 2016), and an indication of mineralogy, and its routine measurement, would further aid in prediction of the potential extent of K release, and hence the formulation of K fertiliser recommendations.

Findings reported in this study point to a definite need for the nitric K thresholds currently used by the FAS to be re-evaluated, and also that more accurate KRFs be established. Such work is likely to lead to K fertiliser recommendations being substantially reduced on some soil types, leading to cost savings without sacrificing yield. It should always be kept in mind, however, that longer-term vision is required to ensure sustainability, such that K reserves are not exhausted. Continued depletion of soil K reserves is unsustainable in the long term (Kingston, 2000) and it is not possible to continue "mining" soil K without eventually decreasing crop production (Askegaard *et al.*, 2004).

The results of the present study suggest that the occurrence of vermiculite in the Inceptisol facilitated greater release of slowly-available K than that from the Oxisol, even though the latter soil contained more organic matter, which is also a source of K (Al-Kanani *et al.*, 1984). Metson *et al.* (1956, cited in Haysom, 1971) concluded that K released during boiling with nitric acid is released from primary minerals, especially from the more inaccessible surfaces of hydrous mica, while Al-Kanani *et al.* (1984) indicated chlorite and vermiculite. Further research is warranted into the reasons for this release and the factors that control it.

9.4.2. Nitrogen

Results from this study indicated that N recommendations for plant crops grown on humic soils could be reduced to zero without negatively affecting yield. In many cases, a zero-N recommendation may even increase plant crop yields on these soils, where N application may sometimes significantly reduce yield due to, it appears, high N release from the soil. The applicability of this recommendation should be investigated to determine whether it should be a general recommendation for humic soils and possibly others with a high organic matter content. The reason for the depressed plant crop cane yield is also a topic of interest for future research.

9.4.3. Silicon

9.4.3.1. Silicon uptake

Silicon uptake from Calmasil® and cement was poor. This is in line with the findings of other sugarcane researchers, and this aspect should continue to form part of future work, given the importance of Si in promoting crop protection, health and yield.

Researchers have suggested that poor plant uptake of applied Si on weathered (desilicated) soils might be linked to elevated soil acidity and available aluminium (Tavakkoli *et al.*, 2011b). The yield response to Si on the Inceptisol may have been linked to slightly improved Si uptake by the plant facilitated by lower acid saturation in the Inceptisol than the Oxisol. In research by Keeping (2017), Calmasil® application led to the highest plant Si uptake of the six Si carriers tested, apparently due to its alkaline nature. Application of lime prior to Si application did not, however, improve Si uptake (Keeping *et al.*, 2017). The quest for effective and affordable

methods of increasing sugarcane uptake of Si continues, though research suggests that the currently used Si carrier (Calmasil®) is the best one available at present (Keeping, 2017; Keeping *et al.*, 2017).

9.4.3.2. Silicon thresholds

Increased soil Si was generally associated with increased stalk (and hence sucrose) yield. However, the current local recommended threshold of 15 mg L⁻¹ soil Si was too low for both soil types. A more appropriate threshold would appear to be closer to 20 mg Si L⁻¹, but this would need to be confirmed in further trial work.

Poor plant uptake of Si meant that leaf Si seldom reached even marginal levels. It is noteworthy, however, that the greatest percentage increases in sucrose yield were obtained with the greatest percentage increases in leaf Si concentration. On the Oxisol, where measured leaf Si never exceeded 0.58%, a significant stalk yield increase was obtained only in the second ratoon crop, where both the greatest absolute leaf Si increase, and the highest leaf Si percentage, were obtained (Section 4.3.3.1). The use of a single 'ideal' leaf Si threshold across the whole industry may be unrealistic, given the poor uptake of Si in the rainfed areas of the industry. Future research into the adoption of different leaf Si thresholds in different areas may be warranted. Under conditions where plant uptake is low, Si application should be encouraged, regardless of whether a (possibly unrealistic) minimum threshold can be obtained.

9.4.3.3. Cationic antagonism

The very marked reduction in Ca and Mg uptake by the plant under conditions of high soil K has not been widely reported in sugarcane. These antagonistic effects raise the possibility of Ca and Mg deficiencies being induced under conditions of excess soil K supply. It is suggested that Ca and Mg recommendations be amended so that, when soil K is above a certain threshold, Ca and Mg (and hence lime or Calmasil®) application rates be increased in order to overcome this antagonism. Future work should establish a robust threshold value/values which apply to different sugarcane varieties and soil types.

9.4.4. Nutrient interactions

Two of the nutrient interactions warrant further investigation. In one ratoon crop on the Oxisol, Si application led to increased N release and therefore decreased sucrose percent. Although it is likely that this was a liming effect rather than an effect of Si itself, Calmasil® - a liming agent – is currently the most affordable and effective means of Si application for South African sugarcane farmers. This interaction may have relevance for eldana (*Eldana saccharina*) control. Damage inflicted by this stalk borer is reported to increase with increasing N application, and Si application is recommended as a means to reduce eldana numbers without reducing N (Keeping *et al.*, 2014). If sucrose percent is lowered by this practice, however, grower revenue may be negatively affected. Future investigations into N x Si interactions on eldana damage should therefore include measures of crop yield and sucrose percent.

Related to the above interaction is the finding that, in the ratoon crops on the Oxisol, K negated the decrease in sucrose percent with concurrent Si and N application. This effect has bearing on the eldana question as a possible means of reducing any negative effects of Si application on sucrose content, and warrants further research.

9.4.5. General

9.4.5.1. Inclusion of subsoil data into nutrient recommendations

It is evident from the current data that inclusion of subsoil nutrient concentrations would improve soil : yield relationships for some, but not all, nutrients. Further elucidation of this point is required to establish appropriate soil profile (0-60 cm) threshold values for each nutrient for which subsoil data are considered useful.

9.4.5.2. Leaf sampling

It was not possible from the results obtained in the present study to state, definitively, that a certain leaf nutrient level was high enough to support optimal yield. Future work should determine the extent to which moisture stress can affect the analytical results of leaf samples sent to the FAS, and possibly whether a different approach should be followed for leaf sampling. Miles (2010) suggested improvements to the procedure and techniques to strengthen the overall efficacy of leaf sampling, and so

improve the establishment of nutrient sufficiency values of the crop. In particular, a modeling approach, taking into account decreases in the leaf N threshold with increasing biomass, was recommended (Miles, 2010). Studies are currently underway at SASRI to investigate such improvements.

9.4.5.3. Rooting

The effect of Si application on the decrease in root length that was significant in the Inceptisol requires further investigation, given the widespread use of Si products in the South African sugarcane industry and the general impression of poor rooting. In addition, understanding the lack of a rooting response to K application and the origin of the K used by the crop is necessary. Further work should ascertain whether these results are widespread and frequent occurrences or a feature of the specific conditions at the trial sites.

REFERENCES

Adamo P, Barré P, Cozzolino V, di Meo V, Velde B. 2016. Short term clay mineral release and re-capture of potassium in a *Zea mays* field experiment. *Geoderma* 264: 54-60.

Ahmad W, Watts MJ, Imtiaz M, Ahmed I, Zia MH. 2012. Zinc deficiency in soils, crops and humans. *Agrochimica* 56: 65-97.

Ahmed M, Hassen F-u, Qadeer U, Aslam MA. 2011. Silicon application and drought tolerance mechanism of sorghum. *African Journal of Agricultural Research* 6: 594-607.

Al-Kanani T, MacKenzie AF, Ross GJ. 1984. Potassium status of some Quebec soils: K released by nitric acid and sodium tetraphenylboron as related to particle size and mineralogy. *Canadian Journal of Soil Science* 64: 99-106.

Alva AK, Prakash O, Paramasivam S. 1998. Flue-gas desulfurization gypsum effects on leaching of magnesium and potassium from a Candler fine sand. *Communications in Soil Science and Plant Analysis* 29: 459-466.

Anderson DL. 1991. Soil and leaf nutrient interactions following application of calcium silicate slag to sugarcane. *Nutrient Cycling in Agroecosystems* 30: 9-18.

Anderson DL, Bowen JE. 1990. *Sugarcane Nutrition*. Atlanta, Georgia: PPI, PPIC and FAR. 39 pp.

Anonymous. 1983. Conservation of Agricultural Resources Act, 1983 (Act 43 of 1983). South African Government.

Arkcoll DB, Goulding WT, Hughes JC. 1985. Traces of 2:1 layer-silicate clays in Oxisols from Brazil, and their significance for potassium nutrition. *Journal of Soil Science* 36: 123-128.

Ashraf M, Rahmatullah Ahmad R, Afzal M, Tahir MA, Kanwal S, Maqsood MA. 2009. Potassium and silicon improve yield and juice quality in sugarcane (*Saccharum officinarum* L.) under salt stress. *Journal of Agronomy and Crop Science* 195: 284-291.

Ashraf M, Rahmatullah, Ahmad R, Bhatti AS, Afzal M, Sarwar A, Maqsood MA, Kanwal S. 2010. Amelioration of salt stress in sugarcane (*Saccharum officinarum* L.) by supplying potassium and silicon in hydroponics. *Pedosphere* 20: 153-162.

Askegaard M, Eriksen J, Johnston AE. 2004. Sustainable Management of Potassium. Chapter 6 In: Schjønning, P., Elmholt, S. and Christensen, B.T. (Eds). *Managing Soil Quality: Challenges in Modern Agriculture.* Wallingford: CABI Publishing. 344 pp.

Ayres AS. 1966. Calcium silicate slag as a growth stimulant for sugarcane on low-silicon soils. *Soil Science* 101: 216-227.

Bauw P de, Asten P van, Jassogne L, Merckx R. 2016. Soil fertility gradients and production constraints for coffee and banana on volcanic mountain slopes in the East African Rift: a case study of Mt. Elgon. *Agriculture, Ecosystems and Environment* 231: 166-175.

Beater BE. 1955. A new pattern soil sampler. *Proceedings of the South African Sugar Technologists' Association* 29: 113-114.

Bell MJ, Garside AL, Moody PW, Pankhurst C, Halpin NV, Berthelsen J. 2002. Nutrient dynamics and root health in sugarcane soils. *Proceedings of the Australian Society of Sugar Cane Technologists* 24: 92-98.

Berry SD, Spaull VW, Ramouthar PV, Cadet P. 2011. Non-uptake of silicon and variable nematode species relationships between different levels of this element in sugarcane. *South African Journal of Plant and Soil* 28: 110-118.

Berthelsen S, Hurney A, Noble AD, Rudd A, Garside AL, Henderson A. 2001a. An assessment of current silicon status of sugar cane production soils from Tully to Mossman. *Proceedings of the Australian Society of Sugar Cane Technologists* 23: 289-296.

Berthelsen S, Hurney A, Kingston G, Rudd A, Garside AL, Noble AD. 2001b. Plant cane responses to silicated products in the Mossman, Innisfail and Bundaberg districts. *Proceedings of the Australian Society of Sugar Cane Technologists* 23: 297-303.

Berthelsen S, Korndörfer GH. 2005. Methods for Si analysis in plant, soil and fertilizers. *Proceedings of the Third International Conference on Silicon in Agriculture*, Uberlândia, 22-26 October, Federal University of Uberlândia: 86–93.

Berthelsen S, Noble AD, Kingston G, Hurney A, Rudd A, Garside AL. 2003. Improving yield and CCS in sugarcane through the application of silicon based amendments. *CSIRO Land and Water Final Report*: Sugar Research and Development Corporation Project CLW009.

Blackburn F. 1984. Sugar-cane. New York: Longman. 414 pp.

Blum J, Caires EF, Ayub RA, Fonseca AF da, Sozim M, Fauate M. 2011. Soil chemical attributes and grape yield as affected by gypsum application in southern Brazil. *Communications in Soil Science and Plant Analysis* 42: 1434-1446.

Bowen JE. 1969. Absorption of copper, zinc and manganese by sugar cane tissue. *Plant Physiology* 44: 255-261.

Bühmann C, Fey MV, de Villiers JM. 1985. Aspects of the X-ray identification of swelling clay minerals in soils and sediments. *South African Journal of Science* 81: 505-509.

Cade-Menun BJ, He ZQ, Zhang HL, Endale DM, Schomberg HH, Liu CW. 2015. Stratification of phosphorus forms from long-term conservation tillage and poultry litter application. *Soil Science Society of America Journal* 79: 504-516.

Cadet P, McFarlane SA, Meyer JH. 2003. Association between nutrients and rust in sugarcane in KwaZulu-Natal. *Proceedings of the South African Sugar Technologists' Association* 77: 223-229.

Cakmak I, Hengeler C, Marschner H. 1994. Partitioning of shoot and root dry matter and carbohydrates in bean plants suffering from phosphorus, potassium and magnesium deficiency. *Journal of Experimental Botany* 45: 1245-1250.

Calcino D, Schroeder B, Panitz J, Hurney A, Skocaj D, Wood A, Salter B. 2018. *Australian Sugarcane Nutrition Manual*. Indooroopilly: Sugar Research Australia.

Carnegie AJM. 1981. Combating *Eldana saccharina* Walker: A progress report. *Proceedings of the South African Sugar Technologists' Association* 55: 116-119.

Cavalcante VS, Prado RdeM, de Almeida HJr, Cruz FJR, dos Santos DMM. 2015. Gaseous exchanges, growth and foliar anatomy of sugarcane plants grown in potassium (K) deprived nutrient solution. *Australian Journal of Crop Science* 9: 577-584.

Chapman LS. 1994. Fertiliser N management in Australia. *Proceedings of the Australian Society of Sugar Cane Technologists* 16: 83-92.

Clements HF. 1965. Effects of silicate on the growth and leaf freckle of sugarcane in Hawaii. *Proceedings of the International Society of Sugar Cane Technologists* 12: 197-215.

Cornelis J-T, Delvaux B, Georg R B, Lucas Y, Ranger J, Opfergelt S. 2011. Tracing the origin of dissolved silicon transferred from various soil–plant systems towards rivers: a review. *Biogeosciences* 8: 89–112.

Côté-Beaulieu C, Chain F, Menzies JG, Kinrade SD, Bélanger RR. 2009. Absorption of aqueous inorganic and organic silicon compounds by wheat and their effect on growth and powdery mildew control. *Environmental and Experimental Botany* 65: 155–161.

Cuong TX, Ullah H, Datta A, Hanh TC. 2017. Effects of silicon-based fertilizer on growth, yield and nutrient uptake of rice in tropical zone of Vietnam. *Rice Science* 24: 283-290

Curtin D, Smillie GW. 1983. Soil solution composition as affected by liming and incubation. *Soil Science Society of America Journal* 47: 701-707.

DAFF. 2016. South African fertilizers market analysis report. Arcadia: Department of Africulture, Forestry and Fishing, 34 pp.

Daliparthy J, Barker AV, Mondal SS. 1994. Potassium fractions with other nutrients in crops: a review focusing on the tropics. *Journal of Plant Nutrition* 17: 1859-1886.

Davis AM, Tink M, Rohde K, Brodie JE. 2016. Urea contributions to dissolved 'organic' nitrogen losses from intensive, fertilised agriculture. *Agriculture, Ecosystems & Environment* 223: 190-196.

Day PR. 1965. Particle Fractionation and Particle-Size Analysis. In: Black CA, Evans DD, White JL, Ensminger LE, Clark FE, Dinauer RC (eds), *Methods of Soil Analysis Part 1: Physical and Mineralogical Properties, including statistics of measurements and sampling.* Madison, WI: American Society of Agronomy. pp 545–567.

De Beaucorps G. 1980. The potassium requirements of crops harvested green, with special reference to grassland. In: Potassium Requirements of Crops. *IPI Research Topic No. 7*. Worblaufen-Bern: International Potash Institute. pp 105-122.

De Camargo MS, Korndörfer GH, Wyler P. 2014. Silicate fertilization of sugarcane cultivated in tropical soils. *Field Crops Research* 167: 64–75.

De Camargo MS, Pereira HS, Korndörfer GH, Queiroz AA, Borges dos Reis C. 2007. Soil reaction and absorption of silicon by rice. *Scientia Agricola* (Piracicaba, Brazil) 64: 176-180.

De Oliveira MW, Macêdo GAR, Martins JA, da Silva VSG, de Oliveira AB. 2017. Mineral nutrition and fertilization of sugarcane. *IntechOpen*. Available at http://dx.doi.org/10.5772/intechopen.72300.

Desjardins J. 2014. The world's most valuable cash crop. http://www.visualcapitalist.com/the-worlds-most-valuable-cash-crop/. Accessed 31 January 2020.

Dibb DW, Thompson Jr WR. 1985. Interaction of potassium with other nutrients. In: Munson RD (ed), *Potassium in Agriculture*. Madison: ASA/CSSA/SSSA. pp.515-534.

Donaldson RA, Meyer JH, Wood RA. 1990. Response to potassium by sugarcane grown on base saturated clay soils in the Eastern Transvaal Lowveld. *Proceedings of the South African Sugar Technologists' Association* 64: 17-21.

Dordas C. 2008. Role of nutrients in controlling plant diseases in sustainable agriculture: a review. *Agronomy for Sustainable Development* 28: 33-46.

Du Preez P. 1970. The effect of silica on cane growth. *Proceedings of the South African Sugar Technologists' Association* 44: 183-188.

Du Toit DL. 1957. Fertilizer responses as revealed by the 3 x 3 x 3 fertilizer exploratory trials. *Proceedings of the South African Sugar Technologists' Association* 31: 129-137.

Du Toit JL. 1959. Plant Nutrition. *South African Sugar Association Experiment Station Annual Report, 1958-59.* Mount Edgecombe: South African Sugar Association Experiment Station. p 6-11.

Du Toit JL. 1960. Plant Nutrition. *South African Sugar Association Experiment Station Annual Report 1959-60.* Mount Edgecombe: South African Sugar Association Experiment Station.

Elephant DE, Miles N. 2016. Prediction of the potassium requirement factor for soils of the South African sugar industry. *Proceedings of the South African Sugar Technologists' Association* 89: 262-265.

Elephant DE, Miles N, Muchaonyerwa P. 2018. Potassium dynamics in the soil-plant system and implications for fertiliser recommendations. *Proceedings of the South African Sugar Technologists' Association* 91: 134-138.

Eneji AE, Inanaga S, Muranaka S, Li J, Hattori T, An P, Tsuji W. 2008. Growth and nutrient use in four grasses under drought stress as mediated by silicon fertilizers. *Journal of Plant Nutrition* 31: 355-365.

Epstein E. 1999. Silicon. *Annual Review of Plant Physiology and Plant Molecular Biology* 50: 641-664.

Ernani PR, Miquelluti DJ, Fontoura SMV, Kaminski J, Almeida JA. 2006. Downward movement of soil cations in highly weathered soils caused by addition of gypsum. *Communications in Soil Science and Plant Analysis* 37: 571-586.

Evans H. 1936. The root-system of the sugar-cane: II. Some typical root-systems. *Empire Journal of Experimental Agriculture* 4: 208–221.

Evans H. 1965. Tissue diagnostic analyses and their interpretation in sugarcane. *Proceedings of the International Society of Sugar Cane Technologists* 12: 156-180.

FAO. 2019. World fertilizer trends and outlook to 2022. Rome: Food and Agriculture Organization of the United Nations, 40 pp.

Farina MPW, Channon P. 1991. A field comparison of lime requirement indices for maize. *Plant and Soil* 134: 127-135.

Fernández FG, Farmaha BS, Nafziger ED. 2012. Soil fertility status of soils in Illinois *Communications in Soil Science and Plant Analysis* 43: 2897-2914.

Fessehazion MK, Abraha AB, Everson CS, Truter WF, Annandale JG, Moodley M. 2012. Water use and nitrogen application for irrigation management of annual ryegrass and kikuyu pasture production. *Water Research Commission Report No. TT 520/12.* March 2012.

Fox RL, Silva JA, Younge OR, Plucknett DL, Sherman GD. 1967. Soil and plant silicon and silicate response by sugar cane. *Proceedings of the Soil Science Society of America* 6: 775-779.

Franzluebbers AJ. 2002. Soil organic matter stratification ratio as an indicator of soil quality. *Soil and Tillage Research* 66: 95-106.

Gaines TP, Mitchell GA. 1979. Boron determination in plant tissue by the azomethine H method. *Communications in Soil Science and Plant Analysis* 10: 1099-1108.

Gale MR, Grigal DF. 1987. Vertical root distributions of northern tree species in relation to successional status. *Canadian Journal of Forestry Research* 17: 829–834.

Gao X, Zou C, Wang L, Zhang F. 2004. Silicon improves water use efficiency in maize plants. *Journal of Plant Nutrition* 27: 1457-1570.

Garbuio FJ, Jones DL, Alleoni LRF, Murphy DV, Caires EF. 2011. Carbon and nitrogen dynamics in an Oxisol as affected by liming and crop residues under no-till. *Soil Science Society of America Journal* 75: 1723-1730.

Garcia M, Daverede C, Gallego P, Toumi M. 1999. Effect of various potassiumcalcium ratios on cation nutrition of grape grown hydroponically. *Journal of Plant Nutrition* 22: 417-425.

Gascho GJ, Andreis HJ. 1974. Sugarcane response to calcium silicate slag applied to organic and sand soils. *Proceedings of the International Society of Sugar Cane Technologists* 15: 543-551.

Gething PA. 1993. Improving returns from nitrogen fertilizer: the potassium-nitrogen partnership. *IPI Research Topic No. 13*. Basel: International Potash Institute.

Goebel FR, Way MJ, Gossard C. 2005. The status of *Eldana saccharina* (Lepidoptera: Pyralidae) in the South African sugar industry based on regular survey data. *Proceedings of the South African Sugar Technologists' Association* 79: 337-346.

Goli-Kalanpa E, Roozitalab MH, Malakouti MJ. 2008. Potassium availability as related to clay mineralogy and rates of potassium application. *Communications in Soil Science and Plant Analysis* 39: 2721-2733.

Gong H, Zhu X, Chen K, Wang S, Zhang C. 2005. Silicon alleviates oxidative damage of wheat plants in pots under drought. *Plant Science* 169: 313-321.

Gonzalez E. 1976. Effect of depth of lime incorporation on the growth of corn in oxisols of central Brazil. *Thesis for degree of Ph.D.* Raleigh: North Carolina State University. 125pp.

Gosnell JM, Long AC. 1971. Some factors affecting foliar analysis in sugarcane. *Proceedings of the South African Sugar Technologists' Association* 45: 217-232.

Goussain MM, Moraes JC, Carvalho JG, Nogueira NL, Rossi ML. 2002. Effect of silicon application on corn plants upon the biological development of the Fall Armyworm *Spodoptera frugiperda* (J.E. Smith) (Lepidoptera: Noctuidae). *Neotropical Entomology* 32: 305-310.

Grunes DL, Huang JW, Smith FW, Joo PK, Hewes DA. 1992. Potassium effects on minerals and organic acids in three cool-season grasses. *Journal of Plant Nutrition* 15: 1007-1025.

Gu B, Schulz RK. 1991. Anion retention in soil: Possible application to reduce migration of buried Technetium and Iodine. *NUREG/CR-5464*. Washington, DC: Division of Regulatory Applications, US Nuclear Regulatory Commission.

Guntzer F, Keller C, Meunier JD. 2012. Benefits of plant silicon for crops: a review. *Agronomy for Sustainable Development* 32: 201–213.

Halais P. 1962. The detection of NPK deficiency trends in sugarcane crops by means of foliar diagnosis run from year to year on a follow-up basis. *Proceedings of the International Society of Sugar Cane Technologists* 11: 214-221.

Hartt CE. 1934. Some effects of potassium upon the growth of sugarcane and upon the absorption and migration of ash constituents. *Plant Physiology* 9: 399-452.

Hattori T, Inanaga S, Araki H, An P, Morita S, Luxová M, Lux A. 2005. Application of silicon enhanced drought tolerance in *Sorghum bicolor*. *Physiologia Plantarum* 123: 459-466.

Hattori T, Ishii K, An P, Inanaga S. 2009. Growth enhancement of rye by silicon application under two different soil moisture regimes. *Journal of Plant Nutrition* 32: 187-196.

Haynes RJ. 2014. A contemporary overview of the availability and crop use of soil silicon. *Journal of Plant Nutrition and Soil Science* 177: 831-844.

Haynes RJ. 2017. The nature of biogenic Si and its potential role in Si supply in agricultural soils. *Agriculture, Ecosystems and Environment* 25: 100-111.

Haynes RJ, Belyaeva ON, Kingston G. 2013. Evaluation of industrial wastes as sources of fertilizer silicon using chemical extractions and plant uptake. *Journal of Plant Nutrition and Soil Science* 176: 238–248.

Haysom MB. 1971. The estimation of potassium availability in Mackay soils. *Proceedings of the Queensland Society of Sugar Cane Technologists* 38: 113-119.

Hossner LR, Doll EC. 1970. Magnesium fertilization of potatoes as related to liming and potassium. *Soil Science Society of America Proceedings* 34: 772-774.

Hsieh CF, Tsai TR, Huang HC, Lin CY. 1983. Effects of rice hull, lime, and nitrogen fertilizer on the silica content and growth of rice in a strongly acid sand-shale alluvial soil. *Contribution number 004 from Taichung District Agricultural Improvement Station:* 108-120.

Huang JY, Yu HL, Zhang SX. 2009a. Effects of potassium addition and water supply on xylem embolism in *Acer truncatum* and *Ligustrum lucidum*. *Journal of Plant Ecology* 33: 1199-1207.

Huang H, Bokhtiar SM, Xu L, Li Y, Yang L. 2009b. Effect of silicon fertilization on yield and photosynthetic attributes in Sugarcane (*Saccharum officinarum* L. hybrid). *Guangxi Agricultural Sciences* 40: 1564-1569.

Huber DM, Arny DC. 1985. Interactions of potassium with plant disease. In: Munson RD (ed) *Potassium in Agriculture*. Madison: ASA, CSSA, SSA. pp. 467-488.

Ihsan MZ, Shahzad N, Kanwal S, Naeem M, Khaliq A, El-Nakhlawy FS, Matloob A. 2013. Potassium as foliar supplementation mitigates moisture induced stresses in Mung bean (*Vigna radiata* L.) as revealed by growth, photosynthesis, gas exchange capacity and Zn analysis of shoot. *International Journal of Agronomy and Plant Production* 4: 3828-3835.

Inoue K, Yamane I, Kaji T. 2009. Effects of nitrogen topdressing and number of tillers at maximum tillering stage on the yield and extract quality of ratoon sugarcane cultivar Ni17. *Japanese Journal of Soil Science and Plant Nutrition* 80: 1-6.

IUSS Working Group WRB. 2014. World Reference Base for Soil Resources 2014. International soil classification system for naming soils and creating legends for soil maps. *World Soil Resources Reports No. 106.* Rome: FAO.

Jakobsen ST. 1993. Interaction between plant nutrients: III. Antagonism between potassium, magnesium and calcium. *Acta Agriculturae Scandinavica, Section B — Soil & Plant Science* 43: 1-5.

James DW, Weaver WH, Reeder RL. 1970. Chloride uptake by potatoes and the effects of potassium chloride, nitrogen and phosphorus fertilization. *Soil Science* 109: 48-53.

Jaskulska I, Jaskulski D, Piekarczyk M, Kotwica K, Gałęzewski L, Wasilewski P. 2015. Magnesium content in the leaves of winter wheat in a long-term fertilization experiment. *Plant, Soil and Environment* 61: 208-212.

Johnston AE. 1986. Potassium fertilization to maintain a K-balance under various farming systems. pp. 199–227. In: *Proc. of the 13th IPI congress on "Nutrient Balance and the Need of Potassium"*, August 1986, Reims, France.

Johnston AE, Barraclough PB, Poulton PR, Dawson CJ. 1998. Assessment of some spatially variable factors limiting crop yield. Proceedings No. 419, *The International Fertiliser Society*, York, UK.

Johnston MA, Miles N, Thibaud GR, Hughes JC. 1999. Quantities of potassium fertilizer required to raise soil test value. *Communications in Soil Science and Plant Analysis* 30: 2485-2497.

Jones LHP, Handreck KA. 1963. Effects of iron and aluminium oxides on silica in solution in soils. *Nature* 198: 852-853.

Jones LHP, Handreck KA. 1967. Silica in soils, plants, and animals. *Advances in Agronomy* 19: 107-149.

Karlen DL, Ellis RJ, Whitney DA, Grunes DL. 1978. Influence of soil moisture and plant cultivar on cation uptake by wheat with respect to grass tetany. *Agronomy Journal* 70: 918-921.

Kee Kwong KFNg, Gauthier J, Deville J. 1990. Foliar diagnosis of sugarcane – variation of leaf potassium value with age of cane and rainfall regime. *Proceedings of the South African Sugar Technologists' Association* 64: 22-25.

Keeping MG. 2017. Uptake of silicon by sugarcane from applied sources may not reflect plant-available soil silicon and total silicon content of sources. *Frontiers in Plant Science* 8: Article 760.

Keeping MG, Kvedaras OL, Bruton AG. 2009. Epidermal silicon in sugarcane: cultivar differences and role in resistance to sugarcane borer *Eldana saccharina*. *Environmental and Experimental Botany* 66: 54-60.

Keeping MG, Meyer JH. 2002. Calcium silicate enhances resistance of sugarcane to the African stalk borer *Eldana saccharina* Walker (Lepidoptera: Pyralidae). *Agricultural and Forest Entomology* 4: 265-274.

Keeping MG, Meyer JH, Sewpersad C. 2013. Soil silicon amendments increase resistance of sugarcane to stalk borer *Eldana saccharina* Walker (Lepidoptera: Pyralidae) under field conditions. *Plant and Soil* 363: 297-318.

Keeping MG, Miles N, Rutherford RS. 2017. Liming an acid soil treated with diverse silicon sources: Effects on silicon uptake by sugarcane (*Saccharum* spp. hybrids), *Journal of Plant Nutrition* 40: 1417-1436.

Keeping MG, Miles N, Sewpersad C. 2014. Silicon reduces impact of plant nitrogen in promoting stalk borer (*Eldana saccharina*) but not sugarcane thrips (*Fulmekiola serrata*) infestations in sugarcane. *Frontiers in Plant Science* 5: 1-12.

Kidder G, Gascho GJ. 1977. Silicate slag recommended for specified conditions in Florida. *Agronomy Facts*, No. 65, Florida Cooperative Extension. Service, University of Florida.

Kingston G. 1982. Ash in first expressed cane juice at Rocky Point – 1. Factors affecting the inorganic composition of juice. *Proceedings of the Australian Society of Sugar Cane Technologists* 52: 11-17.

Kingston G. 2000. Nutrition of sugarcane: concepts for improved nutrient management and opportunities for intervention at molecular level. Piracicaba, Brazil: *International Symposium on Physiology of Sugarcane,* Sociedad dos Técnicos Açucareiros e Alcooleiros do Brasil (*STAB*), October, 2000.

Knobeloch L, Salna B, Hogan A, Postle J, Anderson H. 2000. Blue babies and nitrate-contaminated well water. *Environmental Health Perspectives* 108: 675-678.

Korndörfer GH, Lepsch I. 2001. Effect of silicon on plant growth and crop yield. In: Datnoff LE, Snyder GH, Korndörfer GH. (eds) *Silicon in Agriculture*. Amsterdam: Elsevier. 403 pp.

Korndörfer AP, Grisoto E, Vendramim JD. 2011. Induction of insect plant resistance to the spittlebug *Mahanarva fimbriolata* Stål (Hemiptera: Cercopidae) in sugarcane by silicon application. *Neotropical Entomology* 40: 387-392.

Kraska JE, Breitenbeck GA. 2010. Survey of the silicon status of flooded rice in Louisiana. *Agronomy Journal* 102: 523-529.

Krauss A. 2001. Global and regional potash consumption and deriving K balance in agriculture. *RICP-CISTA-IPI-IMPHOS Workshop on Balanced fertilization for crop yield and quality*, Prague, 17-19 September 2001.

Kuchenbuch R, Claassen N, Jungk A. 1986. Potassium availability in relation to soil moisture. *Plant and Soil* 95: 233-243.

Kumara BH, Yogendra ND, Prakash NB, Kumar A. 2016. Effect of calcium silicate and need based nitrogen on pest management in aerobic rice (*Oryza sativa* L.). *International Journal of Plant Protection* 9: 133-136.

Kvedaras OL, Keeping MG. 2007. Silicon impedes stalk penetration by the borer *Eldana saccharina* in sugarcane. *Entomologia Experimentalis et Applicata* 125: 103-110.

Kvedaras OL, Keeping MG, Goebel FR, Byrne MJ. 2007. Larval performance of the pyralid borer *Eldana saccharina* Walker and stalk damage in sugarcane: Influence of plant silicon, cultivar and feeding site. *International Journal of Pest Management* 53: 183-194.

Laing MD, Gatarayiha MC, Adandonon A. 2006. Silicon use for pest control in agriculture: a review. *Proceedings of the South African Sugar Technologists' Association* 80: 278-286.

Lalitha M, Dhakshinamoorthy M. 2014. Forms of soil potassium – a review. *Agriculture Reviews* 35: 64-68

Lamari L. 2008. Assess 2.0 Image Analysis Software for Plant Disease Quantification. St. Paul: APS Press.

Le Blond J, Horwell CJ, Williamson BJ, Oppenheimer C. 2010. Generation of crystalline silica from sugarcane burning. *Journal of Environmental Monitoring* 12:1459-1470.

Lecler NL, Tweddle PB. 2010. Double profits with a controlled traffic zero-till irrigation farming system? *Proceedings of the South African Sugar Technologists' Association* 83: 46–62.

Lemire E, Taillon KM, Hendershot WH. 2006. Using pH-dependent CEC to determine lime requirement. *Canadian Journal of Soil Science* 86: 133–139.

Løes AK, Øgaard AF. 2003. Concentrations of soil potassium after long-term organic dairy production. *International Journal of Agricultural Sustainability* 1: 14-29.

Lyngstad I. 1992. Effect of liming on mineralization of soil nitrogen as measured by plant uptake and nitrogen released during incubation. *Plant and Soil* 144: 247-253.

Ma JF, Takahashi E. 1993. Interaction between calcium and silicon in water-cultured rice plants. *Plant and Soil* 148: 107-113.

Ma JF, Miyake Y, Takahashi E. 2001. Silicon as a beneficial element for crop plants. In: Datnoff LE, Snyder GH, Korndörfer GH. (eds). *Silicon in Agriculture*. Amsterdam: Elsevier. 403 pp.

Madaras M, Lipavský J. 2009. Interannual dynamics of available potassium in a long-term fertilization experiment. *Plant, Soil and Environment* 55: 334-343.

Manson AD. 1995. The response of Italian ryegrass to sodium, lime and potassium on an acidic Natal soil. *South African Journal of Plant and Soil* 12: 117-123.

Marschner H. 1995. *Mineral Nutrition of Higher Plants*. New York: Academic Press. 889 pp.

Marschner H, Kirkby EA, Cakmak I. 1996. Effect of mineral nutritional status on shoot-root partitioning of photoassimilates and cycling of mineral nutrients. *Journal of Experimental Botany* 47: 1255-1263

Marshall CGA, Von Brunn V. 1999. The stratigraphy and origin of the Natal Group. *South African Journal of Geology* 102: 15-25.

Martin JC, Overstreet R, Hoagland DR. 1946. Potassium fixation in soils in replaceable and nonreplaceable forms in relation to chemical reactions in the soil. *Soil Science Society of America Proceedings* 10: 94-101.

Massey FP, Hartley SE. 2009. Physical defences wear you down: progressive and irreversible impacts of silica on insect herbivores. *Journal of Animal Ecology* 78: 281-291.

McDonnell RP, Staines MvH, Bolland MDA. 2018. Determining the critical plant test potassium concentration for annual and Italian ryegrass on dairy pastures in south-western Australia. *Grass and Forage Science* 73: 112-122

McFarlane K, Maher GW. 1993. Assessment of sugarcane farms in terms of the Conservation of Agricultural Resources Act 1983. *Proceedings of the South African Sugar Technologists' Association* 67: 110-113.

McKeague JA, Cline MG. 1963. Silica in soils. Advances in Agronomy 15: 339-396.

McKenzie RH, Pauly D. 2013. Potassium fertilizer application in crop production. Alberta Ag-Info Centre. http://www1.agric.gov.ab.ca/\$department/deptdocs.nsf/ all/agdex917. Accessed 2 December 2017.

McLean EO, Watson ME. 1985. Soil measurements of plant-available potassium. In: Munson RD. (ed). *Potassium in Agriculture*. Madison, Wisconsin: ASA-CSSA-SSSA.

Mecfel J, Hinke S, Goedel WA, Marx G, Fehlhaber R, Bäucker E, Wienhaus O. 2007. Effect of silicon fertilizers on silicon accumulation in wheat. *Journal of Plant Nutrition and Soil Science* 170: 769-772.

Melis M, Farina MPW. 1984. Potassium effects on stalk strength, premature death and lodging of maize (*Zea mays* L.). *South African Journal of Plant and Soil* 1: 122-124.

Mengel K. 1982. Factors and processes affecting potassium requirements of crops. *Potash Review No. 9.* Basel, Switzerland: International Potash Institute. 12 pp.

Mengel K, Busch R. 1982. The importance of potassium buffer power on the critical potassium levels in soils. *Soil Science* 133: 27-32.

Mengel K, Kirkby EA. 2001. *Principles of Plant Nutrition*. Dordrecht, The Netherlands: Kluwer Academic Publishers. 849 pp.

Meyer JH. 2013. Sugarcane nutrition and fertilization. In: Meyer JH, Rein P, Turner P, Mathias K. (eds). *Good management practices for the cane sugar industry*. Berlin: Verlag Dr. Albert Bartens KG.

Meyer JH, Keeping MG. 2000. Review of research into the role of silicon for sugarcane production. *Proceedings of the South African Sugar Technologists' Association* 74: 29-40.

Meyer JH, Keeping MG. 2005. The impact of nitrogen and silicon nutrition on the resistance of sugarcane varieties to *Eldana saccharina* (Lepidoptera: Pyralidae). *Proceedings of the South African Sugar Technologists' Association* 79: 363-367.

Meyer JH, St John Clowes M. 2013. Sugarcane and its environment. In: Meyer JH, Rein P, Turner P, Mathias K. (eds). *Good management practices for the cane sugar industry*. Berlin: Verlag Dr. Albert Bartens KG.

Meyer JH, Schumann AW, Wood RA, Nixon DJ, van den Berg M. 2007. Recent advances to improve nitrogen use efficiency of sugarcane in the South African sugar industry. *Proceedings of the International Society of Sugar Cane Technologists* 26: 238-246.

Meyer JH, van Antwerpen R. 2010. Advances in sugarcane soil fertility research in southern Africa. *South African Journal of Plant and Soil* 27: 19-31.

Meyer JH, Wood RA. 1994. Nitrogen management of sugar cane in South Africa. *Proceedings of the Australian Society of Sugar Cane Technologists* 16: 93-104.

Meyer JH, Wood RA. 2001. The effects of soil fertility and nutrition on sugarcane quality: a review. *Proceedings of the South African Sugar Technologists' Association* 75: 242-247.

Meyer JH, Wood RA, Leibbrandt NB. 1986. Recent advances in determining the N requirement of sugarcane in the South African sugar industry. *Proceedings of the South African Sugar Technologists' Association* 60: 205-211.

Miao B-H, Han X-G, Zhang W-H. 2010. The ameliorative effect of Si on soybean seedlings grown in K-deficient medium. *Annals of Botany* 105: 967-973.

Miles N. 2010. Challenges and opportunities in leaf nutrient data interpretation. *Proceedings of the South African Sugar Technologists' Association* 83: 205-215.

Miles N. 2014. Crop nutrition guidelines based on soil analyses. Mount Edgecombe: South African Sugarcane Research Institute. 34pp. (Internal document: not available to the public to protect intellectual property).

Miles N. 2016. *Modification of potassium recommendations on the basis of nonexchangeable K reserves.* Mount Edgecombe: South African Sugarcane Research Institute. 3pp. (Internal document: not available to the public to protect intellectual property).

Miles N, Farina MPW. 2014. Towards the more efficient use of fertiliser potassium: prediction of 'slowly-available' potassium reserves in soils. *Proceedings of the South African Sugar Technologists' Association* 87: 330-333.

Miles N, Manson AD, Rhodes R, van Antwerpen R, Weigel A. 2014. Extractable silicon in soils of the South African sugar industry and relationships with crop uptake. *Communications in Soil Science and Plant Analysis* 45: 2949-2958.

Miles N, Titshall L. 2020. *Information Sheet 7.15: Sugarcane leaf sampling.* Mount Edgecombe: South African Sugarcane Research Institute. 4 pp.

Moody PW, Schroeder BL, Wood AW. 2007. Assessing soil potassium reserves in sugarcane soils. *Proceedings of the Australian Society of Sugarcane Technologists* 29: (CD-ROM), 7 pp.

Mottram R, Hutson JL, Goodman PS. 1981. Water retention by some Natal soils as related to soil texture and organic matter content. *Crop Production* 10: 47-50.

Muchow RC, Robertson MJ. 1994. Relating crop nitrogen uptake to sugarcane yield. *Proceedings of the Australian Society of Sugar Cane Technologists* 16:122-130.

Muchow RC, Robertson MJ, Wood AW, Keating BA. 1996. Research effect of nitrogen on the time-course of sucrose accumulation in sugarcane. *Field Crops Research* 47: 143-153.

Munsamy SS. 2013. Investigation into the high ash content in molasses at Nakambala, Zambia. *Proceedings of the South African Sugar Technologists' Association* 86: 67–73.

Nachtigall GR, Nogueirol RC, Alleoni LRF, Cambri MA. 2007. Copper concentration of vineyard soils as a function of pH variation and addition of poultry litter. *Brazilian Archives of Biology and Technology* 50: 941-948.

Nassar AM, El-Sagheir KS, Ramadan BSH. 2005. Effect of nitrogen levels on yield and juice quality of some sugar cane varieties (*Saccharum* spp., L.). *Egyptian Journal of Agricultural Research* 83: 681-692.

Nuernberg NJ, Leal JE, Sumner ME. 1998. Evaluation of an Anion-Exchange Membrane for extracting plant available phosphorus in soils. *Communications in Soil Science and Plant Analysis* 29: 467-479.

Nukaya A, Voogt W, Sonneveld C. 1991. Effect of nitrate, sulphate and chloride anion ratios on tomatoes grown in recirculating system. *Acta Horticulturae* 294:297-304.

Obihara CH, Russell EW. 1972. Specific adsorption of silicate and phosphate by soils. *Journal of Soil Science* 23: 105-117.

Øgaard AF, Krogstad T, Lunnan T. 2002. Ability of some Norwegian soils to supply grass with potassium (K) - soil analyses as predictors of K supply from soil. *Soil Use and Management* 18: 412–420.

Omar MA, El Kobbia TE. 1966. Some observations on the interrelationships of potassium and magnesium. *Soil Science* 101: 437-440.

Pal Y, Gilkes RJ, Wong MTF. 2001. Mineralogy and potassium release from some Western Australian soils and their size fractions. *Australian Journal of Soil Research* 39: 813-822.

Pannuti LEdaR, Baldin ELL, Gava GJdeC, Silva JPGFda, Souza EdeS, Kölln OT. 2015. Interaction between N-fertilizer and water availability on borer-rot complex in sugarcane. *Bragantia* 74: 75-83.

Parker DR, Hendricks GJ, Sparks DL. 1989a. Potassium in Atlantic Coastal Plain soils: II. Crop responses and changes in soil potassium under intensive management. *Soil Science Society of America Journal* 53: 397-401.

Parker DR, Sparks DL, Hendricks GJ, Sadusky MC. 1989b. Potassium in Atlantic Coastal Plain soils: I. Soil characterization and distribution of potassium. *Soil Science Society of America Journal* 53: 392-396.

Paton-Walsh C, Wilson SR, Naylor T, Griffith DWT, Denmead OT. 2011. Transport of NO_x emissions from sugarcane fertilisation into the Great Barrier Reef Lagoon. *Environmental Modeling & Assessment* 16: 441-452.

Peech M. 1965. Hydrogen-Ion Activity. In: Black CA, Evans DD, White JL, Ensminger LE, Clark FE, Dinauer RC (eds), *Methods of Soil Analysis Part 2: Chemical and Microbiological Properties*. Madison, WI: American Society of Agronomy.

Portch S, Hunter A. 2002. A systematic approach to soil fertility evaluation and management. *Modern Agriculture and Fertilizers*. PPI-PPIC China Program, Special Publication No. 5. Beijing: China Agriculture Press. p 1-62.

Poss R, Fardeau JC, Saragoni H. 1997. Sustainable agriculture in the tropics: the case of potassium under maize cropping in Togo. *Nutrient Cycling in Agroecosystems* 46: 205-213.

Queiroz RJB, dos Santos DMM, Ferraudo AS, Carlin SD, Silva MdeA. 2011. Biochemical and physiological responses of sugarcane cultivars to soil water deficiencies. *Scientia Agricola* 68: 469-476.

Ranganathan S, Suvarchala V, Rajesh YBRD, Srinivasa Prasad M, Padmakumari AP, Rao VS. 2006. Effects of silicon sources on its deposition, chlorophyll content, and disease and pest resistance in rice. *Biologia Plantarium* 50: 713-716.

Rao CS, Khera, MS. 1994. Potassium replenishment capacity of illitic soils at their minimal exchangeable K in relation to clay mineralogy. *Journal of Plant Nutrition and Soil Science* 157: 467-470.

Redman H. 1991. Responses to N fertilizer application in Dominican Republic. *Short communication paper presented at ISSCT Sugarcane Nutrition Workshop, Dominican Republic, 1991:* 12-13.

Rhodes R, Miles N, Keeping MG. 2013. Crop nutrition and soil textural effects on eldana damage in sugarcane. *Proceedings of the South African Sugar Technologists' Association* 86: 121-136.

Rich CI. 1968. Mineralogy of soil potassium. In: Kilmer VJ, Younts SE, Brady NC. (eds), *The role of potassium in agriculture*. Madison, Wisconsin: American Society of Agronomy, Crop Science Society of America, Soil Science Society of America.

Rich CI, Black WR. 1964. Potassium exchange as affected by cation size, pH, and mineral structure. *Soil Science* 97: 384-390.

Rich CI, Obenshain SS. 1955. Chemical and clay mineral properties of a red-yellow podzolic soil derived from mica schist. *Soil Science Society of America Proceedings* 19: 334-339.

Ritchey KD. 1979. Potassium fertility in oxisols and ultisols of the humid tropics. *Cornell International Agriculture Bulletin (Vol 37).* New York State College of Agriculture and Life Sciences, Cornell University, Ithaca. 45 pp.

Roberts FJ. 1968. The annual pattern of potassium release from organic matter in sandy soils of south-west Australia. *Australian Journal of Soil Research* 6: 193-202.

Rodriguez P. 1975. Calcium, magnesium, and potassium status in some soils of the Eastern Plains of Colombia. *Thesis for degree of M.S.*, Ithaca, New York: Cornell University. 177 pp.

Ross L, Nababsing P, Wong You Cheong Y. 1974. Residual effect of calcium silicate applied to sugarcane soils. *Proceedings of the International Society of Sugar Cane Technologists* 15: 539-542.

Sá JCdeM, Lal R. 2009. Stratification ratio of soil organic matter pools as an indicator of carbon sequestration in a tillage chronosequence on a Brazilian Oxisol. *Soil and Tillage Research* 103: 46-56.

Sadusky MC, Sparks DL, Noll MR, Hendricks GJ. 1987. Kinetics and mechanisms of potassium release from sandy Middle Atlantic Coastal Plain soils. *Soil Science Society of America Journal* 51: 1460-1465.

Saeroji S, Sunaryo, Gunito H. 2010. The effect of bagasse furnace ash application on sugarcane resistance to top borer *Scirpophaga nivella intacta* Snellen (Lepidoptera: Pyralidae). *Proceedings of the International Society of Sugar Cane Technologists* 27: 1-8.

Samadi A. 2011. Potassium Supplying Power of Selected Alkaline-Calcareous Soils in the North-west of Iran. *Journal of Agricultural Science and Technology* 13: 1197-1208.

Sangster AG, Hodson MJ, Tubb HJ. 2001. Silicon deposition in higher plants. In: Datnoff LE, Snyder GH, Korndörfer GH. (eds.). *Silicon in Agriculture*. Amsterdam: Elsevier. 403 pp.

Sanz Scovino JI, Rowell DL. 1988. The use of feldspars as potassium fertilizers in the savannah of Colombia. *Fertilizer Research* 17: 71-83.

SASA. 2015. South African Sugar Association Industry Overview. http://www.sasa.org.za/sugar_industry/IndustryOverview.aspx. Accessed 1 February 2015.

SASA. 2020. South African Sugar Association Industry Overview. http://www.sasa.org.za/the-sugar-industry/. Accessed 28 January 2020.

SASEX. 1999. *Identification and management of the soils of the South African Sugar Industry*. Mount Edgecombe: South African Sugar Association Experiment Station. 173 pp.

SASRI. 2004. Information Sheet 7.10: Soil Sampling. Mount Edgecombe: South African Sugarcane Research Institute. 2pp.

SASRI. 2006a. Information Sheet 13.23: Variety N37. Mount Edgecombe: South African Sugarcane Research Institute. Available at www.sasa.org.za/Libraries/Variety_Information/N37.sflb.ashx

SASRI. 2006b. Information Sheet 13.24: Variety N39. Mount Edgecombe: South African Sugarcane Research Institute. Available at www.sasa.org.za/Libraries/Variety_Information/N39.sflb.ashx

SASRI. 2013. Information Sheet 7.17: Guidelines for the interpretation of leaf analyses for sugarcane. Mount Edgecombe: South African Sugarcane Research Institute. 2 pp.

Sauer D, Saccone L, Conley DJ, Herrmann L, Sommer M. 2006. Review of methodologies for extracting plant-available and amorphous Si from soils and aquatic sediments. *Biogeochemistry* 80: 89-108.

Savant NK, Korndörfer GH, Datnoff LE, Snyder GH. 1999. Silicon nutrition and sugarcane production: a review. *Journal of Plant Nutrition* 22: 1853-1903.

Shcherbak I, Millar N, Robertson GP. 2014. Global metaanalysis of the nonlinear response of soil nitrous oxide (N_2O) emissions to fertilizer nitrogen. *Proceedings of the National Academy of Sciences* 111: 9199-9204.

Schimansky C. 1981. Die Aufnahme von ²⁸Mg, ⁸⁶Rb und ⁴⁵Ca durch Gerstenpflanzen bei untershiedlichen Wurzeltemperaturen. *Zeitschrift für Pflanzenernährung und Bodenkunde* 144: 356-365.

Schroeder BL. 2007. Soil-specific nutrient management guidelines for sugarcane production in the Bundaberg district. Bundaberg: Bureau of Sugar Experiment Stations Limited. 76 pp.

Schroeder BL, Wood AW. 2002. A re-evaluation of the basis for deriving potassium fertiliser recommendations in the Australian sugar industry. *Proceedings of the Australian Society of Sugar Cane Technologists* 24: (CD-ROM), 11 pp.

Schroeder BL, Wood RA, Meyer JH. 1993. Foliar analysis in the South African Sugar Industry for diagnostic and nutrient trend purposes. In: NJ Barrow (Ed) *Plant*

Nutrition – from Genetic Engineering to Field Practice. Dordrecht: Kluwer Academic Publishers.

Schroeder BL, Wood AW, Moody PW, Bell MJ, Garside AL. 2005. Nitrogen fertilizer guidelines in perspective. *Proceedings of the Australian Society of Sugar Cane Technologists* 27: 291-304.

Schroeder BL, Wood AW, Moody PW, Panitz JH, Agnew JR, Sluggett RJ, Salter B. 2006. Delivering nutrient management guidelines to growers in the central region of the Australian sugar industry. *Proceedings of the Australian Society of Sugarcane Technologists* 28: 142-154.

Schroeder BL, Wood AW, Moody PW, Stewart RL, Panitz JH, Benn J. 2007. Soilspecific nutrient management guidelines for sugarcane production in the Johnstone Catchment. Indooroopilly: *Technical Publication TE07001*, BSES Limited.

Schumann AW. 2000. Prospects for improving nitrogen fertiliser use efficiency with a new soil test and ammonia volatilisation model. *Proceedings of the South African Sugar Technologists' Association* 74: 70-78.

Selim HM, Mansell RS, Zelazny LW. 1976. Modeling reactions and transport of potassium in soils. *Soil Science* 122: 77-84.

Sharpley AN. 1990. Reaction of fertilizer potassium in soils of different mineralogy. *Soil Science* 149: 44-51.

Shen X, Zhou Y, Duan L, Li Z, Eneji AE, Li J. 2010. Silicon effects on photosynthesis and antioxidant parameters of soybean seedlings under drought and ultraviolet-B radiation. *Journal of Plant Physiology* 167: 1248-1252.

Silva RV, Oliveira RDL, Nascimento KJT, Rodrigues FA. 2010. Biochemical responses of coffee resistance against *Meloidogyne exigua* mediated by silicon. *Plant Pathology* 59: 586-593.

Singh US. 1973. Nitrogen and sugarcane. III. Relation between leaf nitrogen and cane yield, sucrose content and purity coefficient of juice. *Indian Sugar* 23: 323-328.

Smith D. 1975. Effects of potassium topdressing a low fertility silt loam soil on alfalfa herbage yields and composition and on soil K values. *Agronomy Journal* 67: 60-64.

Smith DM, Inman-Bamber NG, Thorburn PJ. 2005. Growth and function of the sugarcane root system. *Field Crops Research* 92: 169–183.

Smithson F. 1956. Plant opal in soil. *Nature* 178: 107–108.

Smyth TJ, Sanchez PA. 1980. Effects of lime, silicate, and phosphorus applications to an oxisol on phosphorus sorption and ion retention. *Soil Science Society of America Journal* 44: 500-505.

Soepardi G. 1991. Criteria for making potassium recommendations in Indonesia. *Short communication paper presented at ISSCT Sugarcane Nutrition Workshop, Dominican Republic,* 1991: 45-50.

Soil Classification Working Group. 1991. Soil Classification: A taxonomic system for South Africa. *Memoir on the Agricultural Natural Resources of South Africa No.15* Pretoria: Department of Agricultural Development.

Soil Survey Staff. 2014. *Keys to Soil Taxonomy* (12th edn). Washington: USDA-Natural Resources Conservation Service.

Sonobe K, Hattori T, An P, Tsuji W, Eneji AE, Kobayashi S, Kawamura Y, Tanaka K, Inanaga S. 2011. Effect of silicon application on sorghum root responses to water stress. *Journal of Plant Nutrition* 34: 71-82.

Spalding EP, Hirsch RE, Lewis DR, Qi Z, Sussman MR, Lewis BD. 1999. Potassium uptake supporting plant growth in the absence of AKT1 channel activity: inhibition by ammonium and stimulation by sodium. *Journal of General Physiology* 113: 909–918.

Sparks DL. 2000. Bioavailability of soil potassium. In: Sumner ME. (ed). *Handbook of Soil Science*. Boca Raton: CRC Press LLC.

Sparks DL. 2001. Dynamics of K in soils and their role in management of K nutrition. International Potash Institute PRII: Potassium in nutrient management for sustainable crop production. *Technical Session 1: Potassium in soils. New Delhi, India:* 79-101.

Sparks DL, Huang PM. 1985. Physical Chemistry of Soil Potassium. In: Munson R. (ed). *Potassium in Agriculture*. Madison, Wisconsin: American Society of Agronomy.

Sparks DL, Liebhardt WC. 1982. Temperature effects on potassium exchange and selectivity in Delaware soils. *Soil Science* 133: 10-17.

Stats SA. 2019. Statistical release P0441: Gross domestic product, Second quarter 2019. http://www.statssa.gov.za/?p=12798. Accessed 28 January 2020.

Stevens MH. 1970. The effects of calcium and magnesium on the uptake of potassium by red clover (*Trifolium pratense* L.) and hybrid sorghum grown on selected soils of the Red River flood plain in Louisiana. *PhD thesis*, Louisiana State University. LSU Historical Dissertations and Theses. 2091.

Stevenson DWA, van der Merwe A, Benninga W, Allison JCS. 1992. Response of different sugarcane varieties to greater than normal applications of nitrogen. *Proceedings of the South African Sugar Technologists' Association* 66: 50-53.

Steynberg RE, Nel PC, Rethman NFG. 1994. Soil water use and rooting depth of Italian ryegrass (*Lolium multiflorum* Lam.) in a small plot experiment. *South African Journal of Plant and Soil* 11: 80-83.

Stranack RA, Miles N. 2011. Nitrogen nutrition of sugarcane on an alluvial soil on the KwaZulu-Natal north coast: Effects on yield and leaf nutrient concentrations. *Proceedings of the South African Sugar Technologists' Association* 84: 198-209.

Sumner ME. 2011. Out of sight out of mind: subsurface factors in sugarcane yield decline. *Proceedings of the South African Sugar Technologists' Association* 84: 1-27.

Sumner ME, Farina MPW. 1986. Phosphorus interactions with other nutrients and lime in field cropping systems. *Advances in Soil Science* 5: 201-236.

Sumner ME, Fey MV, Noble AD. 1991. Nutrient Status and Toxicity Problems in Acid Soils. In: Ulrich B, Sumner ME (Eds). *Soil Acidity*. Berlin: Springer. p 149-182.

Tahir MA, Rahmatullah Aziz T, Ashraf M. 2010. Wheat genotypes differed significantly in their response to silicon nutrition under salinity stress. *Journal of Plant Nutrition* 33: 1658-1671.

Tavakkoli E, English P, Guppy CN. 2011a. Interaction of Silicon and Phosphorus Mitigate Manganese Toxicity in Rice in a Highly Weathered Soil. *Communications in Soil Science and Plant Analysis* 42: 503-513.

Tavakkoli E, Lyons G, English P, Guppy CN. 2011b. Silicon nutrition of rice is affected by soil pH, weathering and silicon fertilisation. *Journal of Plant Nutrition and Soil Science* 174: 437-446.

Thomas J, Jennings H. 2014. The science of concrete. http://www.iti. northwestern.edu/cement/monograph/Monograph3_7.html. Accessed 2 February 2018

Thompson GD. 1988. The composition of plant and ratoon crops of variety N14 at Pongola. *Proceedings of the South African Sugar Technologists' Association* 62: 185-189.

Thornley JHM. 1972. A balanced quantitative model for root: shoot ratios in vegetative plants. *Annals of Botany* 36: 431-441.

Tigner R. 2018. Partial Budgeting: A Tool to Analyze Farm Business Changes. *Iowa State University Extension and Outreach File C1-50 (August 2018).* https://www.extension.iastate.edu/agdm/wholefarm/pdf/c1-50.pdf. Accessed 27 March 2020.

Truog E. 1930. The determination of the readily available phosphorus of soils. *Journal of the American Society of Agronomy* 22: 874-878.

Tubana BS, Babu T, Datnoff LE. 2016. A review of silicon in soils and plants and its role in US agriculture: history and future perspectives. *Soil Science* 181: 393-411.

Tweddle P. 2020. Mechanisation Report No. 1 2020. Mount Edgecombe: South African Sugarcane Research Institute. https://sasri.org.za/mechanisation/. Accessed 21 April 2020.

Valeri SV, Cruz MCPda, Vasconcelos RTde. 2016. Potassium doses for African mahogany plants growth under two hydric conditions. *African Journal of Agricultural Research* 11: 1973-1979.

Van Antwerpen R. 1998. Modelling root growth and water uptake of sugarcane cultivar NCo 376. *Ph.D. Thesis.* University of the Orange Free State, Bloemfontein, 133 pp.

Van Antwerpen R. 1999. Sugarcane root growth and relationships to above-ground biomass. *Proceedings of the South African Sugar Technologists' Association* 73: 89-95.

Van Antwerpen R, McGlinchey MG, Inman-Bamber NG, Bennie ATP. 1994. Estimating root water use in sugarcane. *Proceedings of the South African Sugar Technologists' Association* 70: 57-58.

Van der Laan M, Miles N. 2010. Nutrition of the South African sugar crop: current status and long-term trends. *Proceedings of the South African Sugar Technologists' Association* 83: 195-204.

Van der Merwe AJ, Johnson JC, Ras LSK. 1984. An $NH_4HCO_3-NH_4F-(NH_4)_2$ EDTA method for the determination of extractable P, K, Ca, Mg, Cu, Fe, Mn and Zn in soils. *SIRI Information Bulletin* B2/2, Department of Agriculture and Fisheries, South Africa.

VSN International. 2011. GenStat for Windows 14th Edition. Hemel Hempstead, UK: VSN International. GenStat.co.uk.

Voora V, Bermúdez S, Larrea C. 2019. Global Market Report: Sugar. International Institute for Sustainable Development. Sustainable Commodities Marketplace Series 2019. https://www.iisd.org/sites/default/filespublications/ssi-global-market-reportsugar.pdf. Accessed 14 April 2020.

Walkley A, Black IA. 1934. An examination of the Degtjareff Method of determining soil organic matter, and a proposed modification of the chromic acid titration method. *Soil Science* 37: 29-38.

Wang FL, Huang PM. 2001. Effects of organic matter on the rate of potassium adsorption by soils. *Canadian Journal of Soil Science* 81: 325-330.

Whittenberger RT. 1945. Silicon absorption by rye and sunflower. *American Journal of Botany* 32: 539-549.

Wilkinson SR, Grunes DL, Sumner ME. 2000. Nutrient interactions in soil and plant nutrition. In: Sumner ME (ed), *Handbook of Soil Science*. Boca Raton: CRC Press LLC. pp 89-112.

Wong You Cheong Y, Heitz A, DeVille J. 1971. The effect of silicon on enzyme activity in vitro and sucrose production in sugarcane leaves. *Proceedings of the International Society of Sugar Cane Technologists* 14: 777-785.

Wong You Cheong Y, Heitz A, DeVille J. 1973. The effect of silicon on sugar cane growth in pure nutrient solution. *Journal of the Science of Food and Agriculture* 24: 113-115.

Wood RA. 1964. Assessing the potential of sugarbelt soils to supply nitrogen for plant cane. *Proceedings of the South African Sugar Technologists' Association* 38: 176-1 79.

Wood RA. 1965. Mineralization studies on virgin and cultivated sugar belt soils. *Proceedings of the South African Sugar Technologists' Association* 39: 195-202.

Wood RA. 1968. Nitrogen fertilizer use for cane. *South African Sugar Journal* 52: 3-15.

Wood RA. 1979. The effect of lime on release and plant uptake of nitrogen from soils of the Natal midlands. *Proceedings of the South African Sugar Technologists' Association* 53: 173-176.

Wood RA, Burrows JR. 1980. Potassium availability in soils of the South African sugar belt. *Proceedings of the International Society of Sugarcane Technologists* 17: 182-195.

Wood RA, Meyer JH. 1986. Factors affecting potassium nutrition of sugarcane in South Africa. *Proceedings of the South African Sugar Technologists' Association* 60: 198-204.

Wood RA, Schroeder BL. 1991. Release of non-exchangeable potassium reserves from a range of sugar industry soils. *Proceedings of the South African Sugar Technologists' Association* 65: 47-52.

Wood AW, Schroeder BL. 2004. Potassium: a critical role in sugarcane production, particularly in drought conditions. *Proceedings of the Australian Society of Sugar Cane Technologists* 26: (CD-ROM).

Wood GH, Wood RA. 1967. The estimation of cane root development and distribution using radiophosphorus. *Proceedings of the South African Sugar Technologists' Association* 41: 160-168.

Wooldridge J. 1989. The effect of lime, KCl and parent material on the cation exchange capacity of some acid subsoils of the western Cape. *South African Journal of Plant and Soil* 6: 143-147.

Wright AL, Hons FM, Lemon RG, McFarland ML, Nichols RL. 2007. Stratification of nutrients in soil for different tillage regimes and cotton rotations. *Soil and Tillage Research* 96: 19-27.

Zhu Y, Gong H. 2014. Beneficial effects of silicon on salt and drought tolerance in plants. *Agronomy for Sustainable Development* 34: 455-472.

APPENDICES

CHAPTER 3

Appendix 3.1: Laboratory methods used for soil analysis at the South African Sugarcane Research Institute's Fertiliser Advisory Service

Determination	Method	Reference						
Clay	Hydrometer	Day (1965)						
Organic carbon	Modified Walkley-Black	Walkley and Black (1934)						
рН	1:2.5 v/v soil:0.01 <i>M</i> CaCl ₂	Peech (1965)						
^α Available phosphorus	Truog	Truog (1930)						
[€] Available phosphorus	Anion Exchange Resin	Nuernberg <i>et al</i> . (1998)						
Exchangeable potassium, calcium and magnesium	Multi-extractant Ambic-2; ICP-OES $^{\Psi}$	Van der Merwe <i>et al.</i> (1984)						
Silicon (0.01 <i>M</i> CaCl ₂ - exchangeable)	Colorimetric	Miles <i>et al</i> . (2014)						
Exchangeable acidity (Al+H)	10 min extraction with 1M KCI followed by titration with 0.1M NaOH	Farina and Channon (1991)						
$^{\alpha}$ Truog method used during and after the plant and immediately following the first ration								

crops (Oxisol); the plant crop (Inceptisol); and the pot trial (Chapter 8)

[€]Anion Exchange Resin used during and after the second ratoon crop (Oxisol) and first and second ratoon crops (Inceptisol).

 $^{\Psi}$ Inductively coupled plasma optical emission spectrometry (Varian 720ES)

Appendix 3.2: Application dates and nutrients applied following the plant crop and first ration crop harvests at the Oxisol and Inceptisol trial sites. Nutrient rates for all crops are provided in the text (Section 3.2.3)

Soil	Applied following	Nutrient applied	Application date	Time elapsed since harvest	Application method
	Plant crop harvest	N K	11 Feb 2011	2.5 months	Broadcast
Oxisol		K P	26 Jun 2012	1 month	Broadcast
	First ratoon barvest	Si (cement)	29 Aug 2012	3 months	Broadcast, raked in
	naivest	Ν	26 Sep 2012	¹ 4 months	Broadcast
	Plant crop harvest	N K	6 Oct 2011	2 months	Broadcast
Inceptisol	First ratoon	Si (cement)	13 Dec 2012	2 months	Broadcast, raked in
	harvest	N K	22 Dec 2012	2.5 months	Broadcast

¹N applied late to avoid period of poor N uptake in winter.

Appendix 3.3: Harvest date and sugarcane age at harvest for the plant, first and second ratoon crops at the Oxisol and Inceptisol trial sites

Soil	Plant crop	First ratoon	Second ratoon					
	Age at harvest; harvest dates							
Oxisol	12 months;	18 months;	15 months;					
	29 Nov – 2 Dec 2010	21-25 May 2012	2-4 Sep 2013					
Inceptisol	20 months;	14 months;	19 months;					
	1-3 Aug 2011	1-4 Oct 2012	29-30 April 2014					

CHAPTER 4

Appendix	4.1:	Effect	of	applied	potassium	(K)	on	mean	(n=24)	third-leaf	K,	at	4-	and
6-months'	crop a	age, on	the	e Oxisol	and the Ince	eptis	ol							

K applied		ł	(
K applied	(%)								
(kg ha⁻¹)	Ox	isol	Ince	ptisol					
	^a PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months					
0	1.611 ± 0.024 a [∓]	1.332 ± 0.026 a	1.109 ± 0.021 a	nd [#]					
100	1.667 ± 0.032 a	1.414 ± 0.023 b	1.270 ± 0.023 b	nd					
200	1.748 ± 0.024 b	1.427 ± 0.022 b	1.356 ± 0.019 c						
300	1.753 ± 0.022 b	1.464 ± 0.025 b	1.397 ± 0.025 c	nd					
Sig.	P < 0.001	P < 0.001	P < 0.001	nd					
	^b R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months					
0	0.994 ± 0.016 a	1.046 ± 0.013 a	1.014 ± 0.016 a	1.304 ± 0.017 a					
100	1.078 ± 0.016 b	1.147 ± 0.012 b	1.142 ± 0.018 b	1.358 ± 0.012 b					
200	1.089 ± 0.014 b	1.199 ± 0.011 c	1.207 ± 0.011 c	1.375 ± 0.015 b					
300	1.109 ± 0.016 b	.109 ± 0.016 b 1.225 ± 0.012 c 1.263 ± 0.014		1.397 ± 0.011 b					
Sig.	P < 0.001	P < 0.001	P < 0.001	P < 0.001					
	°R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months					
0	1.012 ± 0.024 a	1.198 ± 0.027 a	1.083 ± 0.122 a	1.181 ± 0.010 a					
100	1.255 ± 0.021 b	1.454 ± 0.025 b	1.307 ± 0.044 b	1.361 ± 0.010 b					
200	1.340 ± 0.023 c	1.589 ± 0.025 c	1.397 ± 0.081 c	1.422 ± 0.013 c					
300	1.395 ± 0.026 c	1.625 ± 0.033 c	1.459 ± 0.141 c	1.454 ± 0.015 d					
Sig.	P < 0.001	P < 0.001	P < 0.001	P < 0.001					

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

[#]nd = not determined.

⁺Within sampling events, values with different letters are significantly different to each other.

N applied	N applied N										
		(%)									
(kg ha⁻¹)	Oxi	sol	Incer	otisol							
	^a PC - 4 months	PC - 6 months	PC - 4 months	PC - 6 months							
0	1.743 ± 0.015 a [∓]	1.435 ± 0.016 a	2.031 ± 0.026 a	nd [#]							
[¢] 80/ ^Ψ 105	1.891 ± 0.018 b	1.572 ± 0.020 b	2.137 ± 0.024 b	nd							
[¢] 160/ ^Ψ 210	1.881 ± 0.019 b	1.622 ± 0.020 b	2.169 ± 0.018 b	nd							
Significance	P<0.001	P<0.001	P<0.001	nd							
	^b R1 - 4 months	R1 - 6 months	R1 - 4 months	R1 - 6 months							
0	1.148 ± 0.017 a	1.480 ± 0.012 a	1.364 ± 0.033 a	1.262 ± 0.018 a							
80/105	1.234 ± 0.018 b	1.561 ± 0.013 b	1.729 ± 0.039 b	1.372 ± 0.022 b							
160/210	1.262 ± 0.022 b	1.583 ± 0.010 b	1.845 ± 0.032 b	1.470 ± 0.025 c							
Significance	P<0.001	P<0.001	P<0.001	P<0.001							
	°R2 - 4 months	R2 - 6 months	R2 - 4 months	R2 - 6 months							
0	2.112 ± 0.018 a	2.081 ± 0.023 a	1.854 ± 0.018 a	1.516 ± 0.013 a							
80/105	2.234 ± 0.032 b	2.224 ± 0.019 b	2.141 ± 0.025 b	1.675 ± 0.015 b							
160/210	2.213 ± 0.015 b	2.243 ± 0.015 b	2.269 ± 0.022 c	1.759 ± 0.014 c							
Significance	P<0.001	P<0.001	P<0.001	P<0.001							

Appendix 4.2: Effect of applied nitrogen (N) on mean (n=32) third-leaf N, at 4- and 6-months' crop age, on the Oxisol and the Inceptisol

^aPC = Plant crop; ^bR1 = First Ratoon crop; ^cR2 = Second Ratoon crop.

 $^{\phi}$ = rate applied on the Oxisol; $^{\Psi}$ = rate applied on the Inceptisol

[#]nd = not determined.

⁺Within sampling events, values with different letters are significantly different to each other.



Appendix 4.3: Effect of leaf silicon (Si) concentration on tonnes estimated recoverable crystal (t ERC ha⁻¹), as a percent of the maximum for each harvest on the (A) Oxisol and (B) Inceptisol. Leaf samples were collected at 4 months' crop age. Within soil types and crops, each point represents the mean (n=48), across all replicates, at each of the two Si application rates (0 kg Si ha⁻¹: clear symbol; 300 kg Si ha⁻¹: black symbol).

CHAPTER 5

Appendix 5.1: Effect of nitrogen application on topsoil (0-20 cm) properties

Although applied N tended to have little effect on either soil, some evidence, and effects, of soil acidification could be found at both trial sites, examples of which are given in Table A5.1. Topsoil pH was significantly decreased by N application at one of the three sampling events on the Oxisol, and two on the Inceptisol. Exchangeable acidity was increased at one sampling event at each site, while acid saturation was increased at one event on the Oxisol. Increases in available Fe (at one sampling event on the Oxisol) and Mn (at one sampling event at each trial site) may be linked to the acidification of these soils by the addition of N.

Table A5.1: Oxisol: Effects of applied nitrogen (N) on mean (n=32) selected topsoil (0-20 cm) properties in the second ratoon crop six weeks after application of fertiliser. Only significant effects are shown

N applied Fe		Fe Mn		Exchangeable acidity	Acid saturation
(kg ha⁻¹)	(mg	J L⁻¹)		(cmol _c L ⁻¹)	(%)
0	247.9 a [∓]	2.4 a	4.53 a	1.82 a	16.45 a
80	259.2 ab	2.5 a	4.49 a	2.11 ab	19.09 ab
160	273.3 b	2.8 b	4.41 b	2.35 b	21.18 b
Significance	P<0.05	P<0.001	P<0.001	P<0.001	P=0.01

^{*}Within columns, values with different letters are significantly different to each other.

Appendix 5.2: Effect of nitrogen application on mean soil profile characteristics following harvest

Samples were collected and analysed for nitrate-N following the harvest of each crop. Nitrogen application significantly increased soil profile nitrate-N each time it was measured, at both trial sites, even 12-20 months after application. An example of this increase is shown in Figure A5.1. Post-harvest soil profile ammonium levels were not affected by the N rate applied at the start of each crop (results not shown).



Figure A5.1: Effect of nitrogen (N) application on mean (n=24) soil profile (Oxisol 0-100 cm; Inceptisol 0-80 cm) nitrate-N, following the plant crop harvests.

Soil profile pH was significantly reduced after two crops on the Oxisol, a finding similar to that in the topsoil samples alone (Table A5.1). Nitrogen application did not, however, affect soil profile pH on the Inceptisol, in contrast with its effect on topsoil pH (Appendix 5.1).

Application of N led to a significant increase in soil profile Si following each harvest of the ratoon crops on the Oxisol (Table A5.2). This may be linked to crop removal since application of N led to a reduction in plant Si at all six leaf sampling events on the Oxisol (Section 4.3.2.2). Reduced Si uptake by the crop would result in greater amounts of Si remaining in the soil. On the Inceptisol there were no significant differences between soil Si concentrations across the N rates on this soil.

Table	A5.2:	Oxisol:	Effect	of	applied	nitrogen	(N)	application	on	mean	±	standard	error
(n=32)	soil pr	ofile (0-	100 cm) si	ilicon (Si) following	g rat	oon crop ha	rves	sts			

N application rate (kg ha ⁻¹)	First ratoon crop	Second ratoon crop			
 0	16.7 ± 0.4 a	11.0 ± 0.4 a			
80	17.3 ± 0.5 ab	11.3 ± 0.4 ab			
160	17.7 ± 0.4 b	11.6 ± 0.4 b			
 Sig.	P<0.01	P<0.05			

⁺Within columns, values with different letters are significantly different to each other.
Nitrogen application did not have consistent effects on any of the other soil properties measured at either trial site.

Appendix 5.3: Nitrogen movement down the profile

Ammonium and nitrate showed no consistent pattern down the soil profile, following crop harvests. The Oxisol tended to have higher nitrate-N levels, with an average of 9.5 mg L^{-1} over all N rates at 0-80 cm (9.3 mg L^{-1} over 0-100 cm) following the three crops, compared to 4.9 mg L^{-1} on the Inceptisol (0-80 cm). On both soils, profile nitrate-N concentrations were higher following the plant crop than the ratoons. Mean ammonium following the crops was similar in the two soils, at 7.9 mg L⁻¹ (Oxisol) and 7.7 mg L⁻¹ (Inceptisol) (0-80 cm).

Appendix 5.4: Nitrogen: Conclusions

As expected, N application was associated with a significant decrease in topsoil pH, along with increased acid saturation, exchangeable acidity, Fe and Mn and, in some instances, non-significant increases in Cu and Zn. Within the whole soil profile, N application significantly increased soil nitrate-N at all sampling times where mineral N was measured, at each trial site, with concentrations remaining elevated 12-20 months after N application. This effect was not evident for soil ammonium. Nitrogen application led to a significant increase in soil profile Si following each ratoon crop harvest on the Oxisol. This effect may have been linked to reduced crop removal of Si on these treatments.



Appendix 5.5: Mean exchangeable calcium (Ca) per soil depth interval following the second ratoon crop in the (A) Oxisol (n=48, 0-20 cm; n=36, >20 cm at both Si rates) and (B) (Inceptisol (n=30, 0 kg Si ha⁻¹; n=32, 300 kg Si ha⁻¹ at all depths). Within each depth interval, points with different letters are significantly different to each other (P<0.01). Where no significant difference occurred between points within a depth interval, the points are circled.



Appendix 5.6: Mean exchangeable magnesium (Mg) per soil depth interval following the second ratoon crop on the (A) Oxisol (n=48, 0-20 cm; n=36, >20cm at both Si rates) and (B) Inceptisol (n=30, 0 kg Si ha⁻¹; n=32, 300 kg Si ha⁻¹ at all depths). Within each trial site and depth interval, points with different letters are significantly different to each other (P < 0.01). Where no significant difference occurred between points within a depth interval, the points are circled.

CHAPTER 6

OXISOL						
			Magnesium		Total Cations	
	(mg L ') (mg L ') (cmol _c L ')					l _c L ')
Depth	Si applied (kg ha ⁻¹)					
(cm)	0	300	0	300	0	300
0-2.5	1101 a	2 237 d	169 abc	204 c	10.04 ab	14.20 d
2.5-7.5	1 445 bc	2 096 d	175 bc	207 c	11.30 bc	13.52 d
7.5-15	1 312 ab	1 676 bc	146 ab	188 bc	10.20 ab	11.62 c
15-28	1 039 a	1 332 ab	131 a	174 bc	9.61 a	10.47 abc
Significance	P<0.001		P<0.001		P<0.001	
	рН		Exch. Acidity		Acid Sat.	
	(CaCl ₂)		(cmol _c L ⁻¹)		(%)	
Depth	Si applied (kg ha ⁻¹)					
(cm)	0	300	0	300	0	300
0-2.5	4.43 ab	5.13 e	2.04 bc	0.22 a	20.61 d	1.61 a
2.5-7.5	4.53 bc	4.93 d	1.82 bc	0.41 a	16.83 cd	3.18 ab
7.5-15	4.43 ab	4.62 c	2.40 c	1.25 b	20.38 d	11.00 bc
15-28	4.28 a	4.47 bc	3.03 d	2.03 bc	32.42 e	19.61 cd
Significance	P<0	.001	P=0	.018	P<0.001	
INCEPTISOL						
	Calcium (mg L ⁻¹)		Magnesium		Total Cations	
			(mg L ⁻¹)		(cmol _c L ⁻¹)	
Depth	Si applied (kg ha ⁻¹)					
(cm)	0	300	0	300	0	300
0-2.5	1101 a	2 237 d	169 abc	204 c	5.61 a	9.31 b
2.5-7.5	1 445 bc	2 096 d	175 bc	207 c	5.42 a	7.43 ab
7.5-15	1 312 ab	1 676 bc	146 ab	188 bc	5.33 a	7.06 ab
15-28	1 039 a	1 332 ab	131 a	174 bc	5.01 a	6.75 ab
Significance	P<0.001		P<0.001		P<0.001	
	рН (CaCl ₂)		Exch. Acidity		Acid Sat.	
			(cmol _c L⁻¹)		(%)	
Depth	Si applied (kg ha ⁻¹)					
(cm)	0	300	0	300	0	300
0-2.5	4.15 b	5.04 d	0.86 bc	0.06 a	15.67 bcd	0.73 a
2.5-7.5	4.07 ab	4.49 c	1.16 c	0.42 ab	21.78 cd	6.77 ab
7.5-15	4.04 ab	4.43 c	1.27 c	0.69 abc	24.11 d	11.32 abc
15-28	3.89 b	4.13 ab	1.91 d	1.30 cd	38.47 e	25.15 d
Significance P<0.001 P<0.001 P<0.001						

Appendix 6.1: Selected soil characteristics in the silicate treatments (Calmasil® (3 t ha⁻¹) and cement (6 t ha⁻¹), each equivalent to 300 kg Si ha⁻¹) at each soil depth interval

⁺ Within each soil type and characteristic, values with different letters are significantly different at the P level specified.